SOMANETICS CORP Form SC 13D June 25, 2010

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# **SCHEDULE 13D**

**Under the Securities Exchange Act of 1934** 

(Amendment No. )\*

# **SOMANETICS CORPORATION**

(Name of Issuer)

Common Shares, \$0.01 par value per share

(Title of Class of Securities)

834445405

(CUSIP Number)

John H. Masterson

Senior Vice President and General Counsel

Covidien

15 Hampshire Street, Mansfield, Massachusetts 02048

(508) 261-8000

(Name, Address and Telephone Number of Person Authorized to Receive Notices and Communications)

with copies to:

Joseph L. Johnson III

**Goodwin Procter LLP** 

**Exchange Place** 

Boston, Massachusetts 02109

(617) 570-1000

June 16, 2010

(Date of Event which Requires Filing of this Statement)

If the filing person has previously filed a statement on Schedule 13G to report the acquisition that is the subject of this Schedule 13D, and is filing this schedule because of §§240.13d-1(e), 240.13d-1(f) or 240.13d-1(g), check the following box.

**Note:** Schedules filed in paper format shall include a signed original and five copies of the schedule, including all exhibits. See §240.13d-7 for other parties to whom copies are to be sent.

<sup>\*</sup> The remainder of this cover page shall be filled out for a reporting person s initial filing on this form with respect to the subject class of securities, and for any subsequent amendment containing information which would alter disclosures provided in a prior cover page.

The information required on the remainder of this cover page shall not be deemed to be filed for the purpose of Section 18 of the Securities Exchange Act of 1934 ( Act ) or otherwise subject to the liabilities of that section of the Act but shall be subject to all other provisions of the Act (however, see the Notes).

# SCHEDULE 13D

| CUSIP No. 834445405   |
|---|
| 1. Names of Reporting Persons.  |
| I.R.S. Identification Nos. of above persons (entities only)                                 |
| Covidien plc  |
| 98-0624794 2. Check the Appropriate Box if a Member of a Group (See Instructions)           |
| (a) " (b) x   |
| 3. SEC Use Only   |
| 4. Source of Funds (See Instructions)   |
| AF 5. Check if Disclosure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e) " |
| 6. Citizenship or Place of Organization   |
| Ireland Number of 7. Sole Voting Power  |
| Shares Beneficially   |
| 8. Shared Voting Power Owned by   |
| Each  |

| Reporting<br>Person |                  | 9.    | 741,410<br>Sole Dispositive Power   |
|---------------------|------------------|-------|---|
| 1                   | With             |       |   |
|                     |                  | 10.   | Shared Dispositive Power  |
| 11.                 | Aggrega          | ate A | 741,410 mount Beneficially Owned by Each Reporting Person                 |
| 12.                 | 741,<br>Check is |       | Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions) " |
| 13.                 | Percent          | of Cl | ass Represented by Amount in Row (11)                                     |
| 14.                 | 5.9%<br>Type of  |       | orting Person (See Instructions)  |
|                     | CO,              | НС    |   |

# SCHEDULE 13D

| CUSIP No. 834445405  |  |  |  |  |
|--|--|--|--|--|
| <ol> <li>Names of Reporting Persons.</li> <li>I.R.S. Identification Nos. of above persons (entities only)</li> </ol> |  |  |  |  |
| Covidien International Finance S.A.  |  |  |  |  |
| 98-0518567  2. Check the Appropriate Box if a Member of a Group (See Instructions)                                   |  |  |  |  |
| (a) " (b) x  |  |  |  |  |
| 3. SEC Use Only  |  |  |  |  |
|  |  |  |  |  |
| 4. Source of Funds (See Instructions)  |  |  |  |  |
| AF 5. Check if Disclosure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e) "                          |  |  |  |  |
| 6. Citizenship or Place of Organization  |  |  |  |  |
| Luxembourg Number of 7. Sole Voting Power Shares   |  |  |  |  |
| Beneficially  8. Shared Voting Power Owned by  |  |  |  |  |

|     | porting<br>erson | 9.            | 741,410<br>Sole Dispositive Power   |
|-----|------------------|---------------|---|
| V   | With             |               |   |
|     |                  | 10.           | Shared Dispositive Power  |
| 11. | Aggrega          | ate A         | 741,410 mount Beneficially Owned by Each Reporting Person                 |
| 12. |                  | ,410<br>f the | Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions) " |
| 13. | Percent          | of Cl         | ass Represented by Amount in Row (11)                                     |
| 14. | 5.99<br>Type of  |               | orting Person (See Instructions)  |
|     | CO,              | , НС          |   |

# SCHEDULE 13D

| CUSIP No. 834445405                                    |   |
|--|---|
| 1. Names of Reporting Persons.                         |   |
| I.R.S. Identification Nos. of above persons            | (entities only)                             |
|  |   |
| United States Surgical Corporation                     | on  |
| 13-2518270 2. Check the Appropriate Box if a Member of | a Group (See Instructions)                  |
| (a) " (b) x  |   |
| 3. SEC Use Only  |   |
|  |   |
| 4. Source of Funds (See Instructions)                  |   |
| WC  5. Check if Disclosure of Legal Proceedings Is     | s Required Pursuant to Items 2(d) or 2(e) " |
| 6. Citizenship or Place of Organization                |   |
| Delaware Number of 7. Sole Voting Power Shares         |   |
| Beneficially  8. Shared Voting Power Owned by          |   |

|     | porting         | 9.             | 741,410<br>Sole Dispositive Power   |
|-----|-----------------|----------------|---|
| •   | With            |                |   |
|     |                 | 10.            | Shared Dispositive Power  |
| 11. | Aggreg          | ate A          | 741,410 mount Beneficially Owned by Each Reporting Person                 |
| 12. |                 | ,410<br>If the | Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions) " |
| 13. | Percent         | of C           | ass Represented by Amount in Row (11)                                     |
| 14. | 5.99<br>Type of |                | orting Person (See Instructions)  |
|     | CO              |                |   |

# SCHEDULE 13D

| CUSIP           | No. 834445405  |
|-----------------|--|
|                 | Names of Reporting Persons.  R.S. Identification Nos. of above persons (entities only)   |
|                 | Covidien DE Corp.  |
| 2. C            | N/A Check the Appropriate Box if a Member of a Group (See Instructions)                  |
| (8              | a) " (b) x   |
| 3. S            | EC Use Only  |
| 4. S            | Source of Funds (See Instructions)   |
| 5. C            | AF Check if Disclosure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e) " |
| 6. C            | Citizenship or Place of Organization   |
| Numb            | · ·  |
| Benefic<br>Owne | cially  8. Shared Voting Power ed by   |

|     | porting          | 9.    | 741,410<br>Sole Dispositive Power   |
|-----|------------------|-------|---|
| V   | With             |       |   |
|     |                  | 10.   | Shared Dispositive Power  |
| 11. | Aggrega          | ate A | 741,410 mount Beneficially Owned by Each Reporting Person                 |
| 12. | 741.<br>Check is |       | Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions) " |
| 13. | Percent          | of Cl | ass Represented by Amount in Row (11)                                     |
| 14. | 5.9%<br>Type of  |       | orting Person (See Instructions)  |
|     | СО               |       |   |

#### ITEM 1. SECURITY AND ISSUER.

This Statement on Schedule 13D relates to the Common Shares, \$0.01 par value per share of Somanetics Corporation, ( *Somanetics* ), a Michigan corporation. The principal executive offices of Somanetics are located at 2600 Troy Center Drive, Troy, Michigan 48084.

#### ITEM 2. IDENTITY AND BACKGROUND.

This Statement is being jointly filed by Covidien plc, Covidien International Finance S.A., United States Surgical Corporation and Covidien DE Corp. (the *Reporting Persons*) pursuant to Rule 13d-1 under the Securities Exchange Act of 1934, as amended (the *Act*). Set forth below is certain information with respect to each Reporting Person.

#### Covidien plc

Covidien plc, an Irish company, is a publicly held, global healthcare company focused on the development, manufacture and sale of healthcare products for use in clinical and home settings. Covidien plc operates its businesses through three segments: Medical Devices, Pharmaceuticals and Medical Supplies. Covidien plc s principal place of business and principal office is located at 20 Lower Hatch Street, Dublin 2, Ireland.

To the best of Covidien plc s knowledge as of the date hereof, set forth in Schedule I to this Schedule 13D and incorporated herein by reference is the following information with respect to each director and executive officer of Covidien plc:

- (1) name;
- (2) business address;
- (3) present principal occupation or employment and the name, principal business and address of any corporation or other organization in which such employment is conducted; and
- (4) citizenship.

During the last five years, neither Covidien plc nor, to the best of its knowledge, any of its directors or executive officers has been (1) convicted in a criminal proceeding (excluding traffic violations or similar misdemeanors) or (2) a party to a civil proceeding of a judicial or administrative body of competent jurisdiction and as a result of such proceeding has been or is subject to a judgment, decree or final order enjoining future violations of, or prohibiting or mandating activities subject to, federal or state securities laws or finding any violation with respect to such laws.

## Covidien International Finance S.A. ( CIFSA )

CIFSA, a Luxembourg corporation, is a wholly owned subsidiary of Covidien plc. CIFSA is a holding company established in December 2006 to directly and indirectly own substantially all of the operating subsidiaries of Covidien plc., to issue the notes and to perform treasury operations for Covidien plc. Otherwise, CIFSA conducts no independent business. CIFSA s principal place of business and principal office is located at 3b Bld Prince Henri, L-1724, Luxembourg.

To the best of the knowledge CIFSA as of the date hereof, set forth in Schedule I to this Schedule 13D and incorporated herein by reference is the following information with respect to each director of CIFSA:

- (1) name;
- (2) business address;
- (3) present principal occupation or employment and the name, principal business and address of any corporation or other organization in which such employment is conducted; and
- (4) citizenship.

During the last five years, neither CIFSA nor, to the best of its knowledge, any of its directors has been (1) convicted in a criminal proceeding (excluding traffic violations or similar misdemeanors) or (2) a party to a civil proceeding of a judicial or administrative body of competent jurisdiction and as a result of such proceeding has been or is subject to a judgment, decree or final order enjoining future violations of, or

prohibiting or mandating activities subject to, federal or state securities laws or finding any violation with respect to such laws. The managing directors of CIFSA are the functional equivalents of executive officers.

#### United States Surgical Corporation ( USSC )

USSC is a Delaware corporation. USSC is a wholly owned, indirect subsidiary of CIFSA. USSC s principal place of business and principal office is located at 150 Glover Ave., Norwalk, Connecticut 06856.

To the best of USSC s knowledge as of the date hereof, set forth in Schedule I to this Schedule 13D and incorporated herein by reference is the following information with respect to each director and executive officer of USSC:

- (1) name;
- (2) business address;
- (3) present principal occupation or employment and the name, principal business and address of any corporation or other organization in which such employment is conducted; and
- (4) citizenship.

During the last five years, neither USSC nor, to the best of its knowledge, any of its directors or executive officers has been (1) convicted in a criminal proceeding (excluding traffic violations or similar misdemeanors) or (2) a party to a civil proceeding of a judicial or administrative body of competent jurisdiction and as a result of such proceeding has been or is subject to a judgment, decree or final order enjoining future violations of, or prohibiting or mandating activities subject to, federal or state securities laws or finding any violation with respect to such laws.

#### Covidien DE Corp. ( Purchaser )

Purchaser, a Delaware corporation, is a direct wholly owned subsidiary of USSC and has not conducted any business other than in respect to the potential acquisition of all outstanding capital shares of Somanetics. Purchaser s principal place of business and principal office is 15 Hampshire Street, Mansfield, Massachusetts 02048.

To the best of Purchaser s knowledge as of the date hereof, set forth in Schedule I to this Schedule 13D and incorporated herein by reference is the following information with respect to each director and executive officer of Purchaser:

- (1) name;
- (2) business address;
- (3) present principal occupation or employment and the name, principal business and address of any corporation or other organization in which such employment is conducted; and
- (4) citizenship.

During the last five years, neither Purchaser nor, to the best of its knowledge, any of its directors or executive officers has been (1) convicted in a criminal proceeding (excluding traffic violations or similar misdemeanors) or (2) a party to a civil proceeding of a judicial or administrative body of competent jurisdiction and as a result of such proceeding has been or is subject to a judgment, decree or final order enjoining future violations of, or prohibiting or mandating activities subject to, federal or state securities laws or finding any violation with respect to such laws.

#### ITEM 3. SOURCE AND AMOUNT OF FUNDS OR OTHER CONSIDERATION.

On June 16, 2010, USSC, Purchaser and Somanetics entered into an Agreement and Plan of Merger (the *Merger Agreement*) pursuant to which Purchaser will commence a tender offer (the *Offer*) to acquire all of the issued and outstanding common shares, \$0.01 par value per share, of Somanetics (the *Shares*), at \$25.00 per share, net to the seller, in cash for each outstanding Share (the *Offer Price*) less any applicable withholding and transfer taxes and without payment of interest, and, subject to the satisfaction or waiver of the conditions set forth in the Offer and the Merger Agreement, after consummation of the Offer, Purchaser will merge with and into Somanetics (the *Merger*), whereupon Purchaser s separate corporate existence will cease and Somanetics will continue as the surviving corporation and as a wholly owned subsidiary of USSC.

As an inducement to enter into the Merger Agreement, and in consideration thereof, USSC and Purchaser have entered into a Tender and Voting Agreement (the *Tender and Voting Agreement*) with Bruce J. Barrett, a director of Somanetics and the President and Chief Executive Officer of Somanetics (the *Shareholder*). This agreement provides, among other things, that the Shareholder will tender his Shares in the Offer. The Shareholder s Shares may only be withdrawn from the Offer if the Tender and Voting Agreement is terminated in accordance with its terms, including if the Merger Agreement is terminated. In addition, the Shareholder has agreed, in exchange for the consideration described in the Merger Agreement, to tender to Purchaser in the Offer any Shares (i) then owned of record by the Shareholder or for which the Shareholder has sole dispositive power not later than five business days following the commencement of the Offer, and (ii) subsequently acquired not later than the earlier of (i) 5 business days following such acquisition or the time at which Purchaser first accepts any Shares for payment pursuant to the Offer (the *Acceptance Time*). The Shareholder may not withdraw any such Shares so tendered unless the Offer is terminated. As of June 16, 2010, the Tender and Voting Agreement covered 124,691 Shares, 62,800 restricted Shares subject to vesting and securities convertible into or exercisable for 553,919 Shares, totaling approximately 5.9% of the Shares then outstanding on a fully diluted basis (assuming exercise or conversion of all derivative securities into Shares held by the Shareholder).

Pursuant to the Tender and Voting Agreement, the Shareholder has granted USSC, until the earlier of (i) the effective time of the Merger, (ii) December 15, 2010, provided that such date may be extended to March 15, 2011 by either USSC or Somanetics in the event that all conditions to the Merger have been satisfied other than the Minimum Condition (as defined below) and the receipt of approvals or clearance under, or the expiration or termination of any waiting periods under, the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, or any provision under comparable applicable laws of foreign jurisdictions, (iii) the date of any modification, waiver or amendment to the Merger Agreement in

a manner that reduces the Offer Price, and (iv) the termination of the Merger Agreement (the *Termination Date*), a limited irrevocable proxy to vote all Securities (as defined below) for the Shareholder and in Shareholder's name, place and stead. Such proxy may be exercised at any annual or special meeting or action of the shareholders of Somanetics, as applicable, or at any adjournment thereof or pursuant to any consent of the shareholders of Somanetics, in lieu of a meeting or otherwise, whether before or after the Acceptance Time, for the adoption and approval of the Merger Agreement. *Securities*, as used above, means (i) Shares, (ii) all securities exchangeable, exercisable for or convertible into Shares, and (iii) any securities issued or exchanged with respect to such Shares, and upon any recapitalization, reclassification, merger, consolidation, spin-off, partial or complete liquidation, stock dividend, split-up or combination of the securities of Somanetics or upon any other change in Somanetics capital structure, in each case whether now owned or hereafter acquired by the Shareholder.

The Shareholder has granted to USSC an irrevocable option (the *Option*) to purchase all of the Shareholder s right, title and interest in and to any Shares owned of record by the Shareholder or for which the Shareholder has sole dispositive power and any Shares acquired by the Shareholder after the commencement of the Offer, other than unexercised Company Stock Options and restricted Shares subject to vesting held by the Shareholder ( *Non-Tender Securities*), at a price per share equal to the Offer Price. USSC may exercise the Option in whole, but not in part, if, but only if, (i) Purchaser shall have acquired Shares pursuant to the Offer and (ii) the Shareholder fails to tender into the Offer any Shares (other than the Shareholder s Non-Tender Securities) or withdraws the tender of any Shares (other than the Shareholder s Non-Tender Securities) into the Offer. USSC may exercise the Option at any time within the 60 days following the date when such Option first becomes exercisable.

During the term of the Tender and Voting Agreement, except as otherwise provided therein, the Shareholder agrees not to (i) directly or indirectly, sell, transfer, assign, pledge, hypothecate, tender, encumber or otherwise dispose of or limit its right to vote in any manner any of the Securities, or agree to do any of the foregoing, or (ii) take any action which would have the effect of preventing or disabling the Shareholder from performing its obligations under the Tender and Voting Agreement.

Shared voting and dispositive power with respect to the Shares owned by the Shareholder may be deemed to have been acquired through execution of the Tender and Voting Agreement. The Reporting Persons have not expended any funds in connection with the execution of the Tender and Voting Agreement, except for the transaction expenses otherwise to be incurred in connection with the Offer and the Merger.

CIFSA, USSC and Purchaser estimate that, if Purchaser acquires all of the Shares (on a *fully diluted basis*, which, for purposes of such calculation, includes the number of Shares outstanding, together with the Shares which Somanetics may be required to issue pursuant to warrants, options or other obligations outstanding at that date under employee stock or similar benefit plans or otherwise, that are then vested and then exercisable at prices equal to or less than the Offer Price) pursuant to the Offer, the total amount of funds required to purchase all outstanding Shares pursuant to the Offer and the Merger and to pay related fees and expenses will be approximately \$348 million. USSC or one of its affiliates will provide all funding required by Purchaser in connection with the Offer from cash on hand. CIFSA has provided a guaranty to Somanetics for the full performance and payment by USSC of its obligations pursuant to or in connection with the Merger Agreement, the Offer, the Merger and the other transactions contemplated by the Merger Agreement and is liable for any breach of, or other failure to perform, any such representation, warranty, covenant, obligation, agreement or undertaking of USSC.

A copy of the Merger Agreement is attached as Exhibit 2 to this Schedule 13D. A copy of the Tender and Voting Agreement is included as Exhibit 3 to this Schedule 13D. A copy of the Guaranty by CIFSA is attached as Exhibit 4 to this Schedule 13D. References to, and descriptions of, the Merger Agreement, the Tender and Voting Agreement and the Guaranty as set forth above in this Item 3 are qualified in their entirety by reference to the copies of the Merger Agreement, the Tender and Voting Agreement and the Guaranty included as Exhibits to this Schedule 13D and which are incorporated herein in their entirety by this reference. The information set forth and/or incorporated by reference in Item 6 is hereby incorporated by reference into this Item 3.

#### ITEM 4. PURPOSE OF TRANSACTION.

The information set forth and/or incorporated by reference in Items 2 and 3 is hereby incorporated by reference into this Item 4.

The purpose of entering into the Merger Agreement and the Tender and Voting Agreement, and the purpose of the Offer, is to enable USSC and Purchaser to acquire control of, and ultimately the entire equity interest in, Somanetics. The Offer is conditioned upon, among other things, immediately prior to the expiration of the Offer, the number of Shares validly tendered and not validly withdrawn, together with any Shares beneficially owned by USSC or any affiliate of USSC, being equal to at least a majority of the Shares then outstanding on a fully diluted basis (the *Minimum Condition*) and the applicable waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, having expired or been terminated. Subject to the terms of the Merger Agreement and applicable law, Purchaser reserves the right to withdraw the Offer and to not take up and pay for any Shares deposited in the Offer unless each of the conditions to the Offer is satisfied or, where permitted, waived by Purchaser at or prior to the expiration of the Offer.

Pursuant to the Merger Agreement, Somanetics has granted Purchaser an irrevocable option (the *Top-Up Option*) to purchase at a price per share equal to the Offer Price up to that number of newly issued Shares (the *Top-Up Shares*) equal to the lowest number of Shares that, when added to the number of Shares directly owned by Purchaser or USSC at the time of exercise

of the Top-Up Option, will constitute one share more than 90% of the Shares outstanding immediately after the issuance of the Top-Up Shares (determined on a fully diluted basis (which, for purposes of such calculation, includes all restricted Shares subject to vesting and all Shares that Somanetics may be required to issue pursuant to options that are outstanding at that date, regardless of the exercise price, the vesting schedule or other terms and conditions thereof)). However, the Top-Up Option will not be exercisable for a number of Shares in excess of the Shares authorized and unissued at the time of exercise of the Top-Up Option, and the Top-Up Option may not be exercised unless, following the time we accept the Shares in the Offer or after a subsequent offering period, 80% or more of the Shares will be directly or indirectly owned by us or USSC. We may exercise the Top-Up Option once at any time following the Acceptance Time.

If the conditions of the Offer are satisfied or waived and Purchaser takes up and pays for any Shares validly deposited under the Offer, Purchaser intends to acquire any Shares not deposited under the Offer through a merger under the Michigan Business Corporation Act (the MBCA), or through a subsequent offering period, in each case, for cash consideration per Share equal to the Offer Price. The exact timing and details of any such transaction will depend upon a number of factors, including the number of Shares acquired pursuant to the Offer. Although Purchaser intends to effect such a transaction generally on the terms described herein, it is possible that, as a result of delays in Purchaser s ability to effect such a transaction, information subsequently obtained by the Reporting Persons, changes in general economic or market conditions or in the business of Somanetics, or other currently unforeseen circumstances, such a transaction may be delayed or abandoned or may be proposed on different terms.

Pursuant to the Merger Agreement, promptly upon the payment by Purchaser for any Shares accepted by Purchaser for payment pursuant to the Offer at the Acceptance Time, which Shares represent at least a majority of the issued and outstanding Shares pursuant to the Offer, USSC shall be entitled to designate such number of directors on the Somanetics board of directors as will give USSC, subject to compliance with Section 14(f) of the Exchange Act of 1934, as amended, representation on the Somanetics board of directors equal to at least that number of directors, rounded up to the next whole number, which is the product of (x) the total number of directors on the Somanetics board of directors (giving effect to the directors elected pursuant to this sentence) multiplied by (y) the percentage that (I) such number of Shares so accepted for payment and paid for by Purchaser plus the number of Shares otherwise owned by USSC. Purchaser or any other subsidiary of USSC bears to (II) the number of such Shares outstanding, and Somanetics shall, at such time, cause USSC s designees to be so elected. From and after the Acceptance Time, Somanetics and USSC shall use commercially reasonable best efforts, as permitted by applicable laws and the rules of NASDAQ Global Select Market, upon USSC s request, to also cause persons elected or designated by USSC to constitute the same percentage (rounded up to the next whole number) as is on the Somanetics board of directors of (i) each committee of the Somanetics board of directors, (ii) each board of directors (or similar body) of each of Somanetics subsidiaries, and (iii) each committee (or similar body) of each such board, in each case only to the extent permitted by applicable law or the rules of any stock exchange on which the Shares are listed. In connection with the foregoing, Somanetics shall promptly, at the option of Purchaser, either increase the size of the Somanetics board of directors or obtain the resignation of such number of its current directors, or both, as is necessary to enable USSC s designees to be elected or appointed to the Somanetics board of directors as provided above.

Following completion of the Offer and the Merger, the Reporting Persons intend to operate the Somanetics business through one or more subsidiaries of Covidien plc under the direction of Covidien plc s management. The Reporting Persons intend to continue to review Somanetics business, operations, capitalization and management and consider various possible business strategies that they may pursue if they acquire control of Somanetics, whether pursuant to the Offer, the Merger or otherwise. These changes could include, among other things, material changes in Somanetics business, corporate structure, capitalization and management. USSC and Covidien plc reserve the right to change their plans and intentions at any time, as they deem appropriate.

If permitted by applicable law, subsequent to the completion of the Offer and a short-form merger or any subsequent offering period, if necessary, the Reporting Persons intend to terminate the registration pursuant to Section 12(g)(4) and Section 15 of the Securities Exchange Act of 1934, as applicable, and delist the Shares from The Nasdaq Global Market.

After Purchaser purchases the Shares pursuant to the Offer, USSC may appoint its representatives to the Somanetics board of directors in proportion to its ownership of the outstanding Shares, as described above. Prior to the effective time of the Merger, Somanetics will cause its directors, other than the designees of Purchaser or USSC, to execute and deliver a letter effectuating his or her resignation as a director of Somanetics upon the effective time of the Merger.

Except as otherwise set forth in this Schedule 13D, the Reporting Persons have no present plans or proposals which relate to or would result in:

- a) The acquisition by any person of additional securities of the issuer, or the disposition of securities of the issuer;
- b) An extraordinary corporate transaction, such as a merger, reorganization or liquidation, involving the issuer or any of its subsidiaries;
- c) A sale or transfer of a material amount of assets of the issuer or any of its subsidiaries;

- d) Any change in the present board of directors or management of the issuer, including any plans or proposals to change the number or term of directors or to fill any existing vacancies on the board;
- e) Any material change in the present capitalization or dividend policy of the issuer;

9

- f) Any other material change in the issuer s business or corporate structure including but not limited to, if the issuer is a registered closed-end investment company, any plans or proposals to make any changes in its investment policy for which a vote is required by section 13 of the Investment Company Act of 1940;
- g) Changes in the issuer s charter, bylaws or instruments corresponding thereto or other actions which may impede the acquisition of control of the issuer by any person;
- h) Causing a class of securities of the issuer to be delisted from a national securities exchange or to cease to be authorized to be quoted in an inter-dealer quotation system of a registered national securities association;
- i) A class of equity securities of the issuer becoming eligible for termination of registration pursuant to Section 12(g)(4) of the Securities Exchange Act of 1934; or
- j) Any action similar to any of those enumerated above.

#### ITEM 5. INTEREST IN SECURITIES OF THE ISSUER.

The information set forth and/or incorporated by reference in Items 2, 3 and 4 is hereby incorporated by reference into this Item 5.

a) As of the date hereof, the Reporting Persons own no Shares. For purposes of Rule 13d-3 under the Exchange Act, however, as a result of entering into the Tender and Voting Agreement, the Reporting Persons may be deemed to possess beneficial ownership of an aggregate of 741,410 Shares, representing approximately 5.9% of the Shares then outstanding on a fully diluted basis (assuming exercise or conversion into Shares of all derivative securities held by the Shareholder). The Reporting Persons and the other persons listed on Schedule I hereto, however, disclaim beneficial ownership of such Shares, and this statement shall not be construed as an admission that any of the Reporting Persons or those listed on Schedule I hereto is the beneficial owner for any purpose of the Shares covered by this 13D disclosure.

Except as set forth in this Schedule 13D, (1) to the best of Covidien plc s knowledge as of the date hereof, neither Covidien plc nor any of its directors and executive officers named in Schedule I hereto owns any Shares, (2) to the best of CIFSA s knowledge as of the date hereof, neither CIFSA nor any of its directors and executive officers named in Schedule I hereto owns any Shares, (3) to the best of USSC s knowledge as of the date hereof, neither USSC nor any of its directors and executive officers named in Schedule I hereto owns any Shares, and (4) to the best of Purchaser s knowledge as of the date hereof, neither Purchaser nor any of its directors and executive officers named in Schedule I hereto owns any Shares.

- b) Prior to June 16, 2010, none of the Reporting Persons owned or was the beneficial owner (as defined in Rule 13d-3 promulgated under the Exchange Act) of any Shares. Upon execution of the Tender and Voting Agreement, the Reporting Persons may be deemed to have acquired beneficial ownership (as defined in Rule 13d-3 promulgated under the Exchange Act) of Shares, because pursuant to the Tender and Voting Agreement, the Reporting Persons may be deemed to have acquired the shared power to vote or direct the vote and to dispose or to direct the disposition of the 741,410 Shares, representing approximately 5.9% of the Shares then outstanding on a fully diluted basis (assuming exercise or conversion of all securities convertible into Shares held by the Shareholder). Upon the consummation of the Offer, the Reporting Persons would have sole voting and investment power over such Shares (assuming exercise or conversion of all securities convertible into Shares held by the Shareholder), excluding 62,800 restricted Shares subject to vesting, which the Reporting Persons would have sole voting and investment power over upon consummation of the Merger.
- c) Except for the transactions described herein, (1) to the best of Covidien plc s knowledge as of the date hereof, neither Covidien plc nor any of its directors and executive officers named in Schedule I hereto has effected any transaction in Shares during the past 60 days, (2) to the best of CIFSA s knowledge as of the date hereof, neither CIFSA nor any of its directors and executive officers named in Schedule I hereto has effected any transaction in Shares during the past 60 days, (3) to the best of USSC s knowledge as of the date hereof, neither USSC nor any of its directors and executive officers named in Schedule I hereto has effected any transaction in Shares during the past 60 days and (4) to the best of Purchaser s knowledge as of the date hereof, neither Purchaser nor any of its directors and executive officers named in Schedule I hereto has effected any transaction in Shares during the past 60 days.
- d) Other than the Shareholder identified in Item 3 who is party to the Tender and Voting Agreement attached as Exhibit 3 to this Schedule 13D and incorporated herein by reference, (1) to the best of Covidien plc s knowledge as of the date hereof, neither Covidien plc nor any of its directors and executive officers named in Schedule I hereto has or knows any other person who has the right to receive or the power to direct the receipt of dividends from, or the proceeds from the sale of, any Shares beneficially owned by Covidien plc, (2) to the best of CIFSA s knowledge as of the date hereof, neither CIFSA nor any of its directors and executive officers named in Schedule I hereto has or knows any other person who has the right to receive or the power to direct the receipt of dividends from, or the proceeds from the sale of, any Shares beneficially owned

by CIFSA, (3) to the best of USSC s knowledge as of the date hereof, neither USSC nor any of its directors and executive officers named in Schedule I hereto has or knows any other person who has the right to receive or the power to direct the receipt of dividends from, or the proceeds from the sale of, any Shares beneficially owned by USSC, and (4) to the best of Purchaser s knowledge as of the date hereof, neither Purchaser nor any of its directors and executive officers named in Schedule I hereto has or knows any other person who has the right to receive or the power to direct the receipt of dividends from, or the proceeds from the sale of, any Shares beneficially owned by Purchaser.

#### ITEM 6. CONTRACTS, ARRANGEMENTS, UNDERSTANDINGS OR RELATIONSHIPS WITH RESPECT TO SECURITIES OF THE ISSUER.

The information set forth and/or incorporated by reference in Items 2 through 5 is hereby incorporated by reference into this Item 6.

As described in Item 3, on June 16, 2010, USSC and Purchaser entered into the Tender and Voting Agreement with the Shareholder. During the term of the Tender and Voting Agreement, except as otherwise provided therein, the Shareholder agreed not to take any action, in its capacity as a shareholder of Somanetics, that Somanetics is prohibited from taking pursuant to Section 5.2 of the Merger Agreement.

Except for the agreements described above or in Item 3, there are no contracts, arrangements, understandings or relationships (legal or otherwise), including, but not limited to, transfer or voting of any of the securities, finder s fees, joint ventures, loan or option arrangements, puts or calls, guarantees of profits, division of profits or loss, or the giving or withholding of proxies, between the persons enumerated in Item 2, and any other person, with respect to any securities of Somanetics, including any securities pledged or otherwise subject to a contingency the occurrence of which would give another person voting power or investment power over such securities other than standard default and similar provisions contained in loan agreements.

References to, and descriptions of, the Merger Agreement and the Tender and Voting Agreement as set forth above in this Item 6 are qualified in their entirety by reference to the copies of the Merger Agreement included as Exhibit 2 to this Schedule 13D and the Tender and Voting Agreement included as Exhibit 3 to this Schedule 13D and which are incorporated herein in their entirety by this reference.

#### ITEM 7. MATERIAL TO BE FILED AS EXHIBITS.

| Exhibit | Description  |
|---------|--|
| 1       | Joint Filing Agreement, among Covidien plc, Covidien International Finance, S.A., United States Surgical Corporation and Covidien DE Corp., dated June 25, 2010. |
| 2       | Agreement and Plan of Merger, by and among United States Surgical Corporation, Covidien DE Corp., and Somanetics Corporation, dated June 16, 2010.               |
| 3       | Tender and Voting Agreement, dated as of June 16, 2010, by and among United States Surgical Corporation, Covidien DE Corp. and Bruce J. Barrett.                 |
| 4       | Guaranty, dated as of June 16, 2010, by Covidien International Finance S.A.  |

#### **SIGNATURES**

After reasonable inquiry and to the best of its knowledge and belief, the undersigned certifies that the information set forth in this Statement is true, complete and correct.

Dated: June 25, 2010

#### **COVIDIEN PLC**

By: /s/ JOHN W. KAPPLES
Name: John W. Kapples
Title: Vice President and Secretary

#### COVIDIEN INTERNATIONAL FINANCE S.A.

By: /s/ MICHELANGELO F. STEFANI
Name: Michelangelo F. Stefani
Title: Managing Director

#### UNITED STATES SURGICAL CORPORATION

By: /s/ MATTHEW J. NICOLELLA
Name: Matthew J. Nicolella
Title: Vice President and Assistant Secretary

#### COVIDIEN DE CORP.

By: /s/ MATTHEW J. NICOLELLA
Name: Matthew J. Nicolella
Title: Vice President and Assistant Secretary

#### SCHEDULE I

The names of the directors and executive officers (or functional equivalents) of Covidien plc, CIFSA, USSC and Purchaser and their present principal occupations or employment are set forth below. Unless otherwise indicated, each director and executive officer (or functional equivalent) has been so employed for a period in excess of five years. Unless otherwise indicated, (i) the principal business address for each director of Covidien plc is 20 Lower Hatch Street, Dublin 2, Ireland, and the business telephone number for each such director or officer is +353 1 438-1700, (ii) the principal business address for each director of CIFSA is 3b Bld Prince Henri, L-1724, Luxembourg and (iii) the principal business address for each officer of Covidien plc and each director and officer of USSC and Purchaser is 15 Hampshire Street, Mansfield, Massachusetts 02048 and the business telephone number for each such director and officer is (508) 261-8000. Unless otherwise indicated, each director and executive officer (or functional equivalent) is a citizen of the United States. References below to Covidien include Covidien plc, its predecessor, Covidien Ltd., and the healthcare business of Tyco International Ltd. for all periods prior to the separation of Covidien Ltd. from Tyco International Ltd.

#### **COVIDIEN PLC**

#### **Directors**

Richard J. Meelia Chairman of the Board of Directors, President and Chief Executive Officer of Covidien. Mr. Meelia is also President of USSC and the Purchaser. Mr. Meelia s principal business address is 15 Hampshire Street, Mansfield, Massachusetts 02048 and his business telephone number is (508) 261-8000.

Craig Arnold Vice Chairman and Chief Operating Officer, Industrial Sector of Eaton Corporation, a diversified industrial manufacturer.

Robert H. Brust Chief Financial Officer of Sprint Nextel Corporation, a wireless and wireline communications company.

John M. Connors, Jr. Chairman Emeritus of Hill, Holliday, Connors, Cosmopulos, Inc., a full-service advertising agency that is part of The Interpublic Group of Companies, Inc.

Christopher J. Coughlin Executive Vice President and Chief Financial Officer of Tyco International Ltd., a global provider of security products and services, fire protection and detection products and services, valves and controls, and other industrial products.

Timothy M. Donahue Member of the Board of Directors of Eastman Kodak Company, NVR, Inc. and Tyco International Ltd.

Kathy J. Herbert Member of the Board of Directors of Covidien plc.

Randall J. Hogan, III Chairman and Chief Executive Officer of Pentair, Inc., an industrial manufacturing company.

Dennis H. Reilley Member of the Board of Directors of H.J. Heinz Company, Marathon Oil Corporation and The Dow Chemical Company.

Tadataka Yamada President of the Global Health Program of the Bill & Melinda Gates Foundation.

Joseph A. Zaccagnino Member of the Board of Directors of NewAlliance Bancshares, Inc.

#### **Officers**

Richard J. Meelia See above under Covidien plc Directors.

Charles J. Dockendorff Executive Vice President and Chief Financial Officer of Covidien and Vice President of USSC and Purchaser.

Jose E. Almeida Senior Vice President of Covidien and President, Medical Devices of Covidien.

Timothy R. Wright Senior Vice President of Covidien and President, Pharmaceuticals of Covidien. Mr. Wright s principal business address is 675 McDonnell Blvd., Hazelwood, Missouri 63042 and his business telephone number is (314) 654-2000.

Eric A. Kraus Senior Vice President, Corporate Communications of Covidien.

John H. Masterson Senior Vice President and General Counsel of Covidien and Vice President and Assistant Secretary of Purchaser.

Amy A. McBride-Wendell Senior Vice President, Strategy and Business Development of Covidien.

Michael P. Dunford Senior Vice President, Human Resources of Covidien.

13

Richard G. Brown, Jr. Vice President, Chief Accounting Officer and Corporate Controller of Covidien and Vice President of USSC and Purchaser.

Kevin G. DaSilva Vice President and Treasurer of Covidien, member of the board of directors of CIFSA and member of the board of directors and Vice President and Treasurer of USSC and Purchaser.

Eric C. Green Vice President and Chief Tax Officer of Covidien, member of the board of directors of CIFSA and Vice President and Assistant Treasurer of USSC and Purchaser.

Coleman N. Lannum Vice President, Investor Relations of Covidien.

James C. Clemmer President, Medical Supplies Sector of Covidien.

#### COVIDIEN INTERNATIONAL FINANCE S.A.

Anton Stadtbaumer Member of the board of directors of CIFSA, manager of Covidien Group S.a.r.l. and Regional Treasurer of Covidien Finance GmbH. Mr. Stadtbaumer s principal business address is Victor von Bruns Strasse 19, 8212 Neuhausen am Rheinfall, Switzerland and his business telephone number is +41 52 556 0677. Mr. Stadtbaumer has German citizenship.

Michelangelo Stefani Managing Director of CIFSA. Mr. Stefani has Italian citizenship.

Kevin G. DaSilva See above under Covidien plc Officers.

Eric C. Green See above under Covidien plc Officers.

Erik De Gres Member of the board of directors of CIFSA. Mr. De Gres has Belgian citizenship.

Thomas Ford Member of the board of directors of CIFSA and Manager and Deputy General Counsel-EMEA of Covidien Group. Mr. Ford has United Kingdom citizenship.

#### UNITED STATES SURGICAL CORPORATION

#### Directors

Kevin G. DaSilva See above under Covidien plc Officers.

John W. Kapples Member of the board of directors of both USSC and Purchaser, Vice President and Secretary of USSC and Vice President and Secretary of Covidien and Tyco Healthcare Group LP, d/b/a Covidien.

Matthew J. Nicolella Member of the board of directors and a Vice President and Assistant Secretary of both USSC and Purchaser and Vice President and Chief Mergers & Acquisitions/Licensing Counsel of Tyco Healthcare Group LP d/b/a/ Covidien.

#### Officers

Richard J. Meelia See above under Covidien plc Directors.

Richard G. Brown See above under Covidien plc Officers.

Stephen C. Carey Vice President and Assistant Treasurer of USSC and Purchaser and Vice President, Tax Reporting of Tyco Healthcare Group LP d/b/a/ Covidien.

Kevin G. DaSilva See above under Covidien plc Officers.

Charles J. Dockendorff See above under Covidien plc Officers.

Mark Farber Vice President and Assistant Secretary of USSC and Vice President and General Counsel, Surgical Devices, of Tyco Healthcare Group LP, d/b/a Covidien. Mr. Farber s principal business address is 150 Glover Avenue, Norwalk, Connecticut 06850 and his business telephone number is (203) 845-1000.

Eric. C. Green See above under Covidien plc Officers.

John W. Kapples See above under United States Surgical Corporation Directors.

Michael W. Lyons Vice President and Assistant Secretary of USSC and Associate General Counsel, Surgical Devices, of Tyco Healthcare Group LP, d/b/a Covidien. Mr. Lyons principal business address is 150 Glover Avenue, Norwalk, Connecticut 06850 and his business telephone number is (203) 845-1000.

John H. Masterson See above under Covidien plc Officers.

Matthew J. Nicolella See above under United States Surgical Corporation Directors.

Lawrence T. Weiss Vice President and Assistant Secretary of USSC and Purchaser and Vice President and Chief International Counsel of Tyco Healthcare Group LP, d/b/a Covidien.

Joseph Wuestner Vice President and Assistant Secretary of USSC and Vice President, Chief Corporate Counsel, Pharmaceuticals, of Tyco Healthcare Group LP, d/b/a Covidien. Mr. Wuestner s principal business address is 675 McDonnell Blvd., Hazelwood, MO 63042 and his business telephone number is (314) 654-2000.

#### COVIDIEN DE CORP.

#### **Directors**

Kevin G. DaSilva See above under Covidien plc Officers.

John W. Kapples See above under United States Surgical Corporation Directors.

Matthew J. Nicolella See above under United States Surgical Corporation Directors.

#### **Officers**

Richard J. Meelia See above under Covidien plc Directors.

Richard G. Brown See above under Covidien plc Officers.

Stephen C. Carey See above under United States Surgical Corporation Officers.

Kevin G. DaSilva See above under Covidien plc Officers.

Charles J. Dockendorff See above under Covidien plc Officers.

Eric. C. Green See above under Covidien plc Officers.

Lisa K. Golod Vice President and Assistant Treasurer of Purchaser and Vice President of Tax Planning of Tyco Healthcare Group LP, d/b/a Covidien.

John H. Masterson See above under Covidien plc Officers.

Matthew J. Nicolella See above under United States Surgical Corporation Directors.

Lawrence T. Weiss See above under United States Surgical Corporation Officers.

Exhibit 1

#### JOINT FILING AGREEMENT

In accordance with Rule 13d-1(k)(1) under the Securities Exchange Act of 1934, as amended, the undersigned hereby agree that this Statement on Schedule 13D relating to the beneficial ownership of Common Shares, \$0.01 par value per share, of Somanetics Corporation is being filed with the Securities and Exchange Commission on behalf of each of them. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

Dated: June 25, 2010

#### **COVIDIEN PLC**

By: /s/ JOHN W. KAPPLES
Name: John W. Kapples
Title: Vice President and Secretary

#### COVIDIEN INTERNATIONAL FINANCE S.A.

By: /s/ MICHELANGELO F. STEFANI
Name: Michelangelo F. Stefani
Title: Managing Director

#### UNITED STATES SURGICAL CORPORATION

By: /s/ MATTHEW J. NICOLELLA
Name: Matthew J. Nicolella
Title: Vice President and Assistant Secretary

# COVIDIEN DE CORP.

By: /s/ MATTHEW J. NICOLELLA
Name: Matthew J. Nicolella
Title: Vice President and Assistant Secretary

Exhibit 2

AGREEMENT AND PLAN OF MERGER

AMONG

UNITED STATES SURGICAL CORPORATION

COVIDIEN DE CORP. AND

SOMANETICS CORPORATION

Dated as of June 16, 2010

# TABLE OF CONTENTS

| CECTI         | ION 1 THE OFFER AND THE MERGER                            | Page 2 |
|---------------|---|--------|
| SEC 11<br>1.1 | The Offer   | 2      |
| 1.1           | Company Actions   | 4      |
| 1.3           | The Merger  | 5      |
| 1.4           | Effective Time  | 5      |
| 1.5           | Closing   | 6      |
| 1.6           | Directors and Officers of the Surviving Corporation       | 6      |
| 1.7           | Subsequent Actions  | 6      |
| 1.8           | Shareholders Meeting                                      | 6      |
| 1.9           | Merger Without Meeting of Shareholders                    | 7      |
| 1.10          | Top-Up Option   | 7      |
|               |   | ·      |
|               | ION 2 CONVERSION OF SECURITIES                            | 8      |
| 2.1           | Conversion of Capital Stock                               | 8      |
| 2.2           | Exchange of Certificates                                  | 9      |
| 2.3           | Company Stock Plans                                       | 11     |
| 2.4           | Section 16  | 12     |
| 2.5           | Withholding   | 12     |
| 2.6           | Transfer Taxes  | 12     |
| SECTI         | ION 3 REPRESENTATIONS AND WARRANTIES OF COMPANY           | 12     |
| 3.1           | Organization and Qualification                            | 12     |
| 3.2           | Authority to Execute and Perform Agreement                | 13     |
| 3.3           | Capitalization  | 14     |
| 3.4           | Company Subsidiaries                                      | 15     |
| 3.5           | SEC Reports   | 16     |
| 3.6           | Financial Statements                                      | 17     |
| 3.7           | Absence of Undisclosed Liabilities                        | 18     |
| 3.8           | Absence of Adverse Changes                                | 18     |
| 3.9           | Compliance with Laws                                      | 19     |
| 3.10          | Actions and Proceedings                                   | 22     |
| 3.11          | Contracts and Other Agreements                            | 22     |
| 3.12          | Property  | 24     |
| 3.13          | Insurance   | 27     |
| 3.14          | Commercial Relationships                                  | 28     |
| 3.15          | Tax Matters   | 28     |
| 3.16          | Employee Benefit Plans                                    | 31     |
| 3.17          | Employee Relations  | 33     |
| 3.18          | Environmental Matters                                     | 35     |
| 3.19          | No Breach   | 36     |
| 3.20          | Board Approvals; Anti-Takeover; Vote Required             | 36     |
| 3.21          | Financial Advisor   | 37     |
| 3 22          | Information in the Offer Documents and the Schedule 14D 0 | 29     |

| 3.23       | Information in the Proxy Statement   | 38       |
|------------|--|----------|
|            | •  |          |
| SECTION    |  | 39       |
| 4.1        | Organization   | 39       |
| 4.2        | Authority to Execute and Perform Agreement                                     | 39       |
| 4.3<br>4.4 | Information in the Offer Documents Information in the Proxy Statement          | 39<br>40 |
| 4.4        | Sub  | 40       |
| 4.6        | Financing  | 40       |
| 4.7        | Ownership of Shares  | 40       |
| 4.8        | Litigation   | 40       |
| 4.9        | Financial Advisor  | 41       |
| 4.10       | No Additional Representations  | 41       |
|            |  |          |
|            | N 5 COVENANTS AND AGREEMENTS   | 41       |
| 5.1        | Conduct of Business  | 41       |
| 5.2        | No Solicitation  | 45       |
| SECTION    | N 6 ADDITIONAL AGREEMENTS  | 49       |
| 6.1        | Proxy Statement  | 49       |
| 6.2        | Meeting of Shareholders of the Company   | 49       |
| 6.3        | NASDAQ; Post-Closing SEC Reports   | 49       |
| 6.4        | Access to Information  | 50       |
| 6.5        | Public Disclosure  | 50       |
| 6.6        | Regulatory Filings; Reasonable Efforts   | 51       |
| 6.7        | Notification of Certain Matters  | 52       |
| 6.8        | Litigation   | 53       |
| 6.9        | Indemnification  | 53       |
| 6.10       | Directors  | 54       |
| 6.11       | 401(k)   | 56       |
| 6.12       | Employee Benefit Matters   | 56       |
| SECTION    | N 7 CONDITIONS PRECEDENT TO THE OBLIGATION OF PARTIES TO CONSUMMATE THE MERGER | 57       |
| 7.1        | Conditions to Obligations of Each Party to Effect the Merger                   | 57       |
| 7.2        | Frustration of Conditions  | 58       |
| SECTION    | N 8 TERMINATION, AMENDMENT AND WAIVER  | 58       |
| 8.1        | Termination  | 58       |
| 8.2        | Effect of Termination  | 60       |
| 8.3        | Fees and Expenses  | 61       |
| 8.4        | Amendment  | 61       |
| 8.5        | Waiver   | 61       |
| SECTION    | N 9 MISCELLANEOUS  | 61       |
| 9.1        | No Survival  | 61       |
| 9.2        | Notices  | 61       |
| 9.3        | Entire Agreement   | 63       |
| 9.4        | Governing Law  | 63       |
| 9.5        | Binding Effect; No Assignment; No Third-Party Beneficiaries                    | 63       |

| 9.6  | Counterparts and Signature                 | 63 |
|------|--|----|
| 9.7  | Severability                               | 64 |
| 9.8  | Submission to Jurisdiction; Waiver         | 64 |
| 9.9  | Time is of the Essence                     | 64 |
| 9.10 | Parent Guarantee                           | 64 |
| 9.11 | Enforcement                                | 64 |
| 9.12 | Rules of Construction; Certain Definitions | 65 |
| 9.13 | No Waiver; Remedies Cumulative             | 66 |
| 9.14 | Waiver of Jury Trial                       | 66 |
|      |  |    |

# **ANNEXES**

Annex I Conditions to the Offer

Annex II Form of Tender and Voting Agreement

# **Index of Defined Terms**

|                                   | Section    |
|-----------------------------------|------------|
| Acceptance Time                   | 1.1(a)     |
| Acquisition Proposal              | 5.2(a)     |
| Adverse Recommendation Change     | 5.2(c)     |
| Affiliate                         | 3.3(c)     |
| Agreement                         | Preamble   |
| Alternative Acquisition Agreement | 5.2(c)     |
| Antitrust Counsel Only Material   | 6.6(d)     |
| Antitrust Laws                    | 6.6(b)     |
| Assignee                          | 9.5(a)     |
| Book-Entry Shares                 | 2.1(c)     |
| CERCLA                            | 3.18(b)    |
| Certificate of Merger             | 1.4        |
| Certificate                       | 2.1(c)     |
| CEO Shareholder                   | Recitals   |
| Closing                           | 1.5        |
| Closing Date                      | 1.5        |
| Code                              | 2.5        |
| Commencement Date                 | 1.1(a)     |
| Company                           | Preamble   |
| Company Balance Sheet             | 3.6(a)     |
| Company Board of Directors        | Recitals   |
| Company Board Recommendation      | 3.20(a)    |
| Company Disclosure Schedule       | Section 3  |
| Company Joint Venture             | 3.4(c)     |
| Company Intellectual Property     | 3.12(a)    |
| Company s knowledge               | 9.11(b)    |
| Company Material Adverse Effect   | 3.1(a)     |
| Company Option                    | 2.3(a)     |
| Company Preferred Stock           | 3.3(a)     |
| Company Restricted Shares         | 2.3(b)     |
| Company SEC Reports               | 3.5        |
| Company Shareholder Approval      | 6.1        |
| Company Stock Plans               | 2.3(a)     |
| Company Subsidiary                | 3.4(a)     |
| Confidentiality Agreement         | 1.2(c)     |
| Continuing Employees              | 6.12(a)    |
| Current D&O Insurance             | 6.9(b)     |
| Delisting Period                  | 6.3        |
| Department                        | 1.4        |
| DGCL                              | Recitals   |
| DSOS                              | 1.3(c)     |
| Effective Time                    | 1.4        |
| Environmental Laws                | 3.18(e)(i) |
|                                   |            |

| ERISA                         | 3.16(a)     |
|-------------------------------|-------------|
| ERISA Affiliate               | 3.16(a)     |
| Exchange Act                  | 1.1(a)      |
| Exchange Fund                 | 2.2(a)      |
| Expiration Date               | 1.1(a)      |
| FCPA                          | 3.9(c)      |
| FDA                           | 3.9(d)      |
| Financial Advisor             | 3.21(a)     |
| Good Clinical Practices       | 3.9(g)      |
| Good Laboratory Practices     | 3.9(g)      |
| Good Manufacturing Practices  | 3.9(h)      |
| Governmental Entity           | 9.11(b)     |
| Governmental Antitrust Entity | 6.6(b)      |
| Hazardous Materials           | 3.18(e)(ii) |
| HSR Act                       | 3.19        |
| Indemnified Parties           | 6.9(a)      |
| Independent Directors         | 6.10(a)     |
| Initial Expiration Date       | 1.1(b)      |
| IRS                           | 3.16(c)     |
| Intellectual Property         | 3.12(a)     |
| Laws                          | 3.9(b)      |
| made available                | 9.11(b)     |
| Material Contract             | 3.11(a)     |
| Maximum Premium               | 6.10(b)     |
| MBCA                          | Recitals    |
| Merger                        | 1.3(a)      |
| Merger Agreement              | Annex I     |
| Merger Consideration          | 2.1(c)      |
| Minimum Condition             | Annex I     |
| Notice Period                 | 5.2(d)      |
| OECD Convention               | 3.9(c)      |
| Offer                         | Recitals    |
| Offer Conditions              | 1.1(a)      |
| Offer Documents               | 1.1(d)      |
| Offer Price                   | Recitals    |
| Offer to Purchase             | 1.1(a)      |
| on a fully diluted basis      | 9.11(b)     |
| Outside Date                  | 8.1(b)(ii)  |
| Parent                        | Preamble    |
| Paying Agent                  | 2.2(a)      |
| Permits                       | 3.9(a)      |
| Permitted Lien                | 3.12(a)     |
| person                        | 9.11(b)     |
| Plans                         | 3.16(a)     |
| Post-Closing SEC Reports      | 6.3         |
| Prohibited Payment            | 3.9(c)      |

| Proxy Statement                  | 1.8(a)(ii)   |
|----------------------------------|--------------|
| Real Property                    | 3.12(b)      |
| Registered Intellectual Property | 3.12(a)      |
| Regulation M-A                   | 1.1(d)       |
| Release                          | 3.18(e)(iii) |
| Reporting Tail Endorsement       | 6.9(b)       |
| Representatives                  | 5.2(a)       |
| Sarbanes-Oxley Act               | 3.6(b)       |
| Schedule 14D-9                   | 1.2(b)       |
| Schedule TO                      | 1.1(d)       |
| SEC                              | 1.1(b)       |
| Section 409A                     | 3.16(h)      |
| Securities Act                   | 3.11(c)      |
| Shares                           | Recitals     |
| Special Meeting                  | 1.8(a)(i)    |
| Sub                              | Preamble     |
| Sub Common Stock                 | 2.1          |
| Superior Proposal                | 5.2(d)       |
| Surviving Corporation            | 1.3(a)       |
| Tax                              | 3.15(a)      |
| Taxable                          | 3.15(a)      |
| Tax Return                       | 3.15(a)      |
| Tender Completion Time           | 6.2          |
| Tender and Voting Agreement      | Recitals     |
| Termination Fee                  | 8.2(b)       |
| Top-Up Option                    | 1.10(a)      |
| Top-Up Shares                    | 1.10(a)      |

#### AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER (this <u>Agreement</u>) dated as of June 16, 2010 is among United States Surgical Corporation (<u>Parent</u>), a Delaware corporation, Covidien DE Corp. (<u>Sub</u>), a newly-formed Delaware corporation and a direct or indirect wholly-owned subsidiary of Parent, and Somanetics Corporation (the <u>Company</u>), a Michigan corporation.

### RECITALS

WHEREAS, Parent and the Board of Directors of each of Sub and the Company have adopted and approved the acquisition of the Company by Parent on the terms and conditions set forth in this Agreement;

WHEREAS, in furtherance of the contemplated acquisition of the Company by Parent: (a) Sub shall commence a cash tender offer (as it may be amended from time to time as permitted by this Agreement, the <u>Offer</u>) to acquire all of the issued and outstanding common shares, par value \$0.01 per share, of the Company (the <u>Shares</u>), for \$25.00 for each Share, net to the seller in cash (such price, or any such higher price per Share as may be paid in the Offer, referred to herein as the <u>Offer Price</u>), without interest and (b) after acquiring Shares pursuant to the Offer, Sub shall merge into the Company upon the terms and subject to the conditions set forth in this Agreement (the merger of Sub into the Company being referred to in this Agreement as the <u>Merger</u>);

WHEREAS, the Board of Directors of the Company has, in light of, and subject to, the terms and conditions of this Agreement, adopted and approved this Agreement and the transactions contemplated hereby, including the Merger (as defined in Section 1.3(a)) following the Offer in accordance with the Michigan Business Corporation Act ( $\underline{MBC}A$ ) and upon the terms and subject to the conditions set forth herein;

WHEREAS, the Board of Directors of Sub has, in light of, and subject to, the terms and conditions of this Agreement, adopted and approved this Agreement and the transactions contemplated hereby, including the Merger following the Offer in accordance with the General Corporation Law of the State of Delaware (the <u>DGC</u>L) and upon the terms and subject to the conditions set forth herein;

WHEREAS, the Board of Directors of the Company (the <u>Company Board of Directors</u>) has, in light of, and subject to, the terms and conditions of this Agreement, determined that the Offer and the Merger are in the best interests of the Company and its shareholders and has resolved to recommend that the holders of Shares accept the Offer and, if necessary, adopt and approve this Agreement and thereby approve the Merger upon the terms and subject to the conditions set forth herein;

WHEREAS, concurrently with the execution and delivery of this Agreement, and as a condition and inducement to Parent entering into this Agreement, the chief executive officer of the Company (the <u>CEO Shareholder</u>) has entered into a tender and voting agreement, dated as of the date hereof, in substantially the form set forth in <u>Annex II</u> hereof, pursuant to which, among other things, the CEO Shareholder has agreed to tender his Shares to Sub in the Offer (the <u>Tender and Voting Agreement</u>);

WHEREAS, in furtherance of such acquisition, the board of directors of Parent has, in light of and subject to the terms and conditions set forth herein (i) determined that each of the Offer and the Merger is in the best interests of Parent and its shareholders, (ii) approved and adopted this Agreement and the transactions contemplated hereby and (iii) resolved to recommend that Parent, as the sole shareholder of Sub, approve this Agreement and the Merger; and

WHEREAS, the Company, Parent and Sub desire to make certain representations, warranties, covenants and agreements in connection with the Offer, the Merger and the other transactions contemplated hereby.

NOW, THEREFORE, in consideration of the foregoing and the respective covenants, agreements, representations and warranties set forth herein, the parties agree as follows:

### SECTION 1 THE OFFER AND THE MERGER

## 1.1 The Offer.

(a) Provided that this Agreement shall not have been terminated in accordance with Section 8.1, Parent shall cause Sub to, and Sub shall, commence (within the meaning of Rule 14d-2 under the Securities Exchange Act of 1934, as amended (together with the rules and regulations promulgated thereunder, the <u>Exchange Act</u>)) the Offer within ten (10) business days following the date hereof (the date on which the Offer is commenced being referred to herein as the Commencement Date ). The obligations of Sub to accept for payment and to pay for any Shares validly tendered and not withdrawn prior to the expiration of the Offer (as it may be extended in accordance with requirements of this Section 1.1(a)) shall be subject only to the conditions set forth in Annex I hereto (the Offer Conditions). Subject to the prior satisfaction or, to the extent permitted, waiver by Parent or Sub of the Offer Conditions, Parent shall cause Sub to, and Sub shall, consummate the Offer in accordance with its terms and accept for payment all Shares validly tendered and not withdrawn pursuant to the Offer and pay the Offer Price in exchange for each such Share promptly following such acceptance in compliance with Rule 14e-1(c) of the Exchange Act (the time at which Sub first accepts any Shares for payment pursuant to the Offer being referred to herein as the Acceptance Time ). The Offer shall be made by means of an offer to purchase (the Offer to Purchase ) that contains the terms set forth in this Agreement and the Offer Conditions. Parent expressly reserves the right to waive any of the Offer Conditions, to increase the Offer Price and to make any other changes in the terms of the Offer; provided, however, that Sub shall not, and Parent shall cause Sub not to, without the prior written consent of the Company, (i) decrease the Offer Price, (ii) change the form of consideration payable in the Offer, (iii) decrease the number of Shares sought in the Offer, (iv) waive or change the Minimum Condition, (v) impose additional conditions to the Offer or amend any of the Offer Conditions so as to broaden the scope of such Offer Condition, (vi) extend the Offer beyond a date that is twenty-one (21) business days after commencement of the Offer or the last extension (in accordance with this Agreement), if any, of the Offer, whichever is later (the Expiration Date ) except as set forth in this Agreement, or (vii) otherwise amend any other term or condition of the Offer in a manner materially adverse to the holders of Shares.

(b) The Offer shall initially be scheduled to expire at 12:01 a.m. on the twenty-first (21st) business day following the Commencement Date (calculated as set forth in Rule 14d-1(g)(3) and Rule 14e-1(a) under the Exchange Act) (the Initial Expiration Date Initial Expiration Initial Expiration Initial Expiration Initial Expiration Initial Expiration Initial Expirati

(c) Subject to the parties respective termination rights set forth in Section 8.1, if, at the time as of which the Offer is scheduled to expire, any Offer Condition has not been satisfied or has not been validly waived and there has not been an Adverse Recommendation Change, then, if so requested by the Company by written notice at least two (2) business days prior to the date the Offer is then scheduled to expire, Sub shall extend the Offer for up to two (2) successive periods of not more than ten (10) business days per extension period, until all of the Offer Conditions have been satisfied or, to the extent permitted, validly waived.

(d) On the Commencement Date, Parent and Sub shall file with the SEC, pursuant to Regulation M-A under the Exchange Act (Regulation M-A), a Tender Offer Statement on Schedule TO with respect to the Offer (together with all amendments, supplements and exhibits thereto, the Schedule TO). The Schedule TO shall include, as exhibits, the Offer to Purchase and a related form of letter of transmittal and summary advertisement (collectively, together with any amendments and supplements thereto, the Offer Documents). Subject to Section 5.2, the Company hereby consents to the inclusion in the Offer Documents of the Company Board Recommendation referred to in Section 3.20(a). Parent and Sub shall take all steps necessary to cause the Offer Documents to be filed with the SEC and, subject to the Company s compliance with Section 1.2(c), disseminate the Offer Documents to holders of Shares, in each case as and to the extent required by applicable Law. Parent and Sub, on the one hand, and the Company, on the other hand, agree to promptly correct any information provided by it for use in the Offer Documents if and to the extent that it shall have become false or misleading in any material respect or as otherwise required by Law. Parent and Sub further agree to take all steps necessary to cause the Offer Documents, as so corrected, to be filed with the SEC and disseminated to holders of Shares, in each case as and to the extent required by applicable Law. The Company shall be given a reasonable opportunity to review and comment on the Offer Documents, Schedule TO and any amendment thereto before they are filed with the

SEC, and Parent and Sub shall include all additions, deletions or changes thereto suggested by the Company and its legal counsel that Parent reasonably determines to be appropriate. In addition, Parent and Sub agree to provide the Company with any comments, whether written or oral, that Parent, Sub or their counsel may receive from time to time prior to the expiration or termination of the Offer, from the SEC or its staff with respect to the Offer Documents, promptly upon receipt of such comments, and any written or oral responses thereto, and the Company shall have the right to consult with Parent, Sub and their counsel prior to responding to any such comments, either in written or oral form, and Parent and Sub shall incorporate in each response those views and comments of the Company and its legal counsel related thereto that Parent reasonably determines to be appropriate.

(e) Parent shall provide or cause to be provided to Sub as promptly as practicable following the Expiration Date and any subsequent offering period, as applicable, all funds necessary to pay for those Shares that have been validly tendered and not withdrawn pursuant to the Offer and that Sub is obligated to accept for payment pursuant to the Offer and permitted to accept for payment under applicable Law.

### 1.2 Company Actions.

(a) Subject to Section 5.2 and to any consents or approvals of the Company s shareholders required under applicable Law, the Company hereby approves of and consents to the Offer, the Merger and the other transactions contemplated hereby.

(b) On the date the Offer Documents are filed with the SEC, the Company shall, in a manner that complies with Rule 14d-9 under the Exchange Act, file with the SEC a Tender Offer Solicitation/Recommendation Statement on Schedule 14D-9 (together with all amendments, supplements and exhibits thereto, the Schedule 14D-9 ) which shall, subject to the provisions of Section 5.2, contain the Company Board Recommendation. The Company further agrees to take commercially reasonable steps to cause the Schedule 14D-9 to be filed with the SEC and disseminated to holders of Shares, in each case as and to the extent required by applicable Law. The Company, on the one hand, and Parent and Sub, on the other hand, agree to promptly correct and supplement any information provided by it for use in the Schedule 14D-9 if and to the extent that it shall have become false or misleading in any material respect or as otherwise required by Law. The Company agrees to take all steps necessary to cause the Schedule 14D-9 as so corrected to be filed with the SEC and disseminated to holders of the Shares, in each case as and to the extent required by applicable Law. Parent and Sub shall be given the opportunity to review and comment on the Schedule 14D-9 and any amendment thereto before filing with the SEC, and the Company shall include all additions, deletions or changes thereto suggested by Parent and its legal counsel that the Company reasonably determines to be appropriate. In addition, the Company agrees to promptly provide Parent and Sub any comments, whether written or oral, that the Company or its counsel may receive from time to time from the SEC or its staff with respect to the Schedule 14D-9 promptly after receipt of such comments, and to consult with Parent, Sub and their counsel prior to responding to any such comments, either in written or oral form, and the Company shall incorporate in each response those views and comments of Parent and its legal counsel related thereto that the Company reasonably determines to be appropriate. Parent and Sub shall promptly furnish to the Company all information concerning Parent, Sub and the Offer that may be required or reasonably requested in connection with any action contemplated by this Section 1.2(b).

(c) The Company shall promptly furnish or cause to be furnished to Parent or Sub mailing labels, security position listings and all available listings and computer files containing the names and addresses of the record holders of the Shares as of a recent date, and of those persons becoming record holders subsequent to such date, and shall promptly furnish Parent or Sub with such information and assistance (including, but not limited to, lists of holders of the Shares, updated periodically, and their addresses, mailing labels and lists of security positions) as Parent or Sub or its agent(s) may reasonably request. Except as may be required by applicable Law, such information shall be held confidential by Parent and Sub under the terms of the Confidentiality and Standstill Agreement, dated March 16, 2010 entered into between Parent and the Company (as amended, the Confidentiality Agreement).

#### 1.3 The Merger.

- (a) Subject to the terms and conditions of this Agreement, at the Effective Time (as defined in Section 1.4), the Company and Sub shall consummate the Merger in accordance with the MBCA and the DGCL pursuant to which (i) Sub will be merged with and into the Company and the separate corporate existence of Sub will thereupon cease; (ii) the Company will be the successor or surviving corporation in the Merger and will continue to be governed by the Laws of the State of Michigan; (iii) the separate corporate existence of the Company with all its rights, privileges, immunities, powers and franchises will continue unaffected by the Merger; and (iv) the Company will succeed to and assume all the rights and obligations of Sub. The corporation surviving the Merger is sometimes hereinafter referred to as the <a href="Surviving Corporation">Surviving Corporation</a>. The Merger shall have the effects set forth in the MBCA and the DGCL. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time, all the property, rights, privileges, powers and franchises of the Company and Sub shall be vested in the Surviving Corporation, and all debts, liabilities and duties of the Company and Sub shall become the debts, liabilities and duties of the Surviving Corporation.
- (b) The articles of incorporation of the Company immediately prior to the Effective Time shall, from and after the Effective Time, by virtue of the Merger, be the articles of incorporation of the Surviving Corporation until thereafter changed or amended as provided therein or by applicable Law.
- (c) The bylaws of the Company immediately prior to the Effective Time shall, from and after the Effective Time, without any further action on the part of the Company or Sub, be the bylaws of the Surviving Corporation until thereafter changed or amended as provided by the MBCA, the articles of incorporation of the Surviving Corporation and such bylaws.
- 1.4 Effective Time. Parent, Sub and the Company shall cause an appropriate certificate of merger (the <u>Certificate of Merger</u>) to be executed and filed on the Closing Date (as defined in Section 1.5) (or on such other date as Parent and the Company may agree) with the Department of Energy, Labor and Economic Growth of the State of Michigan (the <u>Department</u>) as provided in the MBCA and with the Secretary of State of the State of Delaware (the <u>DSOS</u>) as provided in the DGCL. The Merger shall become effective on the

later of (A) such time and date on which the Certificate of Merger has been duly filed with the Department and (B) such time and date on which the Certificate of Merger has been duly filed with the DSOS, or such later time and date as is mutually agreed upon by Company and Sub and specified in the Certificate of Merger, such time hereinafter referred to as the <u>Effective Time</u>.

- 1.5 <u>Closing</u>. The closing of the Merger (the <u>Closing</u>) will take place at 9:00 a.m. (Boston time) on a date to be specified by the parties, such date to be no later than the second (2nd) business day after satisfaction or, to the extent permitted, waiver of all of the conditions set forth in Section 7 capable of satisfaction or waiver prior to Closing (the <u>Closing Date</u>), at the offices of Goodwin Procter LLP, 53 State Street, Boston, Massachusetts 02109, unless another date or place is agreed to in writing by the parties hereto.
- 1.6 <u>Directors and Officers of the Surviving Corporation</u>. The directors of Sub immediately prior to the Effective Time shall, from and after the Effective Time, be the directors of the Surviving Corporation, and the officers of Sub immediately prior to the Effective Time shall, from and after the Effective Time, be the officers of the Surviving Corporation, in each case until their respective successors shall have been duly elected, designated or qualified, or until their earlier death, resignation or removal in accordance with the Surviving Corporation s articles of incorporation and bylaws. Prior to the Effective Time, the Company shall use commercially reasonable best efforts to cause each member of the Company Board of Directors, other than Parent s or Sub s designees pursuant to Section 6.10, to execute and deliver a letter effectuating his or her resignation as a director of the Company upon the Effective Time.
- 1.7 Subsequent Actions. If at any time after the Effective Time the Surviving Corporation shall determine, in its sole discretion, or shall be advised, that any deeds, bills of sale, assignments, assurances or any other actions or things are necessary or desirable to vest, perfect or confirm of record or otherwise in the Surviving Corporation its right, title or interest in, to or under any of the rights, properties or assets of either of the Company or Sub acquired or to be acquired by the Surviving Corporation as a result of, or in connection with, the Merger or otherwise to carry out this Agreement, then the officers and directors of the Surviving Corporation shall be authorized to execute and deliver, in the name and on behalf of either the Company or Sub, all such deeds, bills of sale, instruments of conveyance, assignments and assurances and to take and do, in the name and on behalf of each such corporation or otherwise, all such other actions and things as may be necessary or desirable to vest, perfect or confirm any and all right, title or interest in, to and under such rights, properties or assets in the Surviving Corporation or otherwise to carry out this Agreement.

### 1.8 Shareholders Meeting.

- (a) If required by applicable Law in order to consummate the Merger, the Company, acting through the Company Board of Directors, in accordance with applicable Law and the Company s articles of incorporation and bylaws, shall, as soon as reasonably practicable following the Tender Completion Time:
- (i) duly call, give notice of, convene and hold a special meeting of its shareholders to consider the adoption and approval of this Agreement and the approval of the Merger (the <u>Special Meeting</u>);

- (ii) prepare and file with the SEC under the Exchange Act a preliminary proxy or information statement relating to the Merger and this Agreement and use its commercially reasonable best efforts to obtain and furnish the information required to be included by the SEC in the Proxy Statement (as hereinafter defined) and, after Parent shall have had a reasonable opportunity to review and comment on the Proxy Statement, respond promptly to any comments made by the SEC with respect to the preliminary proxy or information statement and cause a definitive proxy or information statement (in either case, the <a href="Proxy Statement">Proxy Statement</a>) to be mailed to its shareholders as promptly as reasonably practicable;
- (iii) subject to Section 5.2, include in the Proxy Statement the recommendation of the Company Board of Directors that shareholders of the Company vote in favor of the adoption and approval of this Agreement and the approval of the Merger; and
- (iv) subject to Section 5.2, use its commercially reasonable best efforts to solicit from holders of Shares proxies in favor of the adoption and approval of this Agreement and the approval of the Merger and take all other action reasonably necessary or advisable to secure the approval of shareholders required by the MBCA and any other applicable Law and the Company s articles of incorporation and bylaws (if applicable) to effect the Merger.
- (b) Parent agrees to vote, or cause to be voted, all of the Shares then beneficially owned by it, Sub or any of their respective Affiliates in favor of the adoption and approval of this Agreement and the approval of the Merger.
- 1.9 Merger Without Meeting of Shareholders. Notwithstanding Section 1.8, in the event that Parent, Sub or any other subsidiary of Parent shall acquire at least ninety percent (90%) of the outstanding shares of each class of capital stock of the Company entitled to vote on the Merger, pursuant to the Offer or otherwise, the parties hereto agree, at the request of Parent and subject to Section 7, to take all necessary and appropriate action to cause the Merger to become effective as soon as practicable after such acquisition, without a meeting of shareholders of the Company, in accordance with and subject to the MBCA.

### 1.10 Top-Up Option.

(a) The Company hereby grants to Sub an irrevocable option (the <u>Top-Up Option</u>), exercisable only on the terms and subject to the conditions set forth in this Agreement, to purchase at a price per share equal to the Offer Price paid in the Offer that number of newly issued Shares (the <u>Top-Up Shares</u>) equal to the lowest number of Shares that, when added to the number of Shares directly or indirectly owned by Parent or Sub at the time of exercise of the Top-Up Option, shall constitute one share more than ninety percent (90%) of the Shares outstanding immediately after the issuance of the Top-Up Shares (determined on a fully diluted basis); <u>provided, however</u>, that (i) the Top-Up Option shall not be exercisable for a number of Shares in excess of the Shares authorized and unissued at the time of exercise of the Top-Up Option and (ii) the Top-Up Option may not be exercised unless, (x) following the Acceptance Time or after a subsequent offering period, eighty percent (80%) or more of the Shares shall be directly or indirectly owned by Parent or Sub and (y) after the giving effect to the exercise of the Top-Up Option, Parent, Sub and any wholly owned subsidiary of Parent or Sub would own in the aggregate one share more than ninety percent (90%) of the number of outstanding Shares (after

giving effect to the issuance of the Top-Up Option but excluding Shares tendered pursuant to guaranteed delivery procedures that have not yet been delivered in settlement or satisfaction of such guarantee). The Top-Up Option shall be exercisable only once at any time following the Acceptance Time and prior to the earlier to occur of (A) the Effective Time and (B) the termination of this Agreement in accordance with its terms; provided, however, that so long as the Minimum Condition is satisfied upon the Acceptance Time, Parent and Sub may exercise the Top-Up Option, on a second occasion, on the same terms and conditions set forth in this Section 1.10 in the event the Shares that were subject to guarantees of delivery were not properly tendered in the Offer.

- (b) If Sub wishes to exercise the Top-Up Option, Sub shall give the Company one (1) business day prior written notice, specifying (i) the number of Shares directly or indirectly owned by Parent and Sub at the time of such notice, (ii) a place and a time for the closing of such purchase, (iii) the number of Shares Parent or Sub intends to purchase pursuant to the Top-Up Option and (iv) manner in which Parent or Sub intends to pay the applicable purchase price. The aggregate purchase price payable for the Shares purchased by Sub or Parent pursuant to the Top-Up Option shall be determined by multiplying the number of such Shares by the Offer Price. The Company shall, as soon as practicable following receipt of such notice, deliver written notice to Sub specifying, based on the information provided by Sub in its notice, the number of Top-Up Shares. At the closing of the purchase of Top-Up Shares, the purchase price owed by Sub to the Company therefor shall be paid to the Company (A) by wire transfer or cashier s check or (B) by issuance by Sub to the Company of a promissory note, which shall bear interest at three percent (3%) per annum, shall mature on the first anniversary of the date of execution and delivery of such promissory note and may be prepaid, in whole or in part, without premium or penalty.
- (c) Parent and Sub acknowledge that the Shares that Sub may acquire upon exercise of the Top-Up Option shall not be registered under the Securities Act and shall be issued in reliance upon an exemption for transactions not involving a public offering. Parent and Sub represent and warrant to the Company that Sub is, or shall be upon any purchase of Top-Up Shares, an accredited investor, as defined in Rule 501 of Regulation D under the Securities Act. Sub agrees that the Top-Up Option, and the Top-Up Shares to be acquired upon exercise of the Top-Up Option, if any, are being and shall be acquired by Sub for the purpose of investment and not with a view to, or for resale in connection with, any distribution thereof (within the meaning of the Securities Act). Any certificates evidencing the shares acquired pursuant to the Top-Up Option shall include any legends required by applicable securities laws.

#### SECTION 2 CONVERSION OF SECURITIES

- **2.1** Conversion of Capital Stock. As of the Effective Time, by virtue of the Merger and without any action on the part of the holders of any Shares or any common shares, par value \$0.01 per share, of Sub (<u>Sub Common Stock</u>):
- (a) <u>Sub Common Stock</u>. Each issued and outstanding share of Sub Common Stock shall be converted into and become one (1) fully paid and nonassessable common share of the Surviving Corporation.

- (b) <u>Cancellation of Treasury Stock and Parent-Owned Stock</u>. All Shares that are owned by the Company as treasury stock and any Shares owned by Parent or Sub shall automatically be cancelled and retired and shall cease to exist, and no consideration shall be payable in exchange therefor.
- (c) Conversion of Shares. Each issued and outstanding Share (other than Shares to be cancelled in accordance with Section 2.1(b)) shall be converted into the right to receive the Offer Price, payable to the holder thereof in cash, without interest (the Merger Consideration ). From and after the Effective Time, all such Shares shall no longer be outstanding and shall automatically be cancelled and retired and shall cease to exist, and each holder of a certificate (a Certificate ) or book-entry share (a Book-Entry Share ) representing any such Shares shall cease to have any rights with respect thereto, except the right to receive the Merger Consideration therefor, without interest thereon, upon the surrender of such certificate or book-entry share in accordance with Section 2.2.
- (d) Any right of repurchase, risk of forfeiture or other condition under any restricted stock agreement or other agreement with the Company with respect to Shares outstanding immediately effective prior to the Effective Time shall lapse and any vesting thereon shall fully accelerate, in each case, as of immediately prior to the Acceptance Time.

### 2.2 Exchange of Certificates.

(a) Paying Agent. Parent shall designate a bank or trust company that is reasonably acceptable to the Company to act as agent for the holders of Shares in connection with the Merger (the Paying Agent) and to receive the funds to which holders of Shares shall become entitled pursuant to Section 2.1(c) or Section 2.3(b). Parent shall cause the Surviving Corporation to provide to the Paying Agent on a timely basis, promptly after the Effective Time and as and when needed after the Effective Time, cash necessary to pay for the Shares converted into the right to receive the Merger Consideration (such cash being hereinafter referred to as the <u>Exchange Fund</u>). The Exchange Fund shall not be used for any other purpose. If for any reason the Exchange Fund is inadequate to pay the amounts to which holders of Shares shall be entitled under Section 2.1(c), Parent shall promptly deposit or cause the Surviving Corporation promptly to deposit additional cash with the Paying Agent sufficient to make all payments of Merger Consideration, and Parent and the Surviving Corporation shall in any event be liable for payment thereof. The Paying Agent may invest the cash in the Exchange Fund as reasonably directed by Parent; provided, however, that such investments shall be in obligations of or guaranteed by the United States or any agency or instrumentality thereof and backed by the full faith and credit of the United States, in commercial paper obligations rated P-1 or A-1 or better by Moody's Investors Service, Inc. or Standard & Poor's Corporation, respectively, in money market funds that invest only in such United States government and commercial paper obligations, or in certificates of deposit, bank repurchase agreements or banker s acceptances of commercial banks with capital exceeding \$1 billion (based on the most recent financial statements of such bank that are then publicly available). Any interest and other income resulting from such investments shall be paid to Parent. No investment of the Exchange Fund shall relieve Parent, Surviving Corporation or the Paying Agent from promptly making the payments required under this Article II, and following any losses from such investment, Parent shall promptly provide any additional cash funds to the Paying Agent for the benefit of the Company s shareholders at the Effective Time in the amount of such losses, which additional funds will be deemed to be part of the Exchange Fund.

- (b) Exchange Procedures. Promptly after the Effective Time, Parent shall cause the Paying Agent to mail to each holder of record of a Certificate or a Book-Entry Share, which immediately prior to the Effective Time represented outstanding Shares, whose Shares were converted pursuant to Section 2.1(c) into the right to receive the Merger Consideration (i) a letter of transmittal (which shall specify that delivery shall be effected, and risk of loss and title to the Certificates or Book-Entry Shares, as applicable, shall pass, only upon delivery of the Certificates (or affidavits of loss of such Certificates) or Book-Entry Shares to the Paying Agent and shall be in such form and have such other provisions as Parent and the Company may reasonably agree upon prior to the Effective Time); and (ii) instructions for effecting the surrender of the Certificates or Book-Entry Shares in exchange for payment of the Merger Consideration. Upon surrender of a Certificate or Book-Entry Share, as applicable, for cancellation to the Paying Agent or to such other agent or agents as may be appointed by Parent, together with such letter of transmittal, duly executed and properly completed and such other documents as may be reasonably requested by the Paying Agent, the holder of such Certificate or Book-Entry Share shall be entitled to receive in exchange therefor the Merger Consideration for each Share, formerly represented by such Certificate or Book-Entry Share, and the Certificate or Book-Entry Share so surrendered shall forthwith be cancelled. Until surrendered as contemplated by this Section 2.2, each Certificate or Book-Entry Share, as applicable, shall be deemed at any time after the Effective Time to represent only the right to receive the Merger Consideration in cash as contemplated by this Section 2.2, without interest thereon, and shall not evidence any interest in, or any right to exercise the rights of a shareholder or other equity holder of, the Company or the Surviving Corporation.
- (c) <u>Transfer Books</u>; <u>No Further Ownership Rights in Shares</u>. At the Effective Time, the stock transfer books of the Company shall be closed and thereafter there shall be no further registration of transfers of Shares on the records of the Company. From and after the Effective Time, the holders of Certificates or Book-Entry Shares evidencing ownership of Shares outstanding immediately prior to the Effective Time shall cease to have any rights with respect to such Shares, except as otherwise provided for herein or by applicable Law. If, after the Effective Time, Certificates or Book-Entry Shares are presented to the Surviving Corporation for any reason, they shall be cancelled and exchanged as provided in this Section 2, except as otherwise provided by Law.
- (d) Termination of Exchange Fund; No Liability. At any time following nine (9) months after the Effective Time, the Surviving Corporation shall be entitled to require the Paying Agent to deliver to it any funds (including any interest received with respect thereto) made available to the Paying Agent and not disbursed (or for which disbursement is pending subject only to the Paying Agent s routine administrative procedures) to holders of Certificates, and thereafter such holders shall be entitled to look only to Parent or the Surviving Corporation (subject to abandoned property, escheat or other similar Laws) only as general creditors thereof with respect to the Merger Consideration payable upon due surrender of their Certificates or Book-Entry Shares, as applicable, without any interest thereon. Notwithstanding the foregoing, none of Parent, the Surviving Corporation nor the Paying Agent shall be liable to any holder of a Certificate or Book-Entry Share, as applicable, for Merger Consideration delivered to a public

official pursuant to any applicable abandoned property, escheat or similar Law. If Certificates or Book-Entry Shares, as applicable, are not surrendered prior to three (3) years after the Effective Time, unclaimed Merger Consideration payable with respect to such Shares shall, to the extent permitted by applicable Law, become the property of the Surviving Corporation, free and clear of all claims or interest of any person previously entitled thereto.

(e) <u>Lost Certificates</u>. If any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming such Certificate to be lost, stolen or destroyed and, if required by Parent, the posting by such person of a bond in such amount as Parent may reasonably direct as indemnity against any claim that may be made against it or the Surviving Corporation with respect to such Certificate, the Paying Agent shall issue in exchange for such lost, stolen or destroyed Certificate the applicable Merger Consideration with respect thereto.

### 2.3 Company Stock Plans.

- (a) Effective as of the Effective Time, each outstanding stock option, stock equivalent right or right to acquire Shares (each a Company Option and collectively, the Company Options) granted under the Company s 1991 Incentive Stock Option Plan, 1997 Stock Option Plan or 2005 Stock Incentive Plan (the Company Stock Plans) or granted outside of a plan, without regard to the extent then vested and exercisable, shall be cancelled and, in consideration of such cancellation, Parent shall, or shall cause the Surviving Corporation to, promptly following the Effective Time, pay to such holders of Company Options, an amount in respect thereof equal to the product of (x) the excess, if any, of the Offer Price over the exercise price of each such Company Option and (y) the number of unexercised Shares subject thereto (such payment, if any, to be net of applicable Taxes withheld pursuant to Section 2.5).
- (b) Effective as of the Effective Time, restrictions on each restricted stock award (each a <u>Company Restricted Share</u> and collectively, the <u>Company Restricted Shares</u>) granted under any Company Stock Plan, which is outstanding immediately prior to the Effective Time, will lapse as of such Effective Time and such Company Restricted Shares shall be treated in the same manner as other Shares pursuant to Section 2.1(c) (except that with respect to any restrictions which by the terms provide for a lapse to a lesser extent upon the consummation of the Merger, such restrictions shall only lapse to such lesser extent) (such payment, if any, to be net of applicable Taxes withheld pursuant to Section 2.5).
- (c) As of the Effective Time, the Company Stock Plans shall terminate and all rights under any provision of any other plan, program or arrangement providing for the issuance or grant of any other interest in respect of the capital stock of the Company, or any Company Subsidiary (as defined in Section 3.4(a)), shall be cancelled. The Company shall ensure that, after the Effective Time, no person shall have any right under the Company Stock Plans to acquire any capital stock of the Company or the Surviving Corporation or any other equity interest therein (including phantom stock or stock appreciation rights), except as set forth herein. The Company Board of Directors (or, if appropriate, any committee thereof administering the Company Stock Plans) shall adopt such resolutions, or take such other actions as may be required, to effect the foregoing.

- **2.4** Section 16. The Company Board of Directors shall, to the extent necessary, take appropriate action, prior to or as of the Acceptance Time, to approve, for purposes of Section 16(b) of the Exchange Act, the disposition and cancellation of Shares (including derivative securities with respect to Shares) resulting from the transactions contemplated by this Agreement.
- 2.5 Withholding. Each of Parent and Surviving Corporation shall be entitled to deduct and withhold, or cause the Paying Agent to deduct and withhold, from any amounts payable or otherwise deliverable pursuant to this Agreement to any holder or former holder of Shares, Company Options or Company Restricted Shares such amounts as are required to be deducted and withheld therefrom under the United States Internal Revenue Code of 1986, as amended (the <u>Code</u>) or the Treasury Regulations thereunder or any other Tax Law. To the extent such amounts are so deducted and withheld, such amounts shall be treated for all purposes under this Agreement as having been paid to the person to whom such amounts would otherwise have been paid.
- 2.6 <u>Transfer Taxes</u>. If any payment pursuant to the Offer or the Merger is to be made to a person other than the person in whose name the surrendered Certificate or Book-Entry Share, as applicable, is registered, it shall be a condition of payment that the Certificate or Book-Entry Share, as applicable, so surrendered shall be properly endorsed or shall be otherwise in proper form for transfer and that the person requesting such payment shall have paid all transfer and other Taxes required by reason of the issuance to a person other than the registered holder of the Certificate or Book-Entry Share, as applicable, surrendered or shall have established to the satisfaction of Parent that such Tax either has been paid or is not applicable. Any other transfer Taxes shall be paid by Parent.

### SECTION 3 REPRESENTATIONS AND WARRANTIES OF COMPANY

Except as set forth on the disclosure schedule delivered by the Company to Parent on the date hereof (the <u>Company Disclosure Schedule</u>), the Section numbers of which are numbered to correspond to the Section numbers of this Agreement to which they refer (<u>provided</u>, <u>however</u>, that (i) an item disclosed in any Section shall be deemed to have been disclosed for each other Section of this Agreement to the extent the relevance of such disclosure to such other Section of this Agreement is reasonably apparent on the face of such disclosure and (ii) the mere inclusion of an item in such Company Disclosure Schedule as an exception to a representation or warranty shall not be deemed an admission that such item represents a material exception or material fact, event or circumstance or that such item constitutes a Company Material Adverse Effect), the Company hereby makes the following representations and warranties to, and agreements with, Parent and Sub:

### 3.1 Organization and Qualification.

(a) The Company is a corporation duly incorporated, validly existing and in good standing under the Laws of its jurisdiction of incorporation and has corporate power and authority to own, lease and operate its assets and to carry on its business as now being conducted. The Company is qualified to transact business as a foreign corporation in all jurisdictions in which such qualification is required by Law, except for jurisdictions in which the

failure to be so qualified and in good standing does not constitute a Company Material Adverse Effect. Company Material Adverse Effect shall mean any change, event, circumstance, effect or development that, individually or in the aggregate with all other changes, events, circumstances, effects or developments that exist on the date of determination of the occurrence of a Company Material Adverse Effect, has had or is reasonably likely to have a material adverse effect on (i) the assets, properties, business, capitalization, results of operations or condition (financial or other) of the Company and the Company Subsidiaries, taken as a whole or (ii) the ability of the Company to consummate the transactions contemplated by this Agreement; provided, however, that in no event shall changes, events, circumstances, effects or developments to the extent resulting from any of the following be taken into account in determining whether there is, has been or is reasonably likely to be a Company Material Adverse Effect: (A) changes in conditions of the economy or securities markets of the United States, Europe or Japan in general that in each case, do not have a disproportionate and adverse impact on the Company and the Company Subsidiaries, taken as a whole, relative to other persons engaged in business in the medical device industry, (B) changes in conditions affecting the medical device industry, in each case, without a disproportionate and adverse impact on the Company and the Company Subsidiaries, taken as a whole, relative to other persons engaged in business in the medical device industry, (C) changes to applicable Law or generally accepted accounting principles or, in either case, the interpretation thereof that do not have a disproportionate and adverse impact on the Company and the Company Subsidiaries, taken as a whole, relative to other persons engaged in business in the medical device industry, (D) any change in the trading price or trading volume of the Shares (it being understood that the underlying facts or circumstances giving rise to any such change may be taken into account in determining whether there has been or is likely to be a Company Material Adverse Effect if such facts and circumstances are not otherwise excluded pursuant to clauses (A) through (G) of this definition), (E) any other action required by Law, this Agreement or taken at the request or with the consent of Parent, Sub or any of their Affiliates, (F) the announcement of the execution of this Agreement or the pendency of the transactions contemplated hereby, or (G) any failure of the Company to meet securities analysts published or internal projections or forecasts or estimates of earnings or revenues (it being understood that the underlying facts or circumstances giving rise to any such failure may be taken into account in determining whether there has been or is likely to be a Company Material Adverse Effect if such facts and circumstances are not otherwise excluded pursuant to clauses (A) through (G) of this definition); provided further that in the case of clauses A, B and C, only the incremental adverse disproportionate impact, effect or consequence may be taken into account when determining whether a Company Material Adverse Effect has occurred or is reasonably likely to occur.

(b) The Company has previously provided to Parent true and complete copies of the articles of incorporation and bylaws or other organizational documents of the Company and each Company Subsidiary as presently in effect, and none of the Company or any Company Subsidiary is in default in the performance, observation or fulfillment of such documents, except, in the case of the Company Subsidiaries, such defaults that, in the aggregate, does not constitute a Company Material Adverse Effect.

3.2 <u>Authority to Execute and Perform Agreement</u>. The Company has the corporate power and authority to enter into, execute and deliver this Agreement and, subject, in the case of consummation of the Merger, to the adoption and approval of this Agreement by the holders of

the Shares, to perform fully its obligations hereunder. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly authorized by the Company Board of Directors. No other corporate action on the part of the Company is necessary to consummate the transactions contemplated hereby (other than adoption and approval of this Agreement by the holders of the Shares and the filing of a certificate of merger or other appropriate document with the Department and the DSOS). This Agreement has been duly executed and delivered by the Company and (assuming due authorization, execution and delivery by Parent and Sub) constitutes a valid and binding obligation of the Company, enforceable in accordance with its terms, except to the extent enforceability may be limited by the effect of applicable bankruptcy, reorganization, insolvency, moratorium or other Laws affecting the enforcement of creditors rights generally and the effect of general principles of equity, regardless of whether such enforceability is considered in a proceeding at Law or in equity.

#### 3.3 Capitalization.

- (a) The authorized capital stock of the Company as of the date of this Agreement consists of 20,000,000 Shares and 1,000,000 preferred shares, par value \$0.01 per share (<u>Company Preferred Stock</u>). The rights and privileges of each class of the Company s capital stock are as set forth in the Company s articles of incorporation. As of the date of this Agreement, (i) 11,953,384 Shares were issued and outstanding and (ii) no shares of Company Preferred Stock were issued or outstanding.
- (b) Section 3.3(b) of the Company Disclosure Schedule includes a list, as of the date of this Agreement, of (i) each outstanding Company Option under the Company Stock Plans or issued independent of such plans, including the name of the applicable holder, the Company Stock Plan under which each Company Option is granted, the grant date and acceleration provisions, the expiration date, the exercise price, and whether any option is an incentive stock option, (ii) the total number of Shares issued under each Company Stock Plan, (iii) the total number of Shares reserved for future issuance under each Company Stock Plan and (iv) each grant of Shares that are subject to repurchase by the Company pursuant to Company Restricted Shares, including the name of the applicable holder, the Company Stock Plan under which such Company Restricted Shares were issued, the issue date, the applicable vesting and acceleration provisions, any performance targets or market conditions related to vesting and the repurchase price relating to each grant of Company Restricted Shares. The Company Stock Plans (including all amendments) have been duly approved by the Company shareholders. All outstanding Company Options were granted with an exercise price not less than the fair market value of the Shares on the date of grant. The Company has made available to Parent complete and accurate copies of all (x) Company Stock Plans, (y) forms of stock option agreements evidencing Company Options and (z) forms of agreements evidencing Company Restricted Shares.
- (c) Except as set forth in Section 3.3(a) of this Agreement, Sections 3.3(b) or 3.3(c) of the Company Disclosure Schedule or Schedule 5.1 of this Agreement, (i) there are not as of the date of this Agreement, and at the Acceptance Time there will not be, any equity securities of any class of the Company, or any security exchangeable into or exercisable for such equity securities, issued, reserved for issuance or outstanding and (ii) there are not as of the date

of this Agreement, and at the Acceptance Time there will not be, any options, warrants, equity securities, calls, rights, commitments or agreements to which the Company or any of its Subsidiaries is a party or by which the Company or any of its Subsidiaries is bound obligating the Company or any of its subsidiaries to issue, exchange, transfer, deliver, sell or cause to be issued, exchanged, transferred, delivered or sold, additional shares of capital stock or other equity or voting interests of the Company or any security or rights convertible into or exchangeable or exercisable for any such shares or other equity or voting interests, or obligating the Company or any of its subsidiaries to grant, extend, accelerate the vesting of, otherwise modify or amend or enter into any such option, warrant, equity security, call, right, commitment or agreement, other than the Top-Up Option. The Company does not have any outstanding stock appreciation rights, restricted stock units, phantom stock, performance based rights or similar rights or obligations. None of the Company or, to the Company s knowledge, any of its Affiliates, is a party to or is bound by any agreement with respect to the voting (including proxies) or sale or transfer of any shares of capital stock or other equity or voting interests of the Company. For all purposes of this Agreement, the term Affiliate when used with respect to any person means any other person who is an affiliate of that first person within the meaning of Rule 405 under the Securities Act. Except as contemplated by this Agreement and except to the extent arising pursuant to applicable state takeover or similar Laws, there are no registration rights, and there is no rights agreement, poison pill anti-takeover plan or other similar agreement to which the Company or any Company Subsidiary is bound with respect to any securities of the Company.

- (d) All outstanding Shares are, and all Shares subject to issuance as specified in Section 3.3(b) above, upon issuance on the terms and conditions specified in the instruments pursuant to which they are issuable, will be, duly authorized, validly issued, fully paid and nonassessable and not subject to or issued in violation of any purchase option, call option, right of first refusal, preemptive right, subscription right or any similar right under any provision of the MBCA, the Company s articles of incorporation or bylaws or any agreement to which the Company is bound.
- (e) There are no obligations, contingent or otherwise, of the Company or any of its subsidiaries to repurchase, redeem or otherwise acquire any Shares or the capital stock of the Company or any of its subsidiaries. The Company has no outstanding bonds, debentures, notes or other indebtedness that have the right to vote on any matters on which shareholders may vote.

### 3.4 Company Subsidiaries.

(a) Section 3.4(a) of the Company Disclosure Schedule sets forth a true and complete list of the names, jurisdictions of organization and capitalization of each Company Subsidiary and, for the Company and each Company Subsidiary, the jurisdictions in which it is qualified to do business. Section 3.4(a) of the Company Disclosure Schedule also sets forth for each such Company Subsidiary the individuals who comprise the board of directors or comparable body for each such entity. The Company agrees to take, or cause to be taken, the actions necessary so that those individuals will resign and be replaced by individuals specified by Parent effective as of the Effective Time. All issued and outstanding shares or other equity interests of each Company Subsidiary are owned directly by the Company free and clear of any

charges, liens, encumbrances, security interests or adverse claims. As used in this Agreement, <u>Company Subsidiary</u> means any corporation, partnership or other organization, whether incorporated or unincorporated, of which (i) the Company or any Company Subsidiary is a general partner or (ii) at least 50% of the securities or other interests having voting power to elect a majority of the board of directors or others performing similar functions with respect to such corporation, partnership or other organization are directly or indirectly owned or controlled by the Company or by any Company Subsidiary, or by the Company and one or more Company Subsidiaries.

- (b) Each Company Subsidiary is a corporation or B.V. duly organized, validly existing and in good standing (to the extent such concepts are applicable) under the Laws of the jurisdiction of its formation, has all requisite entity power and authority to own, lease and operate its properties and assets and to carry on its business as now being conducted, and is duly qualified to do business and is in good standing as a foreign branch (to the extent such concepts are applicable) in each jurisdiction where the character of its properties owned, operated or leased or the nature of its activities makes such qualification necessary, except for such failures to be so organized, qualified or in good standing, individually or in the aggregate, that do not constitute a Company Material Adverse Effect. There are not as of the date hereof, and at the Effective Time there will not be, any subscriptions, options, conversion or exchange rights, warrants, repurchase or redemption agreements, or other agreements, claims or commitments of any nature whatsoever obligating any Company Subsidiary to issue, transfer, deliver or sell, or cause to be issued, transferred, delivered, sold, repurchased or redeemed, shares of the capital stock or other securities of the Company or any Company Subsidiary or obligating the Company or any Company Subsidiary to grant, extend or enter into any such agreement. To the knowledge of the Company, there are no equity holder agreements, voting trusts, proxies or other agreements, instruments or understandings with respect to the voting of the equity interests of any Company Subsidiary.
- (c) Section 3.4(c) of the Company Disclosure Schedule sets forth, for each Company Joint Venture, the interest held by the Company and the jurisdiction in which such Company Joint Venture is organized. Interests in Company Joint Ventures held by the Company are held directly by the Company, free and clear of any charges, liens, encumbrances, security interests or adverse claims. The term <u>Company Joint Venture</u> means any corporation or other entity (including partnership, limited liability company and other business association) that is not a Company Subsidiary and in which the Company or one or more Company Subsidiaries owns an equity interest (other than equity interests held for passive investment purposes which are less than 5% of any class of the outstanding voting securities or other equity of any such entity).
- (d) The Company does not control, directly or indirectly, any capital stock of any person that is not a Company Subsidiary.
- 3.5 SEC Reports. The Company has filed or furnished (as applicable) all registration statements, forms, reports, certifications and other documents required to be filed by the Company with the SEC since January 1, 2007. All such registration statements, forms, reports and other documents (including those filed or furnished by the Company during such period, whether or not required to be so filed or furnished, and that the Company may file after the date hereof until the Closing) are referred to herein as the <a href="Company SEC Reports">Company SEC Reports</a>. The Company

SEC Reports, and giving effect to any amendments or supplements thereto, (i) were or will be filed on a timely basis, (ii) at the time filed, complied, or will comply when filed, as of each respective filing date as to form in all material respects with the applicable requirements of the Securities Act and the Exchange Act applicable to such Company SEC Reports and (iii) did not or will not at the time they were or are filed contain any untrue statement of a material fact or omit to state a material fact required to be stated in such Company SEC Reports or necessary in order to make the statements in such Company SEC Reports, in the light of the circumstances under which they were made, not misleading in any material respect. No Company Subsidiary is required to file any form, report or other document with the SEC. Section 3.5 of the Company Disclosure Schedule lists all effective registration statements filed by the Company on Form S-3 or Form S-8 or otherwise relying on Rule 415 under the Securities Act.

#### 3.6 Financial Statements.

(a) Each of the consolidated financial statements (including, in each case, any related notes and schedules) contained or to be contained in the Company SEC Reports at the time filed, and giving effect to any amendments or supplements thereto filed prior to the date of this Agreement, (i) complied or will comply as to form in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto, (ii) were or will be prepared in accordance with United States generally accepted accounting principles applied on a consistent basis throughout the periods involved (except as may be indicated in the notes to such financial statements or, in the case of unaudited interim financial statements, as permitted by the SEC on Form 10-Q under the Exchange Act), and (iii) fairly presented or will fairly present in all material respects the consolidated financial position of the Company and the Company Subsidiaries as of the dates indicated and the consolidated results of its operations and cash flows for the periods indicated, consistent with the books and records of the Company and the Company Subsidiaries, except that the unaudited interim financial statements were or are subject to normal and recurring year end adjustments which were or will not be material in amount or effect. The consolidated audited balance sheet of the Company as of November 30, 2009 included in the audited financial statements set forth in the Company s Annual Report on Form 10-K for the year ended November 30, 2009 is referred to herein as the <a href="Company Balance Sheet">Company Balance Sheet</a>.

(b) The Company is in compliance in all material respects with the applicable provisions of the Sarbanes-Oxley Act of 2002, as amended (the Sarbanes-Oxley Act ). Each required form, report and document (including any amendment thereof and supplement thereto) containing financial statements that has been filed with or submitted or will be filed with or submitted to the SEC since January 1, 2007 was or will be accompanied by the certifications required to be filed or submitted by the Company s principal executive officer and principal financial officer pursuant to the Sarbanes-Oxley Act and Rule 13a-14 or 15d-14 promulgated under the Exchange Act and, at the time of filing or submission of each such certification, such certification complied or will comply, in each case in all material respects, with the applicable provisions of the Sarbanes-Oxley Act and Rule 13a-14 or 15d-14 promulgated under the Exchange Act.

- (c) The Company maintains a system of internal accounting controls sufficient to provide reasonable assurance that (i) transactions are executed in accordance with management s general or specific authorizations; (ii) access to assets is permitted only in accordance with management s general or specific authorization; and (iii) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences. The Company maintains disclosure controls and procedures required by Rule 13a-15 or 15d-15 under the Exchange Act. Such disclosure controls and procedures are designed to ensure that all material information concerning the Company is made known on a timely basis to the individuals responsible for the preparation of the Company SEC Reports. Since the date of the filing of the Company s most recent annual report on Form 10-K, prior to the date of this Agreement, the Company s outside auditors and the audit committee of the Company Board have not been advised of (A) any significant deficiencies or material weaknesses in the design or operation of internal control over financial reporting which adversely affect the Company s ability to record, process, summarize and report financial information, and (B) any fraud, whether or not material, that involves management or other employees who have a significant role in the Company s internal control over financial reporting. Any material change in internal control over financial reporting and any significant deficiency or material weakness in the design or operation of internal control over financial reporting required to be disclosed in any Company SEC Report has been so disclosed and each significant deficiency and material weakness previously so disclosed has been remediated. The Company is in compliance in all material respects with the applicable listing and other rules and regulations of The NASDAQ Stock Market.
- (d) The Company is not a party to, or does not have any commitment to become a party to, any joint venture, off-balance sheet partnership or any similar contract (including any contract or arrangement relating to any transaction or relationship between or among the Company, on the one hand, and any unconsolidated affiliate, including any structured finance, special purpose or limited purpose entity or person, on the other hand, or any off-balance sheet arrangements (as defined in Item 303(a) of Regulation S-K under the Exchange Act)), where the result, purpose or intended effect of such contract or arrangement is to avoid disclosure of any material transaction involving, or material liabilities of, the Company in the Company SEC Reports.
- 3.7 <u>Absence of Undisclosed Liabilities</u>. The Company and the Company Subsidiaries do not have any liabilities of a type required to be reflected or disclosed on the consolidated balance sheet of the Company (including the notes thereto) prepared in accordance with United States generally accepted accounting principles, other than liabilities (a) adequately reflected or reserved against on the Company Balance Sheet, (b) included in Section 3.7 of the Company Disclosure Schedule, (c) incurred since the date of the Company Balance Sheet in the ordinary course of business, consistent with past practice or (d) the transactions contemplated hereby.
- 3.8 Absence of Adverse Changes. Since the date of the Company Balance Sheet to the date of this Agreement, there has not occurred any change, event, circumstance or development that is reasonably likely to have a Company Material Adverse Effect. From the date of the Company Balance Sheet until the date of this Agreement, except as contemplated hereby, (a) the business of the Company and its subsidiaries, taken as a whole, has been conducted in the ordinary course of business and (b) none of the Company or any of its subsidiaries has taken any action that would have required the consent of Parent under Section 5.1 of this Agreement, had such action or event occurred after the date of this Agreement.

### 3.9 Compliance with Laws.

- (a) The Company and the Company Subsidiaries, including their respective employees (to the extent applicable), have obtained each material Federal, state, county, local or foreign governmental consent, license, permit, grant or other authorization of a Governmental Entity (i) pursuant to which the Company or any Company Subsidiary currently operates or holds any interest in any of its properties or (ii) that is required for the operation of the business of the Company or any Company subsidiary or the holding of any such interest ((i) and (ii) are herein collectively called Permits ), and all such Permits are in full force and effect, except where the failure to obtain or have any such Permit does not, individually, or in the aggregate, constitute a Company Material Adverse Effect; and no proceeding is pending or, to the knowledge of the Company, threatened to revoke, suspend, cancel, terminate or adversely modify any such Permit.
- (b) The Company and the Company Subsidiaries have, since January 1, 2007, complied in all material respects with all federal, state, local or foreign laws, statutes, regulations, rules, ordinances and judgments, decrees, orders, writs and injunctions, of any court or Governmental Entity (collectively, <u>Laws</u>) applicable to the Company and the Company Subsidiaries or their business or relating to any of the real or tangible personal property owned, leased or used by them.
- (c) None of the Company, the Company Subsidiaries, any of their respective officers or employees, or to the Company s knowledge, any of its suppliers, licensees, agents or distributors, or any other person acting on behalf of the Company or the Company Subsidiaries, directly or indirectly, has at any time since January 1, 2007, (i) violated any provision of the U.S. Foreign Corrupt Practices Act of 1977 (the FCPA), (ii) violated any applicable Law enacted in any jurisdiction in connection with or arising under the OECD Convention Combating Bribery of Foreign Public Officials in International Business Transactions (the OECD Convention), (iii) made, offered to make, promised to make or authorized the payment or giving of, directly or indirectly, any bribe, rebate, payoff, influence payment, kickback or other unlawful payment or gift of money or anything of value prohibited under any applicable Law addressing matters comparable to those addressed by the FCPA or the OECD Convention implementing legislation concerning such payments or gifts in any jurisdiction (any such payment, a Prohibited Payment), (iv) been subject to any investigation by any Governmental Entity with regard to any Prohibited Payment, or (v) violated any other Laws regarding use of funds for political activity or commercial bribery.
- (d) Neither the Company nor any Company Subsidiary has knowledge of any actual or threatened enforcement action by the U.S. Food and Drug Administration (the <u>FDA</u>) or any other Governmental Entity which has jurisdiction over the operations of the Company and the Company Subsidiaries, and since January 1, 2007, none has received notice of any pending or threatened claim by the FDA or any other Governmental Entity which has jurisdiction over the operations of the Company and the Company Subsidiaries against the Company or the Company Subsidiaries, and the Company and the Company Subsidiaries have no knowledge or reason to believe that any Governmental Entity is considering such action.

- (e) Since January 1, 2007, all material reports, documents, claims and notices required to be filed, maintained, or furnished to the FDA or any Governmental Entity by the Company or the Company Subsidiaries have been so filed, maintained or furnished. All such reports, documents, claims, and notices were complete and correct in all material respects on the date filed (or were corrected in or supplemented by a subsequent filing) such that no material liability exists with respect to the completeness or accuracy of such filing.
- (f) The Company and the Company Subsidiaries have not received (i) since January 1, 2007, any FDA Form 483 or Warning Letter from the FDA, or (ii) since January 1, 2007, any untitled letter or other material written correspondence or notice from the FDA or other Governmental Entity alleging or asserting noncompliance with any applicable Laws or Permits, and the Company and the Company Subsidiaries have no knowledge or reason to believe that the FDA or any Governmental Entity is considering such action.
- (g) All studies, tests and preclinical and clinical trials being conducted by the Company or the Company Subsidiaries have been and are being conducted in material compliance with experimental protocols, procedures and controls pursuant to accepted professional scientific standards and applicable local, state and federal Laws, rules, regulations and guidance, including, but not limited to the applicable requirements of Good Laboratory Practices or Good Clinical Practices, as applicable. To the knowledge of the Company, there are no studies, tests or trials the results of which materially call into question the clinical results described or referred to in the Company SEC Reports filed prior to the date hereof, when viewed in the context in which such results are described and the clinical state of development. The Company and the Company Subsidiaries have not received any written notices, correspondence or other communication since January 1, 2007 from the FDA or any other Governmental Entity requiring the termination, suspension or material modification of any ongoing or planned clinical trials conducted by, or on behalf of, the Company or the Company Subsidiaries, or in which the Company or the Company Subsidiaries have participated, and the Company and the Company Subsidiaries have no knowledge that the FDA or any other Governmental Entity is considering such action. For the purposes of this Agreement, (i) Good Clinical Practices means the FDA s standards for the design, conduct, performance, monitoring, auditing, recording, analysis, and reporting of clinical trials contained in 21 C.F.R. Part 50, 54, 56, 312, 314, 320, 812, and 814 and (ii) Good Laboratory Practices means the FDA s standards for conducting non-clinical laboratory studies contained in 21 C.F.R. Part 58.
- (h) Since January 1, 2007, the manufacture of products by the Company and the Company Subsidiaries has been conducted in material compliance with all applicable Laws, including the FDA s current Good Manufacturing Practices. In addition, since January 1, 2007, the Company and the Company Subsidiaries have been in material compliance with all other applicable FDA requirements, including, but not limited to, registration and listing requirements set forth in 21 U.S.C. Section 360 and 21 C.F.R. Part 207 and 807. For the purposes of this Agreement, Good Manufacturing Practices means the requirements set forth in the quality systems regulations for medical devices contained in 21 C.F.R. Part 820.
- (i) Since January 1, 2007, the Company and the Company Subsidiaries have not either voluntarily or involuntarily, initiated, conducted, or issued, or caused to be initiated, conducted or issued, any recall, market withdrawal or replacement, safety alert, warning, dear

doctor letter, investigator notice or other notice or action relating to an alleged lack of safety or efficacy of any product or product candidate. The Company and the Company Subsidiaries have no knowledge of any facts which are reasonably likely to cause (i) the recall, market withdrawal or replacement of any product sold or intended to be sold by the Company or the Company Subsidiaries; (ii) a change in the marketing classification or a material change in labeling of any such products; or (iii) a termination or suspension of marketing of any such products.

- (j) Since January 1, 2007, the Company and the Company Subsidiaries have been in material compliance with federal or state criminal or civil Laws applicable to the business of the Company and the Company Subsidiaries (including without limitation the federal Anti-Kickback Statute (42 U.S.C. § 1320a-7b(b)), Stark Law (42 U.S.C. §1395nn), False Claims Act (31 U.S.C. §3729 et seq.), Health Insurance Portability and Accountability Act of 1996 (42 U.S.C. § 1320d et seq), and any comparable state Laws), or the regulations promulgated pursuant to such Laws, or which are cause for civil penalties or mandatory or permissive exclusion from Medicare (Title XVIII of the Social Security Act), Medicaid (Title XIX of the Social Security Act), or any other state or federal health care program (Program Interior Interio
- (k) The Company and the Company Subsidiaries are in compliance in all material respects with their respective obligations to report accurate pricing information for their products to Governmental Entities and to price reporting services relied upon by Governmental Entities and other payors, including, as applicable and without limitation, their obligation to report accurate Best Price and Average Manufacturer Price to the extent required and defined in 42 U.S.C.A. § 1396r-8 and Medicaid rebate agreements entered into by the Company and the Company Subsidiaries, and Average Sales Price under the Medicare Modernization Act of 2003, and obligation to charge accurate prices to purchasers entitled to Federal Supply Schedule prices, Federal Upper Limit prices and Federal Ceiling Prices.
- (I) To the knowledge of the Company, neither the Company nor any Company Subsidiary has engaged in an unlawful or unauthorized practice of medicine or other professionally licensed activities through any web sites sponsored or operated or formerly sponsored or operated, by the Company or any Company Subsidiary.
- (m) The Company and the Company Subsidiaries have complied in all material respects with all applicable export control Laws, including those administered by the U.S. Department of Commerce and the U.S. Department of State, and applicable asset control Laws, including those administered by the U.S. Department of the Treasury.

### 3.10 Actions and Proceedings.

- (a) There are no outstanding orders, judgments, injunctions, decrees or other requirements of any Governmental Entity against the Company, any Company Subsidiary or any of their securities, assets or properties that, individually or in the aggregate, constitutes a Company Material Adverse Effect or that seeks to materially delay or prevent the consummation of the transactions contemplated by this Agreement. Except as disclosed under the heading Legal Proceedings in the Company SEC Reports filed prior to the date hereof, there are no material actions, suits or claims or legal, administrative or arbitration proceedings pending or, to the knowledge of the Company, threatened against the Company, any Company Subsidiary, or any of their securities, assets or properties.
- (b) There are no pending nor, to the knowledge of the Company, threatened civil, criminal or administrative actions, suits, demands, claims, hearings, notices of violation, investigations, proceedings or demand letters relating to any alleged hazard or alleged defect in design, manufacture, materials or workmanship, including any failure to warn or alleged breach of express or implied warranty or representation, relating to any product manufactured, distributed or sold by or on behalf of the Company or any Company Subsidiary. There are no product liability claims pending against the Company.

### 3.11 Contracts and Other Agreements.

(a) Except as set forth on Section 3.11(a) of the Company Disclosure Schedule, neither the Company nor any Company Subsidiary is a party to or bound by, and neither they nor their properties are subject to, any material contract (as such term is defined in Item 601(b)(10) of Regulation S-K under the Securities Act) required to be filed as an exhibit to the Company SEC Reports prior to the date of this Agreement (a Material Contract ) that has not been so filed. Except Material Contracts that have expired or terminated by their terms, each Material Contract required to be filed as an exhibit to (i) the Company s Annual Report on Form 10-K filed on February 3, 2010 and (ii) any Company SEC Report filed after February 3, 2010, is valid, in full force and effect and binding upon the Company or the applicable Company Subsidiary, and to the knowledge of the Company, binding upon the other parties thereto in accordance with its terms (except to the extent enforceability may be limited by the effect of applicable bankruptcy, reorganization, insolvency, moratorium or other Laws affecting the enforcement of creditors rights generally and the effect of general principles of equity, regardless of whether such enforceability is considered in a proceeding at Law or in equity), the Company and the Company Subsidiaries are not in default under any Material Contract, except for defaults which individually or in the aggregate would not reasonably be expected to result in termination of a Material Contract or result in a material liability, have performed in all material respects their respective covenants thereunder, and to the knowledge of the Company, no other party to any such Material Contract is in default thereunder, nor to the knowledge of the Company does any condition exist that with notice or lapse of time or both would constitute a default thereunder, except for defaults which individually or in the aggregate would not reasonably be expected to result in termination of a Material Contract or result in a material liability for the Company. Except as set forth on Section 3.11(a) of the Company Disclosure Schedule, none of the execution, delivery, or performance of this Agreement, or the commencement or consummation of Offer, the Merger, or the other transactions contemplated by

this Agreement, shall (i) constitute a default under or give rise to rights to any party under any Material Contract or (ii) create obligations for, or alter the obligations of, the Company, any Company Subsidiary, Parent, Sub or the Surviving Corporation in addition to those obligations of the Company or a Company Subsidiary in effect on the date of this Agreement. True and complete copies of all of the Material Contracts required to be filed as an exhibit to the Company s Annual Report on Form 10-K filed on February 3, 2010 or any Company SEC Report filed after February 3, 2010 have been made available to Parent.

- (b) Except as provided in the Company SEC Reports filed prior to the date hereof, neither the Company nor any Company Subsidiary is a party to any agreement that limits or restricts the Company, any Company Subsidiary or any of their affiliates or successors in competing or engaging in any line of business, in any therapeutic area, in any geographic area or with any person.
- (c) Neither the Company nor any Company Subsidiary is a party to any agreement obligating the Company to file a registration statement under the Securities Act of 1933, as amended (the <u>Securities Act</u>), which filing has not yet been made, and the Company is in material compliance with each such agreement, all of which are listed on Section 3.11(c) of the Company Disclosure Schedule.
- (d) Other than Material Contracts and except as set forth on Section 3.11(d) of the Company Disclosure Schedule, neither the Company nor any Company Subsidiary is a party to any agreement (i) involving research, development or the license of Intellectual Property (as defined in Section 3.12(a)) and presently requiring or that could require payment by or to the Company of amounts exceeding \$100,000 in any 12-month period, (ii) granting a right of first refusal, or right of first offer or comparable right with respect to Intellectual Property, (iii) relating to a joint venture, partnership or other arrangement involving a sharing of profits, losses, costs or liabilities with another person, other than indemnities, insurance contracts, licenses of Intellectual Property presently not requiring or that could not require payment by or to the Company of amounts exceeding \$50,000 in any 12-month period, (iv) providing for the payment or receipt by the Company or any Company Subsidiary of milestone payments or royalties (other than off-the-shelf software license fees) exceeding \$100,000 in any 12-month period, (v) including or involving a loan to a director or officer, or (vi) that individually requires or contemplates aggregate expenditures by the Company and/or any Company Subsidiary in any twelve month period of more than \$100,000.
- (e) To the knowledge of the Company, no officer or director of the Company has (whether directly or indirectly through another entity in which such person has a material interest, other than as the holder of less than two percent (2%) of a class of securities of a publicly traded company) any material interest in any property or assets of the Company (except as a shareholder and except for compensation for service as an officer or director) or a Company Subsidiary, any competitor, customer or supplier of the Company or a Company Subsidiary or any person that is currently a party to any Material Contract or agreement with the Company or a Company Subsidiary.
- (f) Neither the Company nor any Company Subsidiary is party to any interest rate, equity or other swap or derivative instrument.

### 3.12 Property.

(a) To the knowledge of the Company, the Company and the Company Subsidiaries own, are licensed to or otherwise possess legally enforceable rights free and clear of any mortgage, security interest, pledge, lien, charge or encumbrance (other than Permitted Liens) to use all Intellectual Property necessary to conduct the business of the Company and the Company Subsidiaries, taken as a whole, as currently conducted or as contemplated to be conducted (in each case excluding generally commercially available, off-the-shelf software programs), subject to the ownership rights of the owner of licensed or third-party developed Intellectual Property, the terms of the related license or development agreement and royalties and development fees payable thereon. For purposes of this Agreement, the term Intellectual Property means (i) patents, trademarks, service marks, trade names, domain names, copyrights, designs and trade secrets, (ii) applications for and registrations of such patents, trademarks, service marks, trade names, domain names, copyrights and designs, (iii) invention disclosures, processes, formulae, methods, schematics, technology, know-how, computer software programs and applications, (iv) other tangible or intangible proprietary or confidential information and materials and (v) any goodwill associated with any of the foregoing. Company Intellectual Property means all Intellectual Property that the Company and the Company Subsidiaries own (either solely or jointly), use, is held for use for, license, and/or have an ownership or licensed interest in, in any case that is relevant to the conduct of the Company s and the Company Subsidiaries business, as currently conducted or as contemplated to be conducted. Section 3.12(a) of the Company Disclosure Schedule identifies (i) all issued patents and registered trademarks that have been issued to the Company or a Company Subsidiary, (ii) each pending application therefor submitted by the Company or a Company Subsidiary and (iii) all issued patents, registered trademarks and pending applications therefor owned by a third party who has granted the Company or a Company Subsidiary exclusive rights thereto (collectively, the Registered Intellectual Property ). To the Company s knowledge, the Company and Company Subsidiaries have the right to sell their products and services (whether now offered for sale or under development) free from any royalty or other obligations to third parties by virtue of the Company s and/or a Company s Subsidiary use of any licensed Company Intellectual Property in connection therewith. Permitted Lien shall mean (i) any lien for Taxes not yet due or being contested in good faith by appropriate proceedings, (ii) any lien arising in the ordinary course by operation of Law with respect to a liability that is not yet due or that is being contested in good faith, (iii) any lien reflected in the Company Balance Sheet, (iv) purchase money liens arising in the ordinary course of business, (v) the terms and conditions of the Material Contracts and the Permits, (vi) pledges and deposits made in the ordinary course of the business in compliance with workers compensation, unemployment insurance and other social security Laws and (vii) any lien that is released on or prior to the Closing.

(b) The execution and delivery of this Agreement by the Company and the consummation of the Offer and the Merger will not result in the breach of, or create on behalf of any third party the right to terminate or modify, (i) any material license, sublicense or other agreement under which the Company or a Company Subsidiary licenses any Company Intellectual Property to any third party, or (ii) any license, sublicense and other agreement as to which the Company or any Company Subsidiary is a party and pursuant to which the Company or any Company Subsidiaries, taken as a whole, as currently conducted or as contemplated to be conducted, excluding generally commercially available, off-the-shelf software programs.

(c) To the knowledge of the Company, all patents and registrations for trademarks, service marks and copyrights included in the Registered Intellectual Property are subsisting and in full force and effect and have not expired or been cancelled. With respect to such patents, registered trademarks and registered copyrights owned by the Company, the Company has taken all steps necessary to maintain registrations thereof, including by payment when due of all maintenance fees and annuities and the filing of all necessary renewals, statements and certifications. Assignment documents have been executed and filed with relevant Governmental Entities as necessary to transfer to the Company or a Company Subsidiary title to any of the Registered Intellectual Property owned by the Company that was previously owned by a third party and to record such transfer. The named inventors of each of the Company s owned patent applications have assigned the Company s owned patent applications to the Company.

(d) To the knowledge of the Company, the Company s and/or Company Subsidiaries conduct of the business of the Company and Company Subsidiaries, taken as a whole, as currently conducted or as contemplated to be conducted, does not infringe, violate or constitute a misappropriation of any Intellectual Property of any third party. Except as set forth on Section 3.12(d) of the Company Disclosure Schedule, neither the Company nor any Company Subsidiary has within the past six (6) years received any charge, complaint, claim, demand, or notice alleging any interference or any such infringement, violation or misappropriation related to Intellectual Property (including any claim that the Company, a Company Subsidiary or any Company Affiliate must license or refrain from using any Intellectual Property). Except as set forth on Section 3.12(d) of the Company Disclosure Schedule, neither the Company nor any Company Subsidiary has, within the past six (6) years, received any written claim or notice from any Person (i) alleging any such interference, infringement, violation, opposition or misappropriation or (ii) advising that such Person is challenging or threatening to challenge the ownership, use, validity or enforceability of any Company Intellectual Property. To the knowledge of the Company, none of the activities of the employees of the Company or any Company Subsidiary on behalf of such entity violates any confidentiality agreement or obligations which any such employees have with former employers. To the knowledge of the Company, all employees and consultants who contributed to the discovery or development of any of the subject matter of the Company Intellectual Property owned by the Company or its Subsidiaries did so either; (x) within the scope of their employment such that, in accordance with applicable Law, all rights to such developed subject matter became the exclusive property of the Company or Subsidiary of the Company; or (y) pursuant to agreements assigning all rights to such developed subject matter to the Company or a Company Subsidiary. Assignment documents assigning to the Company all rights of such employees, contractors and consultants in respect of such developed subject matter have been duly filed in all relevant patent offices worldwide for all such patent applications and patents owned in whole or in part by the Company. No current or former director, officer, consultant or employee of the Company will, after giving effect to the Offer and the Merger, own any of the Company Intellectual Property that is currently owned by the Company and/or Company Subsidiaries. To the knowledge of the Company, each employee, contractor or consultant of the Company who has proprietary knowledge of confidential information included in the Company Intellectual Property relating to the manufacturing processes, or the formulation of the products, of the Company or a Company

Subsidiary has executed and delivered to the Company an agreement restricting such person s right to use and disclose confidential information of the Company. Except as does not constitute a Company Material Adverse Effect, there are no settlements, forbearances to sue, consents, judgments, or orders or similar obligations to which the Company or any Company Subsidiary is party that: (i) restrict the use of, ownership, or rights to any Company Intellectual Property; (ii) restrict the conduct of the business of the Company or any of its employees with respect to the Company Intellectual Property; or (iii) grant third parties any material rights under Company Intellectual Property.

- (e) To the knowledge of the Company and except as disclosed in Section 3.12(e) of the Company Disclosure Schedule, all maintenance and renewal fees necessary to preserve the material rights of the Company and Company Subsidiaries in connection with such Registered Intellectual Property owned by the Company have been paid and all necessary supporting documentation such as statements and certifications, have been filed. At the Tender Completion Time, the Company shall provide Parent with a schedule of any annuities, taxes, maintenance fees or actions falling due within ninety (90) days of such expiration with respect to such patents, trademarks and copyrights.
- (f) The Company and Company Subsidiaries have implemented commercially reasonable measures to maintain the confidentiality of the Company Intellectual Property of a nature that the Company intends to keep confidential. The proprietary and/or confidential algorithms included in the Company Intellectual Property (except insofar as such algorithms are included in Registered Intellectual Property) developed by the Company and incorporated into the Company s products have not been disclosed or authorized to be disclosed to any third party that is not subject to such confidentiality obligations. To the knowledge of the Company and except as does not constitute a Company Material Adverse Effect, no material trade secret of the Company has been disclosed or authorized to be disclosed by the Company to any third party except pursuant to a confidentiality agreement obligating such third party to maintain the confidentiality of such trade secret, nor by any third party to any other third party in violation of confidentiality obligations to the Company. To the knowledge of the Company, no party to a nondisclosure agreement with the Company in respect of any Company Intellectual Property is in material breach or default thereof.
- (g) To the knowledge of Company, no third party has within the past six (6) years or is infringing, violating or misappropriating any of the Company Intellectual Property, except as set forth on Section 3.12(g) of the Company Disclosure Schedule. Except as disclosed in Section 3.12(g) of the Company Disclosure Schedule, to the knowledge of Company, no third party has within the past six (6) years asserted by notice to the Company or a Company Subsidiary any competing claim of right to use or own any of the Company Intellectual Property.
- (h) To the knowledge of the Company, no open source or public library software, including any version of any software licensed pursuant to any GNU public license, is, in whole or in part, embodied or incorporated, in any manner, into any of such software.
- (i) Neither the Company nor any Company Subsidiary has any current unpaid obligation or demand from a third party to indemnify any third party for or against any interference, infringement, misappropriation or other violation with respect to the Company Intellectual Property or third-party Intellectual Property.

- (j) To the knowledge of the Company and except as do not constitute a Company Material Adverse Effect, the execution of, the delivery of, the consummation of the Offer and Merger contemplated by, and the performance of the Company s obligations under, this Agreement will not result in any loss or impairment of any Company Intellectual Property.
- (k) Except as set forth on Section 3.12(k) of the Company Disclosure Schedule, to the knowledge of the Company, neither government funding nor government, academic or non-profit research facilities were used in the development of any of the Company Intellectual Property that is owned by the Company and/or Company Subsidiaries.
- (1) With respect to property other than Intellectual Property, the Company and each Company Subsidiary has all assets, properties, rights and contracts necessary to permit the Company and the Company Subsidiaries to conduct their business as it is currently being conducted, except where the failure to have such assets, properties, rights and contracts does not constitute a Company Material Adverse Effect. The Company and each Company Subsidiary has good, clear and marketable title to all of its properties, interests in properties and assets, real and personal, reflected in the Company Balance Sheet (except properties, interests in properties and assets sold or otherwise disposed of since the date of the Company Balance Sheet in the ordinary course of business consistent with past practice), or with respect to leased properties and assets, valid leasehold interests in such properties and assets, in each case, free and clear of all imperfections of title, restrictions, encroachments, liens and easements, except (i) liens for current Taxes not yet due and payable, that are payable without penalty or that are being contested in good faith by appropriate proceedings, (ii) such imperfections of title, restrictions, encroachments, liens and easements as do not materially detract from or interfere with the use or value of the properties subject thereto or affected thereby, or otherwise materially impair business operations involving such properties and (iii) liens securing debt which are reflected on the Company Balance Sheet. There are no written or oral subleases, licenses, occupancy agreements or other contractual obligations that grant to third parties the right of use or occupancy of any real property leased by the Company or any Company Subsidiary (collectively, the <u>Real Property</u>), and there is no person in possession of the Real Property other than the Company and the Company Subsidiaries. There is no pending, or, to the knowledge of the Company, threatened eminent domain, condemnation or similar proceeding affecting any Real Property leased by the Company or a Company Subsidiary. To the knowledge of the Company, the property and equipment of the Company and each Company Subsidiary that are used in the operations of business are (i) in good operating condition and repair and (ii) have been maintained in accordance with normal industry practices. Section 3.12(1) of the Company Disclosure Schedule lists all Real Property leased by the Company or a Company Subsidiary. Neither the Company nor any Company Subsidiary owns any Real Property.
- **3.13** Insurance. All policies or binders of material fire, liability, product liability, workers compensation, vehicular, directors and officers and other material insurance held by or on behalf of the Company and the Company Subsidiaries are in full force and effect in all material respects, are reasonably adequate for the businesses engaged in by the Company and the Company Subsidiaries and are in conformity in all material respects with the requirements of all

leases or other agreements to which the Company or the relevant Company Subsidiary is a party and, to the knowledge of the Company, are valid and enforceable in accordance with their terms. Neither the Company nor any Company Subsidiary is in default in any material respect with respect to any provision contained in such policy or binder nor has any of the Company or a Company Subsidiary failed to give any notice or present any material claim under any such policy or binder in due and timely fashion. All premiums for each policy or binder have been paid for the current period, and there are no outstanding premium finance payments due for such period. There are no material outstanding unpaid claims under any such policy or binder. Neither the Company nor any Company Subsidiary has received notice of cancellation or non-renewal of any such policy or binder. All applications for the Company s currently effective directors and officers insurance were true, correct and complete in all material respects when submitted to the carrier. No coverage limits of insurance policies covering the Company or a Company Subsidiary have been exhausted.

**3.14** Commercial Relationships. During the last twelve (12) months prior to the date of this Agreement, none of the Company s or the Company Subsidiaries material suppliers, collaborators, distributors, licensors or licensees has canceled or otherwise terminated its relationship with the Company or a Company Subsidiary or has materially altered its relationship with the Company or a Company Subsidiary. The Company has not received any written threat or notice from any such entity, to terminate, cancel or otherwise materially modify its relationship with the Company or a Company Subsidiary.

### 3.15 Tax Matters.

(a) For purposes of this Agreement, the term <u>Tax</u> (and, with correlative meaning, <u>Taxes and Taxable</u>) means all United States federal, state and local, and all non-U.S., income, profits, franchise, gross receipts, payroll, transfer, sales, employment, social security, unemployment insurance, workers compensation, use, property, excise, value added, ad valorem, estimated, stamp, alternative or add-on minimum, recapture, capital, withholding and any other taxes, charges, duties, impositions or assessments, and any other taxes, fees, charges, levies, excises, duties or assessments of any kind whatsoever, together with all interest, penalties and additions imposed on or with respect to such amounts, including any liability for taxes of a predecessor entity. <u>Tax Return</u> means any return, declaration, report, claim for refund, tax shelter disclosure statements or information return or statement filed or required to be filed with any taxing authority in connection with the determination, assessment, collection or imposition of any Taxes, including any attachments thereto and any amendments thereof.

(b) All federal income Tax Returns and other material Tax Returns required to be filed by or with respect to the Company and the Company Subsidiaries have been timely and properly filed. All such Tax Returns are true, correct and complete in all material respects, and all material Taxes due and payable by the Company or the Company Subsidiaries, whether or not shown on any Tax Return, have been paid except for those Taxes that are being contested in good faith by appropriate proceedings and for which a specific reserve has been established in accordance with generally accepted accounting principles. The Company and the Company Subsidiaries file Tax Returns in all jurisdictions where they are required to so file, and no claim has ever been made by any taxing authority in any jurisdiction where the Company or the Company Subsidiaries do not file Tax Returns that the Company or the Company Subsidiaries are or may be subject to taxation by that jurisdiction.

- (c) There are no material liens or other encumbrances with respect to Taxes upon any of the assets or properties of the Company or the Company Subsidiaries, other than with respect to Taxes not yet due and payable or those Taxes that are being contested in good faith by appropriate proceedings and for which adequate reserves have been established in accordance with United States generally accepted accounting principles.
- (d) No audit is currently pending or threatened with respect to any Tax Return of the Company or the Company Subsidiaries, nor have any material deficiencies for any Taxes been proposed, asserted, threatened or assessed against the Company or the Company Subsidiaries.
- (e) Neither the Company nor the Company Subsidiaries has extended or waived the application of any statute of limitations of any jurisdiction regarding the assessment or collection of any Tax of the Company or the Company Subsidiaries.
- (f) With respect to any period for which Tax Returns have not yet been filed, or for which Taxes are not yet due or owing, the Company has, in accordance with generally accepted accounting principles, made due and sufficient accruals for such Taxes (excluding any deferred taxes or similar items that reflect timing differences between tax and financial accounting principles) in the Company s books and records.
- (g) The Company and the Company Subsidiaries have withheld all material amounts required by Law or contract to be withheld from the wages, salaries or other payments to (i) employees, independent contractors, creditors, shareholders of or consultants to the Company and (ii) any other third party. Such withheld amounts were either duly paid to the appropriate taxing authority to the extent due and payable or, to the extent not due and payable, for which a specific reserve has been established on the Company Balance Sheet in accordance with generally accepted accounting principles. The Company and the Company Subsidiaries have complied in all material respects with all record keeping and reporting requirements in connection with amounts paid or owing to any employee, independent contractor, creditor, shareholder or other third party.
- (h) The Company and the Company Subsidiaries are not a party to or bound by, nor do they have any obligation under, any Tax sharing, Tax indemnification, or Tax allocation agreement, or similar contract or arrangement. Neither the Company nor any Company Subsidiary has any material liability for the Taxes of any other person (other than Taxes of any member of a consolidated group of which the Company is the common parent) under Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or non-U.S. Law), as a transferee or successor, by contract, or otherwise.
- (i) Neither the Company nor any Company Subsidiary has made any payments that has, or has been or is a party to any agreement, contract, arrangement or plan that have, resulted or will result, separately or in the aggregate, in the payment of any excess parachute payment within the meaning of Section 280G of the Code or in the imposition of an

excise Tax under Section 4999 of the Code (or any corresponding provisions of state, local or non-U.S. Tax Law). Neither the Company nor any Company Subsidiary is a party to, nor is otherwise obligated under, any contract, agreement, plan or arrangement that provides for the gross-up of the excise Tax imposed by Section 4999 of the Code (or any corresponding provisions of state, local or non-U.S. Tax Law).

- (j) Neither the Company nor the Company Subsidiaries has made any payments that have, or has been or is a party to any agreement, contract, arrangement or plan that have, resulted or will result, separately or in the aggregate, in the payment of any compensation which would be subject to the deduction limit imposed by Section 162(m) of the Code.
- (k) Neither the Company nor the Company Subsidiaries are, or were during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code, a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code.
- (1) Neither the Company nor any Company Subsidiary has distributed stock of another corporation, or has had its stock distributed by another corporation, in a transaction that was governed, or purported or intended to be governed, in whole or in part, by Sections 355 or 361 of the Code.
- (m) Since December 31, 2005, neither the Company nor the Company Subsidiaries has (i) changed any financial or Tax accounting methods, policies or practices of the Company or any of the Company Subsidiaries, except as required by a change in generally accepted accounting principles or SEC rules, regulations or guidelines or applicable Law, (ii) made, revoked, or amended any material Tax election of the Company or any of the Company Subsidiaries, (iii) filed any amended Tax Return or claim for refund of the Company or any of the Company Subsidiaries, (iv) entered into any closing agreement affecting any material Tax liability or refund of the Company or any the Company Subsidiaries, or (v) settled or compromised any material Tax liability or refund of the Company or any of the Company Subsidiaries.
- (n) The Company has not utilized any net operating losses, net capital losses, or tax credit carryovers that were subject to limitation as a result of an ownership change as described in Section 382(g) of the Code.
- (o) The Company has made available to Parent complete and correct copies of all Tax Returns, supporting working papers, examination reports, cost sharing or similar arrangements and statements of deficiencies assessed against or agreed to by the Company or any Company Subsidiary filed or received by the Company or any Company Subsidiary since November 30, 2004.
- (p) The Company will not be required to include any material item of income in, or exclude any material item of deduction from, Taxable income for any Taxable period (or portion thereof) beginning after the Closing Date as a result of any: (i) adjustment under Section 481 of the Code (or any corresponding or similar provisions of state, local or non-U.S. Tax Law) made prior to the Closing Date or (ii) closing agreement as described in Section 7121 of the Code (or any corresponding or similar provisions of state, local or non-U.S. Tax Law) executed during the six (6) year period ending on the Closing Date.

(q) Neither the Company nor any of the Company Subsidiaries has engaged in any listed transaction identified pursuant to Treasury Regulation Section 1.6011-4(b)(2) or any corresponding or similar provisions of state, local or non-U.S. Law.

## 3.16 Employee Benefit Plans.

(a) Section 3.16(a) of the Company Disclosure Schedule lists each employee benefit plan (as hereinafter defined) (i) that is maintained, contributed (or required to be contributed) to, or sponsored by the Company or any Company Subsidiary, or (ii) to which the Company or any Company Subsidiary is a party, or (iii) with respect to which the Company or any Company Subsidiary has or has had any liability, including any contingent liability, for the payment or delivery of any premiums, compensation or benefits (individually, a Plan and collectively, the Plans ). For purposes of the preceding sentence, an employee benefit plan is any of the following that benefits or is intended to benefit any current or former employee or director (whether or not an employee) of, or consultant or other service provider (whether or not an employee) with respect to the Company or an ERISA Affiliate (as defined in Section 3.16(b)), or the beneficiaries of any of them: (A) a plan described in Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended (ERISA); (B) a stock bonus, stock option, stock purchase, restricted stock, restricted stock unit, stock appreciation right, or other equity-based plan, policy, program, agreement or arrangement; or (C) an incentive, bonus, deferred compensation, welfare, retiree medical or life insurance, retirement, supplemental retirement, termination, salary continuation, severance, change in control, and any material fringe benefit or other material benefit plan, policy, program, agreement or arrangement, whether written or unwritten. With respect to each Plan, the Company has delivered to Parent a true and complete copy of each of the following, together with all amendments: (i) all documents embodying the Plan (or, where a Plan has not been reduced to writing, a summary of all material Plan terms), (ii) in the case of any funded Plan, the trust agreement or similar instrument, (iii) for each Plan subject to the requirement that annual reports be filed on a Form 5500, the two most recently filed annual reports, with schedules, financial statements and auditor s opinion attached, (iv) in the case of each Plan intended to be qualified under Section 401(a) of the Code, the most recent IRS determination or opinion letter applicable to the Plan, (v) all related custodial agreements, insurance policies (including fiduciary liability insurance covering the fiduciaries of the Plan), administrative services and similar agreements, and investment advisory or investment management agreements, if any, and (vi) the most recent summary plan description, summaries of material modifications or similar summary and any employee handbook referencing the Plan.

(b) None of the Company or any Company Subsidiary or any other person (including an entity) that together with the Company or any Company Subsidiary is or at any relevant time was treated as a single employer under Section 414(b), (c), (m) or (o) of the Code (each, together with the Company and any Company Subsidiary, an <u>ERISA Affiliate</u>) has ever contributed or been required to contribute to, or has ever sponsored, maintained or participated in, (i) a pension plan (within the meaning of Section 3(2) of ERISA) subject to Section 412 of the Code or Title IV of ERISA, (ii) a multiemployer plan (within the meaning of Section 3(37) or 4001(a)(3) of ERISA), (iii) a single employer pension plan (within the meaning of Section 4001(a)(15)

of ERISA) for which an ERISA Affiliate would reasonably be expected to incur liability under Section 4063 or 4064 of ERISA, or (iv) a voluntary employees beneficiary association within the meaning of Section 501(c)(9) of the Code or a welfare benefit fund within the meaning of Section 419(e) of the Code. Except as described in Section 3.16(b) of the Company Disclosure Schedule or as provided in Section 5.1(b)(iii), the transactions contemplated by this Agreement will not, by themselves or together with any other event, cause or result in the payment, acceleration or vesting of any payment, right or benefit under any Plan.

- (c) Each Plan that is intended to be qualified under Section 401(a) of the Code is the subject of a favorable determination or may rely upon an opinion letter from the Internal Revenue Service (the <u>IRS</u>). No such determination or opinion letter has been revoked, and revocation has not been threatened. No such Plan has been amended or operated since the date of its most recent determination or opinion letter in any respect, and no act or omission has occurred, that has adversely affected its qualification.
- (d) Each Plan has been operated, maintained and administered at all times in all material respects in accordance with its terms. Each Plan, including any associated trust or fund, has been established, operated and administered in material compliance with the applicable provisions of ERISA, the Code and other applicable Law (including, where applicable, non-U.S. Law), and, to the knowledge of the Company, nothing has occurred with respect to any Plan that has subjected or could reasonably be expected to subject the Company or any Company Subsidiary to any material liability or penalty under Section 502 of ERISA or to any material tax under Chapter 43 of the Code. All filings and reports with respect to each Plan required to have been submitted to the IRS, the United States Department of Labor, or any other Governmental Entity have been duly and timely submitted in all material respects.
- (e) No Plan provides health or life insurance benefits following retirement or other termination of employment, and neither the Company nor any Company Subsidiary has any obligation to provide any such benefits following retirement or other termination of employment, in each case except for benefit continuation coverage to the extent required under Part 6 of Subtitle B of Title I of ERISA.
- (f) With respect to each Plan, no administrative investigation, inquiry, audit or other proceeding by the IRS, Department of Labor or other Governmental Entity, and no other lawsuit, or claim, other than claims for benefits in the ordinary course and proceedings with respect to qualified domestic relations orders, is pending or, to the knowledge of the Company, threatened.
- (g) With respect to each Plan, all contributions (including salary reduction contributions), premiums and other payments (i) to the extent due, have been timely made in all material respects, and (ii) to the extent not yet due, have been appropriately accrued on the books of the Company and the Company Subsidiaries.
- (h) Each Plan subject to Section 409A of the Code (<u>Section 409A</u>) has been operated in good faith compliance with Section 409A. No Plan which is subject to the requirements of Section 409A, violates such requirements. All Company Options granted by the Company after October 3, 2004 or which vest or vested (in whole or in part) after December 31,

2004, have (or, if already terminated, had) an exercise price that was not less than the fair market value of the underlying stock as of the date such Company Option was granted. The Company is not a party to, nor is otherwise obligated under, any contract, agreement, plan or arrangement that provides for the gross-up of Tax imposed by Section 409A(a)(1)(B) of the Code.

- (i) Except for the plans listed on Section 3.16(b) of the Company Disclosure Schedule, no Plan is subject to the Laws of a jurisdiction other than the United States, whether or not United States Law also applies. For purposes of the preceding sentence, the Commonwealth of Puerto Rico, Guam, American Samoa, the Northern Marianna Islands and the Virgin Islands shall be considered jurisdictions other than the United States.
- (j) Each Plan and its related documentation or agreement, summary plan description, or other written communication distributed generally to employees by its terms expressly and adequately reserves the right to amend and terminate such Plan, and, except as listed in Section 3.16(j) of the Company Disclosure Schedule, each Plan may be terminated without material liability to the Company or any Company Subsidiary, except for vested benefits accrued through the Closing Date and the administrative and professional costs incurred in such transaction. Except as listed in Section 3.16(j) of the Company Disclosure Schedule, no Plan includes in its assets any securities issued by the Company or any Company Subsidiary. No Company Plan has been subject to Tax under Section 511 of the Code.

### 3.17 Employee Relations.

- (a) Except as otherwise provided in this Section 3.17(a), upon termination of the employment of any employees or the service relationship of any director or consultant, none of the Company, the Company Subsidiaries, the Surviving Corporation nor Parent shall be liable, by reason of the Offer or Merger or anything done in connection with the Offer or Merger, to any of such employees, directors, or consultants for severance pay or any other similar payments (other than accrued salary, vacation or sick pay in accordance with normal policies), except as set forth in Section 3.17(a) of the Company Disclosure Schedule. True and complete information as of June 1, 2010, as to the employee identification number, current job title, date of hire (or election), base salary and target bonus opportunity (if applicable) for all current employees, directors and executive officers of the Company or any Company Subsidiary has been provided or otherwise made available to Parent.
- (b) The Company and each Company Subsidiary (i) is in compliance in all material respects with all applicable foreign, Federal, state and local Laws, rules and regulations respecting employment, employment practices, terms and conditions of employment and wages and hours, in each case, with respect to employees, (ii) has withheld all material amounts required by Law or by agreement to be withheld from the wages, salaries and other payments to employees, (iii) is not liable for any material arrears of wages, salaries, commissions, bonuses or other direct compensation for any services performed or amounts required to be reimbursed to any employees or consultants or any material taxes or any penalty for failure to comply with any of the foregoing, and (iv) is not liable for any payment to any trust or other fund or to any Governmental Entity, with respect to unemployment compensation benefits, social security or other benefits or obligations for employees (other than routine payments to be made in the ordinary course of business and consistent with past practice), except for matters that, individually or in the aggregate, could not reasonably be expected to result in a liability in excess of \$1 million.

- (c) No work stoppage or labor strike against the Company or any Company Subsidiary is pending or, to the knowledge of the Company, threatened with, any material labor dispute, grievance, or litigation relating to labor, safety or discrimination matters involving any employee, including without limitation charges of unfair labor practices or discrimination complaints that, if adversely determined, could reasonably be expected to result in material liability to the Company. Neither the Company nor any Company Subsidiary has engaged in any unfair labor practices within the meaning of the National Labor Relations Act that is expected to result in material liability to the Company. Neither the Company nor any Company Subsidiary is presently, nor has it been in the past, a party to or bound by any collective bargaining agreement or union contract with respect to employees other than as set forth in Section 3.17 of the Company Disclosure Schedule and no collective bargaining agreement is being negotiated by the Company or any Company Subsidiary. To the knowledge of the Company, no union organizing campaign or activity with respect to non-union employees of the Company or any Company Subsidiary is ongoing, pending or threatened.
- (d) Section 3.17(d) of the Company Disclosure Schedule sets forth a list of all individuals who are independent contractors currently performing consulting services relating to the Company or any Company Subsidiary or under contract to perform such future services for the Company or any Company Subsidiary. All such individuals who are not citizens of the United States have acknowledged in writing that they are not employees of the Company or any Company Subsidiary and are not entitled to any compensation or benefits provided to an employee of the Company or any Company Subsidiary. Accurate and complete copies of all written agreements with any such independent contractor have been provided to Parent. Other than the employees, directors and executive officers described in Section 3.17(a) and the independent contractors described in this Section 3.17(d), no other individuals provide services to the Company and Company Subsidiaries.
- (e) Section 3.17(e) of the Company Disclosure Schedule lists each oral or written agreement with any employee, director or officer of the Company or any Company Subsidiary (i) under which the amount or timing of benefits are contingent, or the terms are altered, upon the occurrence of a transaction involving the Company or any Company Subsidiary of the nature of any of the transactions contemplated by this Agreement, (ii) providing any term of employment or compensation guarantee, or (iii) providing severance benefits or other benefits after the termination of employment of such employee, director or officer.

- **3.18** Environmental Matters. Notwithstanding anything else in this Agreement to the contrary, this Section 3.18 shall constitute the sole representations and warranties with respect to environmental matters. Except as does not constitute, individually or in the aggregate, a Company Material Adverse Effect or as set forth in Schedule 3.18 of the Company Disclosure Schedule:
- (a) Neither the Company nor any of the Company Subsidiaries is in violation of, or has been notified in writing by any Governmental Entity at any time since January 1, 2007 that it is in violation of any Environmental Law (as defined in Section 3.18(e)(i)). There has been no generation, use, handling, storage or disposal of any Hazardous Materials (as defined in Section 3.18(e)(ii)) by the Company or any of the Company Subsidiaries in violation of any Environmental Law at any property currently owned or operated by, or premises leased by, the Company or any of the Company Subsidiaries during the period of the Company s or such Company Subsidiary s ownership, operation or lease which created any requirement pursuant to Environmental Law for the Company or any Company Subsidiary to remediate such property. To the Company s knowledge, there is no underground storage tank at any site owned or operated by, or premises leased by the Company or any Company Subsidiary.
- (b) No property currently or formerly owned or operated by, or premises currently or formerly leased by, the Company or any of the Company Subsidiaries is listed, or to the Company s knowledge, proposed for listing, on the National Priorities List or the Comprehensive Environmental Response, Compensation, and Liability Information System, both as maintained under the Federal Comprehensive Environmental Response, Compensation and Liability Act (<u>CERCLA</u>), or on any comparable state governmental lists. Neither the Company nor any Company Subsidiary has received written notification of, and the Company has no knowledge of, any potential responsibility or liability of the Company or any Company Subsidiary pursuant to the provisions of (i) CERCLA, or (ii) any similar Federal, state, local, foreign or other Environmental Law.
- (c) The Company and the Company Subsidiaries have obtained all permits required by Environmental Law necessary to enable them to conduct their respective businesses as currently conducted and are in compliance with such permits. All such permits are in full force and effect and, to the Company s knowledge, there are no pending or threatened claims that seek the revocation, cancellation, suspension or any adverse modification of any such permits.
- (d) The Company previously has made available to Parent copies of all environmental site assessments prepared by any person, and permits required under Environmental Laws and all other material correspondence with Governmental Entities in the Company s possession relating to compliance with Environmental Laws.
- (e) For purposes of this Agreement:
- (i) Environmental Laws means any applicable Federal, state or local Laws, in each case as amended and in effect in the jurisdiction in which the applicable site or premises are located, pertaining to the protection of human health, safety or the environment, including without limitation, the following statutes and all regulations promulgated thereunder: CERCLA; the Emergency Planning and Community Right-to-Know Act, 42 U.S.C. § 11001 et seq.; the Resource Conservation and Recovery Act, 42 U.S.C. § 6901 et seq.; the Federal Water Pollution Control Act, 33 U.S.C. § 1251 et seq.; the Federal Clean Air Act, 42 U.S.C. § 7401 et seq.; the Federal Insecticide, Fungicide and Rodenticide Act, 7 U.S.C. § 136 et seq.; the Toxic Substance Control Act, 15 U.S.C. § 2601 et seq.; the Oil Pollution Act of 1990, 33 U.S.C. § 2701 et seq.; the Hazardous Materials Transportation Act, as amended, 49 U.S.C. § 1801 et seq.; the Atomic Energy Act, 42 U.S.C. § 2014 et seq.; any state or local statute of similar effect; and any Laws relating to protection of the environment which regulate the management or disposal of biological agents or substances including medical or infectious wastes;

- (ii) <u>Hazardous Material</u>s means (A) any chemicals, materials or substances defined as or included in the definition of hazardous substances, hazardous wastes, hazardous materials, chemical substances, toxic substances, toxic pollutants, pollutants, contaminants, pesticides, defined in any applicable Environmental Law, or (B) any petroleum or petroleum products, radioactive materials, asbestos-containing materials, polychlorinated biphenyls, urea formaldehyde foam insulation, radon and any other substance defined or designated as hazardous, toxic or harmful to human health, safety or the environment under any Environmental Law; and
- (iii) Release has the meaning specified in CERCLA.
- 3.19 No Breach. Except for (a) filings with the SEC under the Exchange Act or the Securities Act, (b) filings with the Department and the DSOS contemplated herein, (c) the filing of a Notification and Report Form under the Hart-Scott-Rodino Antitrust Improvements Act, as amended (the HSR Act ) and any similar filings in foreign jurisdictions, (d) filings with, or notices to, The NASDAQ Stock Market, and (e) matters listed in Section 3.19 of the Company Disclosure Schedule, the execution, delivery and performance of this Agreement by the Company and the consummation by the Company of the transactions contemplated hereby will not (i) violate any provision of the articles of incorporation or bylaws of the Company, (ii) violate, conflict with or result in the breach of any of the terms or conditions of, result in modification of, reguire any notice, waiver or action under, or otherwise give any other contracting party the right to terminate, accelerate obligations under or receive payment under or constitute (or with notice or lapse of time or both constitute) a default under, any Material Contract, (iii) violate any Law applicable to the Company or the Company Subsidiaries or by which any of the Company s or the Company Subsidiaries assets or properties is bound, (iv) violate any Permit, (v) require any filing with, notice to, or permit, consent or approval of, any Governmental Entity, or (vi) result in the creation of any lien or other encumbrance on the assets or properties of the Company or a Company Subsidiary, excluding from the foregoing clauses (ii), (iii), (iv), (v), and (vi) violations, breaches and defaults which, and filings, notices, waivers, permits, consents and approvals the absence of which, in the aggregate, does not constitute a Company Material Adverse Effect and would not materially increase the costs of consummation of the Offer and the Merger. Except as set forth in Section 3.19 of the Company Disclosure Schedule, neither the Company nor any Company Subsidiary is or will be required to give any notice to or obtain any consent or waiver from, or give any notice to, any individual or entity in connection with the execution and delivery of this Agreement or the consummation of the transactions contemplated hereby other than failures which do not, in the aggregate, constitute a Company Material Adverse Effect,

### 3.20 Board Approvals; Anti-Takeover; Vote Required.

(a) The Company Board of Directors, at a meeting duly called and held at which all directors were present, has unanimously (i) duly and validly approved and taken all corporate action required by applicable Law to be taken by the Company Board of Directors to adopt and approve this Agreement and thereby authorize the consummation of the transactions

contemplated hereby, (ii) resolved that this Agreement is advisable and in the best interests of the shareholders of the Company, and (iii) subject to the other terms and conditions of this Agreement, resolved to recommend that the shareholders of the Company accept the Offer, tender their Shares to Sub pursuant to the Offer, and, if necessary, adopt and approve this Agreement and thereby approve the Merger (such recommendation being referred to herein as the <a href="Company Board Recommendation">Company Board Recommendation</a> ), and, as of the date of this Agreement, none of the aforesaid actions by the Company Board of Directors has been amended, rescinded or modified.

- (b) As of the date of this Agreement and assuming the truthfulness of the representation set forth in Section 4.7, the Company Board of Directors has taken all action necessary under applicable Law to ensure that no restrictions contained in any fair price, business combination or similar statute (including Section 780 of the MBCA) will apply to the execution, delivery or performance of this Agreement and the Tender and Voting Agreement with the CEO Shareholder.
- (c) If approval by the Company s shareholders is required under the MBCA to adopt and approve this Agreement and thereby approve the Merger, assuming the truthfulness of the representation set forth in Section 4.7, the only votes necessary for such adoption and approval are the affirmative votes of the holders of a majority of the outstanding Shares.
- (d) The Company (acting through a compensation committee of the Company s Board of Directors composed solely of independent directors (as such term is used in Rule 14d-10(d) under the Exchange Act)) has taken all steps necessary to cause each compensation, severance or other benefit agreement, arrangement or understanding between the Company or a Company Subsidiary and any of its or their current or former officers, directors or employees to be approved as an employment compensation arrangement and to satisfy any applicable requirements of the non-exclusive safe-harbor set forth in Rule 14d-10(d) under the Exchange Act. The Company has provided or made available copies of all resolutions adopted or actions taken in connection with such employment compensation arrangement, and each such employment compensation arrangement in existence as of the date of this Agreement is listed on Section 3.20(d) of the Company Disclosure Schedule.

#### 3.21 Financial Advisor.

- (a) The Company Board of Directors has received the opinion of Leerink Swann LLC (the <u>Financial Advisor</u>) to the effect that, as of the date of such opinion, subject to the assumptions and qualifications set forth therein, the consideration to be received in the Offer and the Merger, taken together, by the holders of the Shares (other than as set forth in such opinion) is fair, from a financial point of view, to such holders. The Company shall forward to Parent, solely for informational purposes, a copy of the written version of such opinion promptly following the Company s receipt thereof.
- (b) No broker, finder, agent or similar intermediary has acted on behalf of the Company in connection with this Agreement or the transactions contemplated hereby, and there are no brokerage commissions, finders—fees or similar fees or commissions payable in connection herewith based on any agreement, arrangement or understanding with the Company, or any action taken by the Company. The Company previously has provided or made available to Parent a copy of the Financial Advisor engagement letter, and the fees set forth therein are the only fees payable by the Company to the Financial Advisor.

3.22 Information in the Offer Documents and the Schedule 14D-9. The information supplied by the Company expressly for inclusion or incorporation by reference in the Offer Documents or the Schedule 14D-9, including any amendments thereof and supplements thereto, will not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading. The Schedule 14D-9, including any amendments thereof and supplements thereto, will comply in all material respects with the provisions of applicable federal securities and Michigan corporation Laws and, on the date filed with the SEC and on the date first published or sent or given to the Company's shareholders, will not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading, except that the Company makes no representation or warranty with respect to statements made in the Schedule 14D-9, including any amendments thereof and supplements thereto, based on information furnished by Parent or Sub expressly for inclusion therein. The Company has obtained all necessary consents to permit the inclusion in its entirety, and a fair summary of the analysis underlying, the fairness opinion of the Financial Advisor in the Schedule 14D-9, including any amendments thereof and supplements thereto. As of the date of this Agreement, each member of the Company Board of Directors and each executive officer of the Company has advised the Company that he or she will tender all Shares, if any, beneficially owned by him or her pursuant to the Offer.

3.23 Information in the Proxy Statement. The Proxy Statement, if any, (and any amendment thereof and supplement thereto) at the date mailed to the Company shareholders and at the time of any meeting of the Company shareholders to be held in connection with the Merger, will not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading, except that no representation is made by the Company with respect to statements made therein based on information supplied in writing by Parent or Sub expressly for inclusion in the Proxy Statement, including any amendments thereof and supplements thereto, as to information supplied by the Company for inclusion therein, will comply in all material respects with the provisions of the Exchange Act and the rules and regulations thereunder and all applicable federal securities and Michigan corporation Laws. The Company has obtained all necessary consents to permit the inclusion in its entirety, and a fair summary of the analysis underlying, the fairness opinion of the Financial Advisor in the Proxy Statement, including any amendments thereof and supplements thereto.

#### SECTION 4 REPRESENTATIONS AND WARRANTIES OF PARENT

Parent and Sub hereby make the following representations and warranties to the Company:

**4.1** Organization. Parent is a corporation duly incorporated, validly existing and in good standing under the Laws of the State of Delaware. All of the issued and outstanding capital stock of Sub is owned, beneficially and of record, by Parent. Except for obligations and liabilities incurred in connection with its incorporation and the transactions contemplated by this Agreement, Sub has not and, prior to the Acceptance Time, will not have incurred, directly or indirectly, any material obligations or liabilities or engaged in any business activities or entered into any agreements or arrangements with any person that would impair in any material respect the ability of each of Parent and Sub, as the case may be, to perform its respective obligations under this Agreement or prevent or materially delay the consummation of the transactions contemplated by this Agreement.

### 4.2 Authority to Execute and Perform Agreement.

- (a) Parent and Sub have the power and authority to enter into, execute and deliver this Agreement and to perform fully their obligations hereunder and the transactions contemplated hereby. The Board of Directors of each of Parent and Sub has adopted and approved this Agreement and the transactions contemplated hereby. Parent, as the sole shareholder of Sub, has approved this Agreement and entry into the transactions contemplated hereby. No approval by Parent s stockholders is required to consummate the transactions contemplated hereby. This Agreement has been duly executed and delivered by Parent and Sub and constitutes a valid and binding obligation, enforceable against them in accordance with its terms, except to the extent enforceability may be limited by the effect of applicable bankruptcy, reorganization, insolvency, moratorium or other Laws affecting the enforcement of creditors rights generally and the effect of general principles of equity, regardless of whether such enforceability is considered in a proceeding at Law or in equity.
- (b) Except for (i) filings with the SEC under the Exchange Act, (ii) filings with the Department and the DSOS contemplated herein, and (iii) the filing of a Notification and Report Form under the HSR Act and any similar filings in foreign jurisdictions, the execution, delivery and performance of this Agreement by Parent and Sub and the consummation by Parent and Sub of the transactions contemplated hereby will not (A) violate any provision of the articles or certificate of incorporation or bylaws or similar organizational documents of Parent or Sub, (B) violate, conflict with or result in the breach of any of the terms or conditions of, result in modification of, require any notice or action under, or otherwise give any other contracting party the right to terminate, accelerate obligations under or receive payment under or constitute (or with notice or lapse of time or both constitute) a default under, any instrument, contract or other agreement to which Parent or Sub is a party or to which either of them or any of their respective assets or properties is bound or subject, (C) violate any Law applicable to Parent or Sub or by which any of their respective assets or properties is bound, (D) violate any governmental permit, (E) require any filing with, notice to, or permit, consent or approval of, any Governmental Entity, excluding from the foregoing clauses (B), (C), (D) and (E) violations, conflicts breaches, modifications and defaults which, and filings, notices, permits, consents and approvals the absence of which, in the aggregate, would not reasonably be expected to have a material adverse effect on the ability of Parent and Sub to consummate the transactions contemplated hereby.
- **4.3** <u>Information in the Offer Documents</u>. The information supplied by either of Parent or Sub expressly for inclusion or incorporation by reference in the Offer Documents, the Schedule 14D-9 or the Schedule TO, including any amendments thereof and supplements thereto, will not contain any untrue statement of a material fact or omit to state any material fact

required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading. The Schedule TO, including any amendments thereof and supplements thereto, will comply in all material respects with the provisions of applicable federal securities Laws and Michigan corporate Laws and, on the date filed with the SEC and on the date first published or sent or given to the Company s shareholders, will not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading, except that neither of Parent or Sub makes any representation or warranty with respect to statements made in the Schedule TO, including any amendments thereof and supplements thereto, based on information furnished by the Company expressly for inclusion therein.

- **4.4** <u>Information in the Proxy Statement</u>. The Proxy Statement, if any (and any amendment thereof and supplement thereto), at the date mailed to the Company's shareholders and at the time of any meeting of the Company shareholders to be held in connection with the Merger, will not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading, except that no representation is made by either of Parent or Sub other than with respect to statements made therein based on information supplied by Parent or Sub expressly for inclusion in the Proxy Statement, including any amendments thereof and supplements thereto. The Proxy Statement, as to information supplied by either of Parent or Sub for inclusion therein, will comply in all material respects with the provisions of the Exchange Act and the rules and regulations thereunder and all applicable federal securities and Michigan corporation Laws.
- **4.5** <u>Sub</u>. Sub is duly organized, validly existing and in good standing as a Delaware corporation. Sub has been formed solely for the purpose of engaging in the transactions contemplated by this Agreement.
- **4.6** <u>Financing.</u> Parent currently has available to it, and Parent will cause Sub to have, (i) at the Acceptance Time, sufficient unrestricted funds to pay in full for all Shares validly tendered into the Offer, subject to the terms and conditions of the Offer and this Agreement and (ii) at the Effective Time, sufficient unrestricted funds to pay in full the Merger Consideration, subject to the terms and conditions of this Agreement.
- **4.7** Ownership of Shares. On the date hereof, Parent and Sub own no Shares, and (other than as provided herein) own no additional rights to purchase Shares through any option from any other person. None of Parent, Sub or their respective Affiliates is or ever has been an interested shareholder (as defined in Section 778 of the MBCA) with respect to the Company.
- **4.8** <u>Litigation</u>. As of the date of this Agreement, there are no actions, suits or claims or legal, administrative or arbitration proceedings pending or, to the knowledge of Parent, threatened against Parent or Sub, or any of their officers or directors (in their capacity as such) that individually or in the aggregate would reasonably be expected to prevent, enjoin, alter or delay the Offer or the Merger.

- **4.9** <u>Financial Advisor</u>. Parent, and not the Company, shall be liable for any fee or other commission payable to any broker, finder, agent or similar intermediary engaged by Parent or Sub in connection with the transactions contemplated hereby.
- **4.10** No Additional Representations. Parent acknowledges and agrees that except as expressly set forth in this Agreement, neither the Company nor any of the Company Subsidiaries nor any of their Representatives has made any representation or warranty, express or implied, to Parent or any of its representatives in connection with this Agreement, the Offer, the Merger or any of the other transactions contemplated hereby. Without limiting the generality of the foregoing, Parent acknowledges and agrees that neither the Company nor any of the Company Subsidiaries nor any of their Representatives has made any representation or warranty, express or implied, as to the accuracy or completeness of any information regarding the Company or any of the Company Subsidiaries made available to Parent and its representatives, except as expressly set forth in this Agreement, and neither the Company nor any other person shall be subject to any liability to Parent or any other person, resulting from the Company s having made available to Parent or its Representatives such information, including in the data room, management presentations (formal or informal) or in any other form in connection with the transactions contemplated by this Agreement. Without limiting the foregoing, neither the Company nor any of the Company Subsidiaries nor any of their Representatives makes any representation or warranty to Parent with respect to any financial projection or forecast relating to the Company or any of the Company Subsidiaries.

#### SECTION 5 COVENANTS AND AGREEMENTS

- **5.1** Conduct of Business. During the period from the date of this Agreement and continuing until the earlier of the termination of this Agreement pursuant to its terms or the Effective Time, except (w) as may be required by Law, (x) to the extent Parent shall otherwise consent in writing, which consent shall not unreasonably be withheld, delayed or conditioned, (y) as otherwise expressly provided in this Agreement and (z) as provided in Section 5.1 of the Company Disclosure Schedule, the Company and each Company Subsidiary shall carry on its business in the ordinary course, in substantially the same manner as heretofore conducted. Without limiting the generality of the foregoing, except (w) as may be required by Law, (x) to the extent Parent shall otherwise consent in writing, which consent shall not unreasonably be withheld, delayed or conditioned, (y) as otherwise specifically provided in, or in furtherance of any action permitted by, this Agreement and (z) as provided in Section 5.1 of the Company Disclosure Schedule, during the period from the date of this Agreement and continuing until the earlier of the termination of this Agreement pursuant to its terms or the Effective Time, the Company shall observe the following covenants:
- (a) Affirmative Covenants Pending Closing. The Company and the Company Subsidiaries shall:
- (i) <u>Preservation of Personnel</u>. Except for actions taken in the ordinary course of business consistent with past practices, use commercially reasonable best efforts to preserve intact and keep available the services of present employees of the Company and the Company Subsidiaries; <u>provided</u>, <u>however</u> that nothing in this Section 5.1(a)(i) shall prevent the Company or any Company Subsidiary from terminating the employment of any officer or employee for cause, as determined by the Company or Company Subsidiary, as applicable;

- (ii) <u>Insurance</u>. Use commercially reasonable best efforts to keep in effect casualty, product liability, workers compensation and other insurance policies in coverage amounts substantially similar to those in effect at the date of this Agreement;
- (iii) <u>Preservation of the Business: Maintenance of Properties, Contracts.</u> Use commercially reasonable best efforts to preserve the business of the Company, to develop, commercialize and pursue regulatory approvals for the Company s product candidates and products and to advertise, promote and market the products of the Company and the Company Subsidiaries, and use commercially reasonable best efforts to keep the Company s properties substantially intact, to preserve its goodwill and business, to maintain all physical properties in such operating condition as will permit the conduct of the Company s business on a basis consistent with past practice, and to perform and comply in all material respects with the terms of the contracts referred to in Section 3.11;
- (iv) Intellectual Property Rights. Use commercially reasonable best efforts to preserve and protect the Company Intellectual Property;
- (v) <u>Company Options and Company Restricted Shares</u>. Use commercially reasonable best efforts with respect to Company Options and Company Restricted Shares to effectuate the terms of this Agreement, <u>provided</u>, <u>however</u>, that Parent shall have the right to approve any agreements to modify material terms of the underlying instruments (such approval not to be unreasonably withheld, delayed or conditioned), except for those modifications necessary to carry out the terms of this Agreement; and
- (vi) <u>Regulatory Matters</u>. Notify and consult with Parent promptly (A) after receipt of any material communication from any Governmental Entity or inspections of any manufacturing or clinical trial site and before giving any material submission to a Governmental Entity, and (B) prior to making any material change to a study protocol, adding new trials, making any material change to a manufacturing plan or process, or making a material change to the development timeline for any of its product candidates or programs.
- (b) Negative Covenants Pending Closing. The Company and the Company Subsidiaries shall not:
- (i) <u>Disposition of Assets</u>. Sell or transfer, or mortgage, pledge, lease, license or otherwise encumber any of its assets, including the Company Intellectual Property, other than (x) sales, transfers, mortgages, pledges, leases, licenses or other encumbrances in the ordinary course of business and in amounts not exceeding, in the aggregate, \$250,000 or (y) sales or transfers of inventory in the ordinary course of business, consistent with past practices;
- (ii) <u>Liabilities</u>. Incur any indebtedness for borrowed money in excess of \$50,000 in the aggregate or incur any obligation or liability or enter into any contract or commitment involving potential payments to or by the Company or any Company Subsidiary, other than in the ordinary course of business consistent with past practice, in an amount aggregating in excess of \$150,000;

- (iii) <u>Compensation</u>. Except for the payments made to officers and employees under the bonus plans of the Company as provided in this <u>Section 5.1(b)(iii)</u>, pay any director of the Company or any Company Subsidiary or materially increase the compensation payable to any officer, or key employee, or enter into any material, new employment, severance, retention or other agreement or arrangement with any officer, or key employee of the Company or a Company Subsidiary, or adopt, or increase the benefits (including fringe benefits) under, any employee benefit plan or otherwise, except, in each case, (A) as required by Law or any Plan or in accordance with existing agreements and (B) in the ordinary course of business consistent with past practice, or make any loans to any of its directors, officers or employees, agents or consultants, except in the ordinary course of business consistent with past practices, or make any change in its existing borrowing or lending arrangements for or on behalf of any such persons pursuant to an employee benefit plan or otherwise;
- (iv) <u>Capital Stock</u>. Make any change in the number of shares of its capital stock authorized, issued or outstanding or grant or accelerate the exercisability of any option, warrant or other right to purchase, or convert any obligation into, shares of its capital stock, declare or pay any dividend or other distribution with respect to any shares of its capital stock, sell or transfer any shares of its capital stock, or redeem or otherwise repurchase any shares of its capital stock; <u>provided</u>, <u>however</u>, the foregoing limitations shall not apply in connection with (1) the payment (which shall be in cash) of any dividends required to made under the Company s articles of incorporation, as in effect on the date hereof, (2) the conversion or exercise of convertible securities outstanding on the date of this Agreement, including, without limitation, the issuance of Shares upon the exercise of Company Options, (3) the Top-Up Option and (4) acceleration of the vesting of Company Restricted Shares and Company Options necessary to carry out the terms of this Agreement;
- (v) <u>Articles of Incorporation, By laws, Directors and Officers</u>. Cause, permit or propose any amendments to the articles of incorporation or bylaws of the Company or any Company Subsidiary or, except as required by applicable Law or rules of The NASDAQ Stock Market, elect or appoint any new directors or officers;
- (vi) <u>Acquisitions</u>. Make, or permit to be made, any material acquisition, lease, investment, or capital contribution outside the ordinary course of business consistent with past practice;
- (vii) <u>Capital Expenditures</u>. Authorize any single capital expenditure in excess of \$50,000 or capital expenditures which in the aggregate exceed \$150,000;
- (viii) <u>Accounting Policies</u>. Except as may be required as a result of a change in Law or in generally accepted accounting principles, change any of the accounting practices or principles used by it;
- (ix) <u>Taxes</u>. Make, revise or amend any material Tax election or settle or compromise any material federal, state, local or non-U.S. Tax audit or proceeding, change any annual tax accounting period, change any material method of Tax accounting, enter into any closing agreement relating to any material Tax, file any amended material Tax Return, prepare or file any material Tax Return in a manner inconsistent with past practice, surrender any right to claim a material Tax refund, or consent to any extension or waiver of the limitations period applicable to any material Tax claim or assessment:

- (x) <u>Legal</u>. Commence, settle or compromise any pending or threatened suit, action or claim which (A) is material to the business of the Company and the Company Subsidiaries, taken as a whole or otherwise involves the payment by the Company of an amount in excess of \$100,000, in the aggregate, for all such suits, actions or claims, (B) would involve restrictions on the business activities of the Company or any Company Subsidiary or (C) would involve the issuance of Company securities;
- (xi) <u>Extraordinary Transactions</u>. Adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization of the Company or any of the Company Subsidiaries (other than the Offer and Merger); or take any action to render inapplicable, or to exempt any person from the provisions of the MBCA or any other Law that purports to limit or restrict business combinations or the ability to acquire or vote shares of capital stock;
- (xii) <u>Payment of Indebtedness</u>. Pay, discharge or satisfy any material claims, liabilities or obligations (absolute, accrued, asserted or unasserted, contingent or otherwise), other than the payment, discharge or satisfaction, in the ordinary course of business and consistent with past practice, of liabilities reflected or reserved against in the Company Balance Sheet or incurred in the ordinary course of business since the date of the Company Balance Sheet;
- (xiii) <u>Loans and Advances</u>. Make any loans, advances or capital contributions to, or investments in, any other person (other than to wholly-owned subsidiaries of the Company or customary advances to employees for travel and business expenses in the ordinary course of business);
- (xiv) <u>WARN Act</u>. Effectuate a plant closing or mass layoff, as those terms are defined in the Worker Adjustment and Retraining Notification Act of 1988 or effectuate any similar action under any foreign Law;
- (xv) <u>New Agreements/Amendments</u>. Except as contemplated in this Agreement, enter into or modify, or permit a Company Subsidiary to enter into or modify, any material license, development, research, or collaboration agreement, lease or other similar contract with any other person;
- (xvi) <u>Confidentiality and Non-Competition Agreements</u>. Modify, amend or terminate, or waive, release or assign any material rights or claims with respect to any confidentiality agreement or non-competition agreement to which the Company is a party; or
- (xvii) <u>Authorizations</u>, <u>Agreements and Obligations</u>. Authorize or commit or agree, in writing or otherwise, to take, any of the foregoing actions.
- (c) <u>Advice of Changes</u>. The Company shall promptly advise Parent orally and in writing of any change or event that constitutes a Company Material Adverse Effect.

#### 5.2 No Solicitation.

(a) The Company has, and will cause its Representatives (as defined below) to have, ceased and terminated all existing discussions, negotiations and communications with any persons (other than Parent, Sub or any of their respective Subsidiaries or Representatives) with respect to any offer or proposal relating to any transaction or proposed transaction or series of related transactions, other than the transactions contemplated hereby, involving: (A) any consolidation, tender offer, business combination, merger or similar transaction involving the Company or any Company Subsidiary; (B) any recapitalization, restructuring, liquidation or dissolution of the Company or any Company Subsidiary, (C) any issuance by the Company individually or in the aggregate of over fifteen percent (15%) of its equity securities or (D) any sale, lease, exchange, transfer, license, acquisition or disposition of assets of the Company or its Subsidiaries (including for this purpose the outstanding equity securities of the Company s Subsidiaries) for consideration equal to fifteen percent (15%) or more of the market value of all of the outstanding Shares on the last trading day prior to the date of this Agreement or fifteen percent (15%) of the consolidated total assets of the Company and the Company Subsidiaries (each of clauses (A)-(D), an <u>Acquisition Proposal</u>). Except as provided in this Section 5.2, from the date of this Agreement until the earlier of termination of this Agreement or the Effective Time, the Company shall not and shall not authorize or permit its officers, directors, employees, investment bankers, attorneys, accountants or other agents or those of the Company Subsidiaries (collectively, Representatives ) to directly or indirectly (i) initiate, solicit or knowingly encourage, or knowingly take any action to facilitate the making of, any offer or proposal which constitutes or is reasonably likely to lead to any Acquisition Proposal, (ii) enter into any agreement with respect to any Acquisition Proposal, or (iii) engage in negotiations or discussions with, or provide any non-public information or data to, any person (other than Parent or any of its affiliates or Representatives) relating to any Acquisition Proposal or, except as set forth in Section 5.2(b), grant any waiver or release under any standstill or other agreement. The Company agrees that any material violations of the restrictions set forth in this Section 5.2 by any of its Representatives shall be deemed to be a material breach of this Agreement (including this Section 5.2) by the Company. Notwithstanding the foregoing, nothing contained in this Section 5.2 or any other provision hereof shall prohibit the Company or the Company Board of Directors from taking and disclosing to the Company s shareholders its position with respect to any tender or exchange offer by a third party pursuant to Rules 14d-9 and 14e-2 promulgated under the Exchange Act.

(b) Notwithstanding anything to the contrary set forth in this Agreement, prior to the Acceptance Time, the Company and its Representatives may furnish non-public information regarding the Company to any person pursuant to a confidentiality agreement with terms no less favorable, in the aggregate, to the Company than those contained in the Confidentiality Agreement and may negotiate and participate in discussions and negotiations with such person concerning an Acquisition Proposal if, but only if, such person has, in the absence of any material violation of this Section 5.2 by the Company, submitted a *bona fide* written proposal to the Company relating to any such Acquisition Proposal which the Board of Directors determines in good faith, after consultation with its financial advisor, is or is reasonably likely to lead to a Superior Proposal. From and after the date hereof and prior to the Acceptance Time, the Company shall within twenty-four (24) hours notify Parent in writing in the event that the Company or any of the Company Subsidiaries or Representatives receives

(i) any Acquisition Proposal. (ii) any request for non-public information relating to the Company or any of the Company Subsidiaries other than requests for information in the ordinary course of business and, in the good faith judgment of the Company Board of Directors, unrelated to an Acquisition Proposal, or (iii) any inquiry or request for discussions or negotiations regarding any Acquisition Proposal. The Company shall provide Parent within such twenty-four (24) hour period with the identity of such Person and a copy of such Acquisition Proposal, inquiry or request (or, where such Acquisition Proposal is not in writing, a description of the material terms and conditions of such Acquisition Proposal, inquiry or request), including any material modifications thereto. The Company shall keep Parent reasonably informed in writing (and in any event no later than twenty-four (24) hours after the occurrence) of any material changes, developments, discussions or negotiations relating to any Acquisition Proposal and provide Parent with copies of all documents received or provided by the Company related to such Acquisition Proposal. Without limiting the foregoing, the Company shall within twenty-four (24) hours notify Parent in writing if the Company determines to begin providing non-public information or to engage in discussions or negotiations concerning an Acquisition Proposal. The Company shall not, and shall cause the Company Subsidiaries not to, enter into any agreement with any person subsequent to the date of this Agreement that would restrict the Company s ability to provide such non-public information to Parent, and, if the Company or any Company Subsidiary is a party to any agreement that would prohibit the Company from providing such non-public information to Parent, prior to providing information to, or engaging in discussions or negotiations with, the counterparty to such agreement, the Company will obtain approval from the counterparty to such agreement to allow the Company to provide such information to Parent. The Company shall not, and shall cause the Company Subsidiaries not to, terminate, waive, amend or modify any provision of, or grant permission or request under, any standstill or confidentiality agreement to which it or any of the Company Subsidiaries is or becomes a party; provided, however, that the Company and the Company Subsidiaries may respond to an unsolicited Acquisition Proposal submitted to the Company by a party that is bound by a standstill agreement and may decline to enforce or cause to be enforced its rights under such standstill agreement relating to the submission of such unsolicited Acquisition Proposal if, in either case, the Company Board of Directors determines in good faith, after consultation with outside counsel, that failure to so respond or failure to decline to take such action, as the case may be, would be inconsistent with the fiduciary duties of the Company Board of Directors to the shareholders of the Company under applicable Law. The Company will promptly provide to Parent any non-public information concerning the Company or the Company Subsidiaries provided or made available pursuant to this Section 5.2(b) which was not previously provided or made available to Parent.

(c) Except as set forth herein, neither the Company Board of Directors nor any committee thereof shall (i) withdraw, withhold, qualify or modify, or propose publicly to withdraw, withhold, qualify or modify, in a manner adverse to the transactions contemplated by this Agreement to Parent or Sub, the Company Board Recommendation, (ii) approve or recommend or propose publicly to approve or recommend, any Acquisition Proposal (any action referred to in the foregoing clauses (i) and (ii) being referred to as an Adverse Recommendation Change ) or (iii) enter into any written agreement providing for the consummation of any Acquisition Proposal (an Alternative Acquisition Agreement ).

(d) Notwithstanding anything in this Agreement to the contrary, prior to the Acceptance Time, if (i) the Company receives a written, bona fide Acquisition Proposal from a third party, (ii) a material breach by the Company of this Section 5.2 has not contributed to the making of an Acquisition Proposal and (iii) the Company Board of Directors concludes in good faith, after consultation with outside counsel and its financial advisors, such Acquisition Proposal constitutes a Superior Proposal after giving effect to all of the adjustments to the terms of this Agreement which have been offered by Parent (if any) pursuant to this Section 5.2, the Company Board of Directors may, if it determines in good faith, after consultation with outside counsel, that failure to take such action would be inconsistent with the fiduciary duties of the Company Board of Directors to the shareholders of the Company under applicable Law, (A) effect an Adverse Recommendation Change and/or (B) terminate this Agreement to enter into an Alternative Acquisition Agreement with respect to such Superior Proposal; provided, however, that the Company shall not terminate this Agreement pursuant to the foregoing clause (B), and any purported termination pursuant to the foregoing clause (B) shall be void and of no force or effect, unless in advance of or substantially concurrently with such termination the Company (1) pays the fee required by and pursuant to the terms of Section 8.2 and (2) immediately following such termination enters into a binding definitive agreement providing for such Superior Proposal; and provided, further, that the Company Board of Directors may not effect an Adverse Recommendation Change pursuant to the foregoing clause (A) or terminate this Agreement pursuant to the foregoing clause (B) unless (1) the Company shall not have breached in any material respect this Section 5.2, (2) the Company shall have provided prior written notice to Parent, at least three (3) business days in advance (the Notice Period ), of the Company s intention to take such action with respect to such Superior Proposal, which notice shall specify the material terms and conditions of any such Superior Proposal (including the identity of the party making such Superior Proposal), and shall have contemporaneously provided a copy of the proposed Alternative Acquisition Agreement with respect to such Superior Proposal, (3) prior to effecting such Adverse Recommendation Change or terminating this Agreement to enter into a definitive Alternative Acquisition Agreement with respect to such Superior Proposal, the Company shall, and shall cause its Representatives to, during the Notice Period, negotiate with Parent in good faith (to the extent Parent desires to negotiate) to make such adjustments in the terms and conditions of this Agreement so that such Acquisition Proposal ceases to constitute a Superior Proposal, and (4) following any negotiation described in the immediately preceding clause (3), such Acquisition Proposal continues to constitute a Superior Proposal. In the event of any material revisions to the terms of the Superior Proposal after the start of the Notice Period, the Company shall be required to deliver a new written notice to Parent and to comply with the requirements of this Section 5.2 with respect to such new written notice, and the Notice Period shall be deemed to have re-commenced on the date of such new notice. Notwithstanding the foregoing, if fewer than three (3) business days remains before the then scheduled expiration date of the Offer, the Notice Period with respect to the Company Board of Directors effecting an Adverse Recommendation Change pursuant to the foregoing clause (A) shall equal twenty-four (24) hours, provided, however, that, in such a circumstance the Notice Period with respect to the Company terminating this Agreement pursuant to the foregoing clause (B) shall remain three (3) business days. Any Adverse Recommendation Change shall not change the approval of the Company Board of Directors for purposes of causing any state takeover statute or other state Law to be inapplicable to the transactions contemplated by this Agreement, including each of the Offer and the Merger or by the Tender and Voting Agreement with the CEO Shareholder. For

purposes of this Agreement, a <u>Superior Proposal</u> is a *bona fide* written Acquisition Proposal to acquire at least fifty percent (50%) of the equity securities of the Company or at least fifty percent (50%) of the consolidated total assets of the Company and the Company Subsidiaries, (A) on terms which the Company Board of Directors determines in its good faith judgment (after consultation with its financial and legal advisors) to be more favorable to the holders of Shares than the Offer and the Merger taking into account all financial, regulatory, legal and other aspects of such proposal, and (B) which the Company Board of Directors has determined in its good faith judgment (after consultation with its financial and legal advisors) to be reasonably capable of being completed on the terms proposed, taking into account all financial, regulatory, legal and other aspects of such proposal.

- (e) Notwithstanding the foregoing, the Company Board of Directors may withdraw or modify the Company Board Recommendation in the absence of a Superior Proposal if the Company Board of Directors has concluded in good faith, after consultation with its outside counsel, that failure to so withdraw or modify the Company Board Recommendation would be inconsistent with the fiduciary obligations of the Company Board of Directors to the shareholders of the Company under applicable Law, provided, however, that the Company Board of Directors shall not so withdraw or modify the Company Board Recommendation unless the Company has (A) provided to Parent at least three (3) business days prior written notice (or such shorter period as remains prior to the then-scheduled expiration date of the Offer) advising Parent that the Company Board of Directors intends to take such action and specifying the reasons therefore in reasonable detail and (B) during such three (3) business day period, or shorter period, as the case may be, if requested by Parent, engaged in good faith negotiations with Parent to amend this Agreement in such a manner that obviates the need or reason for the withdrawal or modification.
- (f) The Company shall promptly (but in no event later than two (2) business days after the date of this Agreement) demand that each person that, as of the date of this Agreement, has executed a confidentiality agreement in connection with a potential Acquisition Proposal return or destroy, in each case, to the extent permitted by the terms of the applicable confidentiality agreement, all confidential information heretofore furnished to such individual or entity by or on behalf of the Company or a Company Subsidiary.
- (g) Nothing contained in this Agreement shall prohibit the Company or the Company Board of Directors from (i) disclosing to the Company s shareholders a position contemplated by Rules 14D-9 and 14e-2(a) promulgated under the Exchange Act or (ii) making any disclosures to its shareholders if the Company Board of Directors has reasonably determined in good faith, after consultation with outside legal counsel, that failure to do so could be inconsistent with any applicable Law. In addition, it is understood and agreed that, for purposes of this Agreement, a factually accurate public statement by the Company that describes the Company's receipt of an Acquisition Proposal and the operation of this Agreement with respect thereto, or any stop, look and listen communication by the Company Board of Directors pursuant to Rule 14d-9(f) of the Exchange Act or any similar communication to the shareholders of the Company, shall not constitute an Adverse Recommendation Change or an approval or recommendation with respect to any Acquisition Proposal.

#### SECTION 6 ADDITIONAL AGREEMENTS

- **6.1** Proxy Statement. If, following the Tender Completion Time, the adoption and approval of this Agreement by the holders of Shares is required under the MBCA in order to consummate the Merger, then in accordance with applicable Law, the Company shall use commercially reasonable best efforts to, as soon as practicable following the Tender Completion Time, prepare and file with the SEC the Proxy Statement in preliminary form, and each of the Company, Parent and Sub shall use all commercially reasonable best efforts to respond as promptly as practicable to any comments of the SEC with respect thereto. The Company shall notify Parent promptly of the receipt of any comments from the SEC or its staff and of any request by the SEC or its staff for amendments or supplements to the Proxy Statement or for additional information and shall supply Parent with copies of all correspondence between the Company or any of its representatives, on the one hand, and the SEC or its staff, on the other hand, with respect to the Proxy Statement. If at any time prior to receipt of the adoption and approval of this Agreement by the required vote of the holders of the outstanding Shares (the Company Shareholder Approval ) there shall occur any event that should be set forth in an amendment or supplement to the Proxy Statement, the Company shall use commercially reasonable effort to promptly prepare and mail to its shareholders such an amendment or supplement to the extent required by applicable Law. The Company shall not mail any Proxy Statement, or any amendment or supplement thereto, to which Parent reasonably objects. The Company shall use all commercially reasonable best efforts to cause the Proxy Statement to be mailed to the Company s shareholders as promptly as practicable after filing with the SEC. Subject to the terms and conditions of this Agreement, the Proxy Statement shall contain the recommendation of the Company Board of Directors in favor of the Merger. For purposes of this Agreement, the term <u>Tender Completion Time</u> means the latest to occur of (x) the Acceptance Time, (y) the closing of the purchase of the Top-Up Shares or the failure by Sub to exercise the Top-Up Option during the exercise period provided in Section 1.10 and (z) if at least one subsequent offering period is commenced by Sub, the expiration of the last subsequent offering period related to the Offer.
- **6.2** Meeting of Shareholders of the Company. If, following the Tender Completion Time, the adoption and approval of this Agreement by the holders of Shares is required under the MBCA in order to consummate the Merger, the Company shall take all actions in accordance with applicable Law, the Company s articles of incorporation and bylaws and the rules of The NASDAQ Stock Market to promptly and duly call, give notice of, convene and hold as promptly as practicable, a Special Meeting.
- **6.3** NASDAQ: Post-Closing SEC Reports. Prior to the Effective Time, the Company shall cooperate with Parent and use commercially reasonable best efforts to take, or cause to be taken, all actions, and do or cause to be done all things, reasonably necessary, proper or advisable on its part under applicable Laws and rules and policies of The NASDAQ Stock Market to enable the delisting by the Surviving Corporation of the Shares from The NASDAQ Stock Market and the deregistration of the Shares under the Exchange Act promptly after the Effective Time. Parent will use all commercially reasonable best efforts to cause the Surviving Corporation to file with the SEC (a) a Form 25 on the Closing Date (if not filed by The NASDAQ Stock Market on behalf of the Company) and (b) a Form 15 on the first business day that is at least ten (10) days after the date the Form 25 is filed (such period between the Form 25

filing date and the Form 15 filing date, the <u>Delisting Period</u>). If the Surviving Corporation is reasonably likely to be required to file any reports pursuant to the Exchange Act during the Delisting Period, the Company will deliver to Parent at least five (5) business days prior to the Closing a substantially final draft of any such reports reasonably likely to be required to be filed during the Delisting Period (<u>Post-Closing SEC Reports</u>). The Post-Closing SEC Reports provided by the Company pursuant to this Section 6.3 will (i) not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading and (ii) comply in all material respects with the provisions of applicable Laws.

**6.4** Access to Information. Prior to the Effective Time, Parent shall be entitled, through its employees and representatives, to have such access to the assets, properties, business and operations of the Company and the Company Subsidiaries as is reasonably necessary or appropriate in connection with Parent s investigation of the Company with respect to the transactions contemplated hereby. Any such investigation and examination shall be conducted at reasonable times during business hours upon reasonable advance notice and under reasonable circumstances so as to minimize disruption to or impairment of the Company s business. No investigation by Parent shall diminish or obviate any of the representations, warranties, covenants or agreements of the Company contained in this Agreement. The Company shall furnish the representatives of Parent during such period with all such information and copies of such documents concerning the affairs of the Company as such representatives may reasonably request and cause its officers, employees, consultants, agents, accountants and attorneys to cooperate fully with such representatives in connection with such investigation. The information and documents so provided shall be subject to the terms of the Confidentiality Agreement. Neither the Company nor any Company Subsidiary shall be required to provide access to or to disclose information where such access or disclosure would contravene any Law or binding agreement entered into prior to the date of this Agreement or would reasonably be expected to violate or result in a loss or impairment of any attorney-client or work product privilege.

**6.5** <u>Public Disclosure</u>. The initial press release concerning the Offer and the Merger shall be a joint press release and, thereafter, subject to the other terms of this Agreement (including Section 5.2), so long as this Agreement is in effect, neither Parent, Sub nor the Company will disseminate any press release or other announcement concerning the Merger, the Offer or this Agreement or the other transactions contemplated by this Agreement to any third party, except as may be required by Law or by any listing agreement with The NASDAQ Stock Market, without the prior consent of each of the other parties hereto, which consent shall not be unreasonably withheld, conditioned or delayed. The parties have agreed to the text of the joint press release announcing the execution of this Agreement. Notwithstanding the foregoing, without prior consent of the other party, each party (a) may communicate information that is not confidential to the other party with financial analysts, investors and media representatives in a manner consistent with its past practice in compliance with applicable Law and (b) may disseminate material substantially similar to material included in a press release or other document previously approved for external distribution by the other parties. Each party agrees to promptly make available to the other parties copies of any written communications made without prior consultation with the other parties.

### **6.6** Regulatory Filings; Reasonable Efforts.

(a) Subject to the terms and conditions of this Agreement, each of the Company, Parent and Sub shall use their commercially reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, and to assist and cooperate with the other parties in doing, all things necessary, proper or advisable under applicable Law to consummate the transactions contemplated by this Agreement, including (i) the obtaining of all necessary actions or nonactions, waivers, consents and approvals from Governmental Entities and the making of all necessary registrations and filings (including filings with Governmental Entities, if any) and the taking of all reasonable steps as may be necessary to obtain an approval or waiver from, or to avoid an action or proceeding by, any Governmental Entities, (ii) the delivery of required notices to, and the obtaining of required consents or waivers from, third parties and (iii) the execution and delivery of any additional instruments necessary to consummate the Offer and the Merger and to fully carry out the purposes of this Agreement. Each of Parent and Sub agrees that, between the date of this Agreement and the Closing Date, each of Parent and Sub shall not, and shall ensure that none of its subsidiaries or other Affiliates shall, take any action or propose, announce an intention or agree, in writing or otherwise, to take any action that would reasonably be expected to materially delay or prevent the consummation of the transactions contemplated hereby. The Company and its Board of Directors shall, if any state takeover statute or similar statute or regulation is or becomes applicable to the Offer, the Merger, this Agreement or any of the other transactions contemplated hereby, use all commercially reasonable best efforts to ensure that the Offer, the Merger and the other transactions contemplated hereby may be consummated as promptly as practicable on the terms contemplated by this Agreement and otherwise to minimize the effect of such statute or regulation on the Offer, the Merger, this Agreement and the other transactions contemplated hereby. The Company will use commercially reasonable best efforts to obtain any consent, approval, or waiver with respect to the Material Contracts such that no party will have a reasonable basis to maintain that the representations and warranties in Section 3.11 and clause (ii) of Section 3.17 are not true and correct.

(b) In furtherance and not in limitation of the undertakings pursuant to this Section 6.6, each of Parent and the Company shall (i) provide or cause to be provided as promptly as practicable to Governmental Entities with regulatory jurisdiction over enforcement of any Antitrust Laws (each such Governmental Entity, a Governmental Antitrust Entity ) information and documents requested by any Governmental Antitrust Entity or necessary, proper or advisable to permit consummation of the transactions contemplated by this Agreement, including preparing and filing any notification and report form and related material required under the HSR Act and any additional consents and filings under any Antitrust Laws as promptly as practicable following the date of this Agreement (but in no event more than ten (10) business days from the date hereof except by mutual consent confirmed in writing) and thereafter to respond as promptly as practicable to any request for additional information or documentary material that may be made under the HSR Act and any additional consents and filings under any Antitrust Laws; and (ii) use their commercially reasonable best efforts to take such actions as are necessary or advisable to obtain prompt approval of consummation of the transactions contemplated by this Agreement by any Governmental Entity. As used in this Agreement, Antitrust Laws means applicable federal, state, local or foreign antitrust, competition, premerger notification or trade regulation laws, regulations, orders, injunctions, judgments, decrees, rulings or other similar requirements enacted, adopted, promulgated or applied by a Governmental Entity.

- (c) Notwithstanding anything to the contrary herein, in connection with the receipt of any necessary governmental approvals or clearances (including under any Antitrust Law), nothing in this Agreement shall require Parent or its subsidiaries to, nor shall the Company or any Company Subsidiary without the prior written consent of Parent, agree or proffer to, divest, hold separate, or enter into any license or similar agreement with respect to, or agree to restrict the ownership or operation of, or agree to conduct or operate in a specified manner, any portion of the business or assets of Parent, the Company or any of their respective subsidiaries.
- (d) Subject to applicable Law relating to the exchange of information, the Company and Parent and their respective counsel shall (i) have the right to review in advance, and to the extent practicable each shall consult the other on, any filing made with, or written materials to be submitted to, any Governmental Entity in connection with the transactions contemplated by this Agreement, (ii) promptly inform each other of any communication (or other correspondence or memoranda) received from, or given to, the U.S. Department of Justice, the U.S. Federal Trade Commission, or any other Governmental Antitrust Entity and (iii) promptly furnish each other with copies of all correspondence, filings and written communications between them or their subsidiaries or affiliates, on the one hand, and any Governmental Entity or its respective staff, on the other hand, with respect to the transactions contemplated by this Agreement. The Company and Parent shall, to the extent practicable, provide the other party and its counsel with advance notice of and the opportunity to participate in any discussion, telephone call or meeting with any Governmental Entity in respect of any filing, investigation or other inquiry in connection with the transactions contemplated by this Agreement and to participate in the preparation for such discussion, telephone call or meeting. None of Parent, Sub or the Company shall commit to or agree with any Governmental Entity to stay, toll or extend any applicable waiting period under the HSR Act or applicable foreign competition Laws, without the prior written consent of the other party. The Company and Parent may, as each deems advisable and necessary, reasonably designate any competitively sensitive material provided to the other under this Section 6.6 as Antitrust Counsel Only Material.

  Notwithstanding anything to the contrary in this Section 6.6, materials provided to the other party or its counsel may be redacted to remove references concerning the valuation of the Company and the Company Subsidi
- 6.7 Notification of Certain Matters. Each party shall give prompt notice to the other parties of (i) the occurrence or non-occurrence of any event of which such party has knowledge the occurrence or non-occurrence of which would cause any representation or warranty made by such party in this Agreement to be untrue or inaccurate in any material respect at any time from the date hereof to the Acceptance Time, (ii) any Offer Condition of which such party has knowledge is unsatisfied in any material respect as of any scheduled Expiration Date, and (iii) any material failure of such party or any of its Representatives of which such party has knowledge to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it hereunder; provided, however, that no such notification shall affect the representations, warranties, covenants or agreements of the parties, the conditions to the obligations of the parties under this Agreement or the remedies available to the party receiving such notification. Without limiting the foregoing, the Company shall, within twenty-four (24)

hours after it has notice of any of the following notify Parent of (1) any notice or other communication from any person alleging that the consent of such person is or may be required in connection with the transactions contemplated by this Agreement, (2) any notice or other communication from any Governmental Entity in connection with the transactions contemplated by this Agreement, and (3) any actions, suits, claims, investigations or proceedings instituted or threatened against the Company or any of its directors, officers or Affiliates, including by any shareholder of the Company, before any court or Governmental Entity, relating to or involving or otherwise affecting the Company or any Company Subsidiaries which, if pending on the date of this Agreement, would have been required to have been disclosed pursuant to this Agreement or relating to this Agreement or the transactions contemplated hereby, or seeking damages or discovery in connection with such transactions.

**6.8** <u>Litigation</u>. Each party shall notify the other parties within twenty-four (24) hours if and after it receives notice of any actions, suits, claims, investigations or proceedings instituted or threatened against Parent, Sub or the Company or any of their respective directors, officers or Affiliates, including by any shareholder of the Company, before any court or Governmental Entity relating to this Agreement, the Tender and Voting Agreement, the Offer, the Merger or the transactions contemplated hereby, or seeking damages or discovery in convection with such transactions. Until the earlier of the termination of this Agreement in accordance with its terms or the Effective Time, each party shall give the other parties the opportunity to participate in the defense or settlement of any shareholder litigation relating to this Agreement or any of the transactions contemplated by this Agreement, and shall not settle any such litigation without such other party s written consent, which will not be unreasonably withheld, conditioned or delayed.

### 6.9 Indemnification.

(a) Parent agrees that any rights to indemnification or exculpation now existing in favor of the directors or officers of the Company and the directors or officers of each Company Subsidiary (the Indemnified Parties and, each, an Indemnified Party) as provided in their respective organizational documents, in effect as of the date hereof, with respect to matters occurring at or prior to the Acceptance Time shall survive the Merger and shall continue in full force and effect for a period of six (6) years after the Acceptance Time, and Parent guarantees any such obligations of the Surviving Corporation. During such period, Parent shall not, nor shall it permit the Surviving Corporation to, amend, repeal or otherwise modify such provisions for indemnification in any manner that would materially and adversely affect the rights thereunder of individuals who at any time on or prior to the Acceptance Time were directors or officers of the Company or directors or officers of any of Company Subsidiary in respect of actions or omissions occurring at or prior to the Acceptance Time (including, without limitation, the transactions contemplated by this Agreement), unless such modification is required by Law; provided, however, that in the event any claim or claims are asserted or made either prior to the Acceptance Time or within such six (6) year period, all rights to indemnification in respect of any such claim or claims shall continue until disposition of any and all such claims. Each Indemnified Party will be entitled to advancement of expenses (including attorneys fees) incurred in the defense of any matter that the Indemnified Party is entitled to indemnification hereunder from the Surviving Corporation to the fullest extent that the Company is obligated to advance expenses to such individuals on the date hereof pursuant to the

Company s articles of incorporation and bylaws as in effect on the date hereof; provided that any Indemnified Party to whom expenses are advanced provides an undertaking, to the extent required by the MBCA, to repay such advances if it is determined by a final determination of a court of competent jurisdiction (which determination is not subject to appeal) that such Indemnified Party is not entitled to indemnification under applicable Law.

(b) Subject to the next sentence, the Surviving Corporation shall, and Parent shall cause the Surviving Corporation to, (i) maintain, at no expense to the beneficiaries, in effect for six (6) years from the Effective Time, the current policies of the directors—and officers—liability insurance maintained by the Company (the \_\_Current D&O Insurance\_\_) with respect to matters existing or occurring at or prior to the Effective Time (including the transactions contemplated by this Agreement), so long as the annual premium therefor would not be in excess of two hundred percent (200%) of the last annual premium paid prior to the Effective Time (such two hundred percent (200%), the \_\_Maximum Premium\_\_), or (ii) purchase a six (6) year extended reporting period endorsement with respect to the Current D&O Insurance (a \_\_Reporting Tail Endorsement\_) and maintain such endorsement in full force and effect for its full term, provided, however, that prior to the Surviving Corporation taking any actions in clauses (i) or (ii) above, Parent shall be provided the opportunity to purchase, in lieu thereof, a substitute policy with the same coverage limits and substantially similar terms as in the Reporting Tail Endorsement proposed to be purchased by the Surviving Corporation. If the Company\_s or the Surviving Corporation sexisting insurance expires, is terminated or canceled during such six (6) year period or exceeds the Maximum Premium, the Surviving Corporation shall obtain, and Parent shall cause the Surviving Corporation to obtain, as much directors\_\_ and officers\_\_ liability insurance as can be obtained for the remainder of such period for an annualized premium not in excess of the Maximum\_\_ Premium, on terms and conditions no less advantageous to the Indemnified Parties than the Company\_s existing directors\_\_ and officers\_\_ liability insurance.

(c) The obligations under this Section 6.9 shall not be terminated or modified in such a manner as to adversely affect any indemnitee to whom this Section 6.9 applies without the consent of such affected indemnitee (it being expressly agreed that the indemnitees to whom this Section 6.9 applies shall be third party beneficiaries of this Section 6.9 and shall be entitled to enforce the covenants contained herein).

### 6.10 Directors.

(a) Promptly upon the payment by Sub for any Shares accepted by Sub for payment pursuant to the Offer at the Acceptance Time, which Shares represent at least a majority of the issued and outstanding Shares pursuant to the Offer, Parent shall be entitled to designate such number of directors on the Company Board of Directors as will give Parent, subject to compliance with Section 14(f) of the Exchange Act, representation on the Company Board of Directors equal to at least that number of directors, rounded up to the next whole number, which is the product of (x) the total number of directors on the Company Board of Directors (giving effect to the directors elected pursuant to this sentence) multiplied by (y) the percentage that (I) such number of Shares so accepted for payment and paid for by Sub plus the number of Shares otherwise owned by Parent, Sub or any other subsidiary of Parent bears to (II) the number of such Shares outstanding, and the Company shall, at such time, cause Parent s designees to be so elected; provided, however, that in the event that Parent s designees are appointed or elected to

the Company Board of Directors, until the Effective Time, the Company Board of Directors shall have at least three (3) directors who are directors on the date of this Agreement and who are independent directors for the purposes of NASDAO listing requirements (the Independent Directors ); and provided, further, that, in such event, if the number of Independent Directors shall be reduced below three (3) for any reason whatsoever, any remaining Independent Directors (or Independent Director, if there shall be only one remaining) shall be entitled to designate persons to fill such vacancies who shall be deemed to be Independent Directors for purposes of this Agreement. The Company may designate, prior to the Acceptance Time, two alternate Independent Directors that the Company Board of Directors shall appoint in the event of the death, disability or resignation of the Independent Directors, each of whom shall, following such appointment to the Company Board of Directors, shall be deemed to be an Independent Director pursuant to this Section 6.10. From and after the Acceptance Time, the Company and Parent shall use commercially reasonable best efforts, as permitted by applicable Laws and the rules of NASDAQ Global Select Market, upon Parent s request, also cause persons elected or designated by Parent to constitute the same percentage (rounded up to the next whole number) as is on the Company Board of Directors of (i) each committee of the Company Board of Directors, (ii) each board of directors (or similar body) of each of the Company s Subsidiaries, and (iii) each committee (or similar body) of each such board, in each case only to the extent permitted by applicable Law or the rules of any stock exchange on which the Shares are listed. Subject to applicable Law, the Company and Parent shall take all action requested by Parent necessary to effect any such election, including mailing to its shareholders the Information Statement containing the information required by Section 14(f) of the Exchange Act and Rule 14f-1 promulgated thereunder, and the Company shall make such mailing with the mailing of the Schedule 14D-9 (provided that Parent and Sub shall have provided to the Company on a timely basis, and shall be solely responsible for, all information required to be included in the Information Statement with respect to Parent, Sub and Parent s designees). In connection with the foregoing, the Company shall promptly, at the option of Sub, either increase the size of the Company Board of Directors or obtain the resignation of such number of its current directors, or both, as is necessary to enable Parent s designees to be elected or appointed to the Company Board of Directors as provided above.

(b) Notwithstanding anything in this Agreement to the contrary, if Parent's designees constitute a majority of the Company Board of Directors after the Acceptance Time and prior to the Effective Time, then the affirmative vote of a majority of the Independent Directors (or if only one (1) exists, then the vote of such Independent Director) shall be required to (i) amend or terminate this Agreement by the Company, (ii) exercise or waive any of the Company's rights, benefits or remedies hereunder, if such action would materially and adversely affect holders of Shares other than Parent or Sub, (iii) amend the certificate of incorporation or bylaws of the Company, or (iv) take any other action of the Company Board of Directors under or in connection with this Agreement or the transactions contemplated hereby; provided, however, that if there shall be no Independent Directors as a result of such persons deaths, disabilities or refusal to serve, then such actions may be effected by majority vote of the entire Company Board of Directors.

**6.11**  $\underline{401(k)}$ . Except with the prior written consent of Parent, during the period from the date of this Agreement to the Effective Time, the Company shall not (i) make any discretionary contribution to the Company s 401(k) plan, other than employer matching contributions at the rate in effect immediately prior to the date of this Agreement, or (ii) make any required contribution to the Company s 401(k) plan in Shares. If requested by Parent in writing at least 10 days prior to the Effective Time, the Company shall terminate the Company s 401(k) plan immediately prior to the Effective Time.

### **6.12** Employee Benefit Matters.

- (a) From the Effective Time and for a period of twelve (12) months following the Effective Time, Parent shall or shall cause the Surviving Corporation to either (i) continue the Plans or certain of the Plans if the provisions of 6.12(a)(iii) apply, (ii) permit employees of the Company and any Company Subsidiary who remain in employment with Parent or Surviving Corporation or their respective subsidiaries following the Effective Time (\_Continuing Employees ) while they remain so employed by Parent or Surviving Corporation or their respective subsidiaries, and as applicable, their eligible dependents, to participate in the employee benefit plans, programs and policies (including without limitation any plan intended to qualify within the meaning of Section 401(a) of the Code and any bonus, severance, vacation, sick, personal time off plans or programs but excluding any equity compensation plans, programs, agreements or arrangements) of Parent or its Affiliates, or (iii) a combination of clauses (i) and (ii), in all cases, such employee benefit plans, programs or policies shall be substantially comparable in the aggregate to those provided to such employees immediately preceding the date of this Agreement. From the Effective Time and for a period of twelve (12) months following the Effective Time, Parent also shall provide to Continuing Employees total compensation opportunities (including base salary and target bonus opportunities) that are substantially comparable in the aggregate to the total compensation opportunities such employees had prior to the date of this Agreement. From and after the Acceptance Time, Parent shall, and shall cause the Company and Surviving Corporation to, honor the severance plan of the Company attached to Section 6.12 of the Company Disclosure Schedule for a period of at least one (1) year following the Effective Time, From and after the Acceptance Time, Parent shall, and shall cause the Surviving Corporation to, pay at the Effective Time, the bonuses pursuant to the Plans attached to S
- (b) To the extent Parent elects to have Continuing Employees and their eligible dependents participate in its or its Affiliate s employee benefit plans, program or policies following the Effective Time, Parent shall, or shall cause the Surviving Corporation or its subsidiaries to, recognize the prior service with the Company or any Company Subsidiary, including prior service with predecessor employers where such prior service is recognized by the Company and the Company Subsidiaries as of immediately prior to the Effective Time, of each Continuing Employee in connection with all employee benefit plans, programs or policies of Parent or its Affiliates in which Continuing Employees are eligible to participate for purposes of eligibility to participate, entitlement to benefits, vesting and determination of level of benefits (but not (A) for purposes of vesting in any stock options or other equity awards or (B) to the extent that such recognition would, as reasonably determined by Parent, result in the duplication of benefits or a result that is inconsistent with the terms of this Agreement).
- (c) From and after the Effective Time, Parent shall, or shall cause the Surviving Corporation or its subsidiaries to, cause any pre-existing conditions or limitations, evidence of insurability requirements and eligibility waiting periods under any welfare plans of

Parent or its Affiliates to be waived with respect to Continuing Employees and their eligible dependents to the extent such Continuing Employees and their eligible dependents were not subject to such preexisting conditions and limitations, evidence of insurability requirements and eligibility waiting periods under the comparable Plans as of the time immediately preceding the date of this Agreement provided, however, that the waiver of any evidence of insurability requirement under a welfare plan of Parent or its Affiliates is subject to the terms of the applicable plan or contract and the approval of the insurer of the underlying benefit. Parent shall, and shall cause Sub and Surviving Corporation to, give credit to Continuing Employees under the applicable plans of Parent, Sub or the Surviving Corporation (and their respective Affiliates) for amounts paid by such Continuing Employees prior to the Effective Time under a Plan administered by the Company during fiscal year 2010 for purposes of applying deductibles, co-payments and out-of-pocket maximums as though such payments had been paid in accordance with the terms and conditions of the plans of the Parent, Sub, Surviving Corporation or their respective Affiliates.

(d) No provision of this Agreement shall create any third party beneficiary rights in any employee, any beneficiary or dependents thereof, or any collective bargaining representative thereof, with respect to the compensation, terms and conditions of employment and benefits that may be provided to any employee by Company, Parent or its Affiliates or the Surviving Corporation or under any benefit plan which Company, Parent or its Affiliates or the Surviving Corporation may maintain.

### SECTION 7 CONDITIONS PRECEDENT TO THE OBLIGATION OF PARTIES TO

### CONSUMMATE THE MERGER

- **7.1** Conditions to Obligations of Each Party to Effect the Merger. The respective obligations of each party to this Agreement to effect the Merger shall be subject to the satisfaction or written waiver at or prior to the Closing Date of the following conditions:
- (a) <u>Shareholder Approval</u>. This Agreement shall have been adopted and approved and the Merger thereby approved by the requisite vote of the holders of the Shares, to the extent required pursuant to the requirements of the articles of incorporation and the MBCA; <u>provided</u> that Parent and Sub shall, and shall cause any other Affiliate of Parent to, vote all Shares held by them in favor of this Agreement, the Merger and the transactions contemplated herein.
- (b) <u>Statutes: Court Orders</u>. No statute, rule, executive order or regulation shall have been enacted, issued, enforced or promulgated by any Governmental Entity which prohibits the consummation of the Merger, and there shall be no order or injunction of a court of competent jurisdiction in effect preventing consummation of the Merger.
- (c) Consummation of Offer. The Shares validly tendered and not withdrawn pursuant to the Offer shall have been accepted for payment and paid for pursuant to the Offer and the terms of this Agreement; provided, however, that neither Parent nor Sub shall be entitled to assert the failure of this condition if, in breach of this Agreement or the terms of the Offer, Sub shall have failed to purchase any of the Shares validly tendered and not withdrawn pursuant to the Offer.

**7.2** Frustration of Conditions. None of the Company, Parent or Sub may rely on the failure of any condition set forth in Section 7.1 or in Annex I to be satisfied if such failure was caused by such party s failure to act in good faith or use its commercially reasonable best efforts to consummate the Offer and the Merger and the other transactions contemplated by this Agreement, as contemplated by Section 6.6.

### SECTION 8 TERMINATION, AMENDMENT AND WAIVER

- **8.1** Termination. This Agreement may be terminated and the transactions contemplated hereby may be abandoned:
- (a) By mutual written consent of Parent and the Company duly authorized by the Board of Directors of Parent and the Company Board of Directors at any time before the Effective Time, whether before or after shareholder adoption of this Agreement; or
- (b) By either Parent or the Company:
- (i) if at any time before the Acceptance Time a court of competent jurisdiction or other Governmental Entity shall have issued an order, decree or ruling or taken any other action, and such order, decree or ruling or other action shall have become final and non-appealable, or there shall exist any statute, rule or regulation, in each case, permanently restraining, enjoining or otherwise prohibiting the consummation of the Offer or the Merger;
- (ii) if the Acceptance Time has not occurred by December 15, 2010 (the Outside Date); provided, however, that the right to terminate this Agreement pursuant to this Section 8.1(b)(ii) shall not be available to any party whose failure (or the failure of such party s Affiliate) to fulfill any obligation under this Agreement has been the principal cause of, or resulted in, the failure of the Acceptance Time to have occurred by such date; provided, further, that if as of the Outside Date, all conditions set forth in Annex I have been satisfied or waived (other than both of the conditions set forth in clauses (i) and (ii) of Annex I), then either the Company or Parent may extend the Outside Date to March 15, 2011;
- (iii) if Company Shareholder Approval shall not have been obtained at the Special Meeting duly convened therefor or at any adjournment or postponement thereof; <u>provided</u>, <u>however</u> that the right to terminate under this Section 8.1(b)(iii) shall not be available to Parent or Sub if the Shares owned by Parent, Sub or any of their respective Affiliates shall not have been voted in favor of adopting this Agreement; or
- (c) By Parent if, prior to the Acceptance Time, there has been a breach by the Company of or inaccuracy in, any representation, warranty, covenant or agreement of the Company set forth in this Agreement, which breach or inaccuracy has resulted in the conditions set forth in paragraphs (c) or (e) of Annex I not being satisfied (and such breach or inaccuracy has not been cured or such condition has not been satisfied within twenty (20) days after the receipt of notice thereof or such breach or inaccuracy is not reasonably capable of being cured or such condition is not reasonably capable of being satisfied within such period), provided that such occurrence or circumstance shall not have resulted from or been caused by the failure of Parent or Sub to perform, in any material respect, any of their covenants or agreements contained in this Agreement, or Parent and Sub have substantially breached any of their representations and warranties contained in this Agreement; or

(d) By the Company if (i) prior to the Acceptance Time, there has been a breach by Parent or Sub of any representation, warranty, covenant or other agreement contained herein, or if a representation or warranty of Parent or Sub is no longer true and correct, which situation in any case (x) would result in any of the representations and warranties of Parent and Sub set forth in this Agreement not being true and correct (without giving effect to any limitation as to materiality or material adverse effect or similar terms set forth therein) except where the failure to be so true and correct does not, and would not reasonably be expected to, prevent or materially delay or hinder Parent or Sub from performing in all material respects its obligations and covenants required to be performed by it under this Agreement and (y) has not been cured within twenty (20) days following notice by the Company, provided that such occurrence or circumstance shall not have resulted from or been caused by the failure of the Company or any Company Subsidiary to perform, in any material respect, any of their covenants or agreements contained in this Agreement; or (ii) due to an occurrence or circumstance that has resulted in a failure to satisfy any condition set forth in Annex I hereto, Sub shall have allowed the Offer to terminate, without having accepted any Shares for payment thereunder, unless such occurrence or circumstance shall have been caused by or resulted from the failure of the Company or any Company Subsidiary to perform, in any material respect, any of their covenants or agreements contained in this Agreement, or the material breach by the Company or any Company Subsidiary of any of their representations or warranties contained in this Agreement; or

(e) By Parent, at any time prior to the Acceptance Time, if (i) (A) the Company Board of Directors shall have (1) effected an Adverse Recommendation Change, (2) recommended to the Company's shareholders an Acquisition Proposal, or publicly announced its intention to enter into an Alternative Acquisition Agreement, (3) failed to publicly reaffirm the Company Board Recommendation within four (4) business days of Parent's request in writing that the Company Board Recommendation be publicly reaffirmed (provided that Parent is only allowed to make two (2) such reaffirmation requests prior to the Acceptance Time), or (4) failed to recommend against, or taken a neutral position with respect to, a tender or exchange offer related to an Acquisition Proposal in any position taken pursuant to Rules 14d-9 and 14e-2 under the Exchange Act; or (ii) after receiving an Acquisition Proposal, the Company shall have violated or breached in any material respect any of its obligations under Section 5.2 with respect to such Acquisition Proposal; or

(f) At any time prior to the Acceptance Time, by the Company, if the Company has received a Superior Proposal, which, after giving effect to all of the adjustments that may be offered by Parent pursuant to Section 5.2, the Company Board of Directors determines in good faith (after consultation with its financial advisors) continues to constitute a Superior Proposal, provided the Company shall not have violated or breached in any material respect any of its obligations under Section 5.2; or

(g) By the Company, prior to commencement of the Offer, if Sub has not commenced the Offer within ten (10) business days following the date hereof; provided, however, that the Company may not terminate this Agreement pursuant to this Section 8.1(g) if such failure to commence the Offer has resulted from the breach of this Agreement by the Company or if the Company has not provided Sub with a Schedule 14D-9 the Company is prepared to file, without further revisions, upon commencement of the Offer; or

(h) By Parent, at any time prior to the Acceptance Time, if there shall have occurred a Company Material Adverse Effect.

### **8.2** Effect of Termination.

(a) Any termination of this Agreement under Section 8.1 hereof will be effective immediately upon the delivery of a written notice of the terminating party to the other party hereto and, if then due, payment of the Termination Fee. In the event of termination of this Agreement as provided in Section 8.1 hereof, this Agreement shall forthwith become null and void and be of no further force or effect and there shall be no liability on the part of Parent, Sub or the Company (or any of their respective directors, officers, employees, stockholders, shareholders, agents or representatives), except as set forth in the last sentence of Section 6.4, Section 8 and Section 9, each of which shall remain in full force and effect and survive any termination of this Agreement; provided, however, that (i) nothing herein shall relieve any party from liability for fraud or intentional breach of any of its representations, warranties, covenants or agreements set forth in this Agreement (it being understood that the failure of Parent or Sub to accept for payment and pay for the Shares validly tendered and not withdrawn pursuant to the Offer promptly following the Expiration Date in the event that all Offer Conditions have been satisfied or, to the extent permitted, waived, as of the Expiration Date shall be deemed an intentional breach by Parent and Sub pursuant to this Agreement), and (ii) no termination of this Agreement shall in any way affect the parties rights or obligations with respect to any Shares accepted for payment pursuant to the Offer prior to such termination.

(b) If Parent shall have terminated this Agreement pursuant to Section 8.1(e), the Company shall promptly pay Parent a termination fee (the <u>Termination Fee</u>) of \$10,500,000, but in no event later than two (2) business days after the date of receipt of Parent's termination notice. If the Company intends to terminate this Agreement pursuant to Section 8.1(f), it shall, substantially concurrent with such termination, pay Parent the Termination Fee. If Parent terminates this Agreement pursuant to Section 8.1(c) due to an intentional or knowing material breach by the Company of a covenant or agreement of the Company set forth in this Agreement, and prior to such time an Acquisition Proposal has been made that has not been withdrawn prior to the time of such termination, then the Company shall pay Parent the Termination Fee concurrent with the consummation of any Acquisition Proposal that is consummated prior to the nine-month anniversary of the date of such termination or for which the definitive written agreement providing for the consummation of such Acquisition Proposal has been executed no later than the nine-month anniversary of the date of such termination (it being understood that for purposes of this Section 8.2(b), the references to fifteen percent (15%) in the definition of Acquisition Proposal shall be replaced with fifty percent (50%). All amounts due hereunder shall be payable by wire transfer in immediately available funds to such account as Parent may designate in writing to the Company. If the Company fails to promptly make any payment required under this Section 8.2(b) and Parent commences a suit to collect such payment, the Company shall indemnify Parent for its fees and expenses (including

attorneys fees and expenses) incurred in connection with such suit and shall pay interest on the amount of the payment at the prime rate of Bank of America (or its successors or assigns) in effect on the date the payment was payable pursuant to this Section 8.2(b). Notwithstanding anything to the contrary in this Agreement, in no event shall the Company be required to pay the Termination Fee on more than one occasion.

- **8.3** Fees and Expenses. Except as set forth in Section 8.2, all fees, costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the party incurring such expenses whether or not the Merger is consummated.
- **8.4** Amendment. Subject to applicable Law and as otherwise provided in this Agreement, this Agreement may be amended, modified and supplemented in any and all respects, whether before or after any vote of the shareholders of the Company contemplated hereby, by written agreement of the parties hereto, by action taken by their respective Boards of Directors, but after the purchase of Shares pursuant to the Offer, no amendment shall be made which decreases the Merger Consideration and, after the approval of this Agreement by the shareholders, no amendment shall be made which by Law requires further approval by such shareholders without obtaining such further approval. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto.
- **8.5** <u>Waiver</u>. At any time prior to the Effective Time, either party hereto may (a) extend the time for the performance of any of the obligations or other acts of the other party hereto or (b) waive compliance with any of the agreements of the other party or any conditions to its own obligations, in each case only to the extent such obligations, agreements and conditions are intended for its benefit; provided, that any such extension or waiver shall be binding upon a party only if such extension or waiver is set forth in a writing executed by such party.

### SECTION 9 MISCELLANEOUS

- **9.1** No Survival. None of the representations and warranties contained herein or in any schedule, instrument or document delivered pursuant to this Agreement shall survive the Acceptance Time or the termination of this Agreement.
- **9.2** <u>Notices</u>. Any notice or other communication required or permitted hereunder shall be in writing and shall be deemed given when delivered in person, by overnight courier, by facsimile transmission (with receipt confirmed by telephone or by automatic transmission report) or by electronic mail, or two (2) business days after being sent by registered or certified mail (postage prepaid, return receipt requested), as follows:

(a) if to Parent or Sub, to:

c/o Covidien

15 Hampshire Street

Mansfield, MA 02048

Attn: Vice President-Chief Mergers and Acquisition Counsel

Telephone: (508) 261-8000

Facsimile: (508) 261-8544

Email: matthew.nicolella@covidien.com

with a copy to:

| Goodwin Procter LLP                               |
|---|
| 53 State Street                                   |
| Boston, Massachusetts 02109                       |
| Attn: Joseph L. Johnson III                       |
| Telephone: (617) 570-1633                         |
| Facsimile: (617) 523-1231                         |
| E-mail: jjohnson@goodwinprocter.com               |
|   |
| (b) if to the Company, to: Somanetics Corporation |
| 2600 Troy Center Drive                            |
| Troy, MI 48084                                    |
| Attn: General Counsel                             |
| Telephone: (248) 244-1400                         |
| Facsimile: (248) 244-0978                         |
| E-mail: mavictor@somanetics.com                   |
| with a copy to:                                   |
| Honigman Miller Schwartz and Cohn LLP             |
| 2290 First National Building                      |
| 660 Woodward Avenue                               |
| Detroit, Michigan 48226-3506                      |
| Telephone: (313) 465-7452                         |
| Facsimile: (313) 465-7453                         |
| Email: rkrueger@honigman.com                      |
| Attention: Robert J. Krueger                      |
| with a copy to:                                   |
| Latham & Watkins LLP                              |

885 Third Avenue

New York, NY 10028

Telephone: (212) 906-4465

Facsimile: (212) 751-4864

Email: charles.nathan@lw.com

Attention: Charles Nathan

Any party may by notice given in accordance with this Section 9.2 to the other parties designate updated information for notices hereunder.

- **9.3** Entire Agreement. This Agreement (including the Schedules, Annexes and Exhibits hereto and the documents and instruments referred to herein) contains the entire agreement among the parties with respect to the Offer, the Merger and related transactions, and supersedes all prior agreements, written or oral, among the parties with respect thereto, other than the Confidentiality Agreement and the Tender and Voting Agreement with the CEO Shareholder.
- **9.4** Governing Law. This Agreement and all actions arising under or in connection therewith shall be governed by and construed in accordance with the Laws of the State of Delaware regardless of the Laws that might otherwise govern under applicable principles of conflicts of law thereof, <u>provided</u>, <u>however</u>, that the Laws of the respective jurisdictions of incorporation of each of the parties shall govern the relative rights, obligations, powers, duties and other internal affairs of such party and its board of directors.

### 9.5 Binding Effect; No Assignment; No Third-Party Beneficiaries.

- (a) This Agreement shall not be assigned by any of the parties hereto (whether by operation of Law or otherwise) without the prior written consent of the other parties, except that Sub may assign, in its sole discretion and without the consent of any other party, any or all of its rights, interests and obligations hereunder to (i) Parent, (ii) to Parent and one or more direct or indirect wholly-owned subsidiaries of Parent, (iii) to one or more direct or indirect wholly-owned subsidiaries of Parent or (iv) any direct or indirect holder of five percent (5%) or more of the capital stock of Parent or any subsidiary thereof (each, an Assignee ). Any such Assignee may thereafter assign, in its sole discretion and without the consent of any other party, any or all of its rights, interests and obligations hereunder to one or more additional Assignees; provided, however, that in connection with any assignment to an Assignee, Parent and Sub (and the assignor, if applicable) shall agree to remain liable for the performance by Parent and Sub (and such assignor, if applicable) of their obligations hereunder, including the obligation to pay in full for all Shares validly tendered into the Offer and the Merger Consideration, subject to the terms and conditions of the Offer and this Agreement. Subject to the preceding sentence, but without relieving any party hereto of any obligation hereunder, this Agreement will be binding upon, inure to the benefit of and be enforceable by the parties and their respective successors and assigns.
- (b) Other than Sections 6.5 and 6.9, each of which shall confer third party beneficiary rights to the parties identified therein, nothing in this Agreement, express or implied, is intended to or shall confer upon any person other than Parent, Sub and the Company and their respective successors and permitted assigns any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- **9.6** Counterparts and Signature. This Agreement may be executed in two or more counterparts (including by facsimile or by an electronic scan delivered by electronic mail), each of which shall be deemed an original but all of which together shall be considered one and the same agreement and shall become effective when counterparts have been signed by each of the parties hereto and delivered to the other parties, it being understood that all parties need not sign the same counterpart. This Agreement may be executed and delivered by facsimile or by an electronic scan delivered by electronic mail.

- 9.7 Severability. If any provision of this Agreement is held invalid or unenforceable by any court of competent jurisdiction, the other provisions of this Agreement shall remain in full force and effect. Any provision of this Agreement held invalid or unenforceable only in part or degree shall remain in full force and effect to the extent not held invalid or unenforceable. The parties further agree to replace such invalid or unenforceable provision of this Agreement with a valid and enforceable provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable provision.
- 9.8 Submission to Jurisdiction; Waiver. Each of the Company, Parent and Sub irrevocably agrees that any legal action or proceeding with respect to this Agreement or the transactions contemplated hereby or for recognition and enforcement of any judgment in respect hereof brought by the other party hereto or its successors or assigns shall be brought and determined in the state courts of the State of Delaware or, the United States District Court for the District of Delaware, and each of the Company, Parent and Sub hereby irrevocably submits with regard to any action or proceeding for itself and in respect to its property, generally and unconditionally, to the exclusive jurisdiction of the aforesaid courts. Each of the Company, Parent and Sub hereby irrevocably waives, and agrees not to assert, by way of motion, as a defense, counterclaim or otherwise, in any action or proceeding with respect to this Agreement, (a) any claim that it is not personally subject to the jurisdiction of the above-named courts for any reason other than the failure to lawfully serve process, (b) that it or its property is exempt or immune from jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise), and (c) to the fullest extent permitted by applicable Law, that (i) the suit, action or proceeding in any such court is brought in an inconvenient forum, (ii) the venue of such suit, action or proceeding is improper and (iii) this Agreement, or the subject matter hereof, may not be enforced in or by such courts.
- **9.9** Time is of the Essence. Time is of the essence with respect to the performance of this Agreement.
- **9.10** Parent Guarantee. Parent shall cause Sub to comply in all respects with each of its representations, warranties, covenants, obligations, agreements and undertakings pursuant to or otherwise in connection with this Agreement, the Offer, the Merger and the other transactions contemplated by this Agreement. As a material inducement to the Company's willingness to enter into this Agreement and perform its obligations hereunder, Parent hereby unconditionally guarantees full performance and payment by Sub of each of its covenants, obligations and undertakings pursuant to or otherwise in connection with this Agreement, the Offer, the Merger and the other transactions contemplated by this Agreement and hereby represents, acknowledges and agrees that any breach of, or other failure to perform, any such representation, warranty, covenant, obligation, agreement or undertaking of Sub shall also be deemed to be a breach or failure to perform by Parent, and the Company shall have the right, exercisable in its sole discretion, to pursue any and all available remedies it may have arising out of any such breach or nonperformance directly against either or both of Parent and Sub in the first instance.
- **9.11** Enforcement. The parties recognize and agree that if for any reason any of the provisions of this Agreement are not performed in accordance with their specific terms or are otherwise breached, immediate and irreparable harm or injury would be caused for which money

damages would not be an adequate remedy. Accordingly, each party agrees that in addition to other remedies the other party shall be entitled to an injunction without posting a bond or other undertaking restraining any violation or threatened violation of the provisions of this Agreement. In the event that any action shall be brought in equity to enforce the provisions of the Agreement, no party shall allege, and each party hereby waives the defense, that there is an adequate remedy at Law.

### **9.12** Rules of Construction; Certain Definitions.

(a) Except where expressly stated otherwise in this Agreement, the following rules of interpretation apply to this Agreement: (i) either and or are not exclusive and include, includes and including are not limiting; (ii) hereof, hereto, hereby, herein and hereunder and words of six when used in this Agreement refer to this Agreement as a whole and not to any particular provision of this Agreement; (iii) date hereof refers to the date set forth in the initial caption of this Agreement; (iv) extent in the phrase to the extent means the degree to which a subject or other thing extends, and such phrase does not mean simply if; (v) descriptive headings, the table of defined terms and the table of contents are inserted for convenience only and do not affect in any way the meaning or interpretation of this Agreement; (vi) definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms; (vii) references to a person are also to its permitted successors and assigns; (viii) references to an Article, Section, Exhibit, Annex or Schedule refer to an Article or Section of, or an Exhibit, Annex or Schedule to, thi Agreement; (ix) references to \$ or otherwise to dollar amounts refer to the lawful currency of the United States; (x) references to a federal, state, local or foreign statute or Law include any rules, regulations and delegated legislation issued thereunder; and (xi) references to a communication by a regulatory agency include a communication by the staff of such regulatory agency.

The language used in this Agreement shall be deemed to be the language chosen by the parties hereto to express their mutual intent, and no rule of strict construction shall be applied against any party hereto. No summary of this Agreement prepared by any party shall affect the meaning or interpretation of this Agreement. The parties hereto agree that they have been represented by counsel during the negotiation and execution of this Agreement and, therefore, waive the application of any Law, regulation, holding or ruling of construction providing that ambiguities in an agreement or other document shall be construed against the party drafting such agreement or document.

(b) The following definitions apply to this Agreement: (i) references to the <u>Company s knowledge , knowledge of the Company</u> and words of similar import shall mean the actual knowledge of the individuals set forth on Schedule 9.12(b); (ii) references to the number of Shares then outstanding <u>on a fully diluted basis</u> means the number of Shares then outstanding, together with the Shares pursuant to Company Restricted Shares and Shares that the Company may be required to issue pursuant to Company Options that are outstanding at that date, under the Company Stock Plans or independent of such plans, whether or not vested and whether or not their exercise prices are less than the Offer Price; (iii) the term <u>Governmental Entity</u> shall mean any foreign or domestic arbitrator, court, nation, government, any state or other political subdivision thereof and any entity exercising executive, legislative, judicial

regulatory or administrative functions of, or pertaining to, government; (iv) the term <u>person</u> shall mean any individual, corporation (including any non-profit corporation), general partnership, limited partnership, limited liability partnership, joint venture, estate, trust, company (including any limited liability company or joint stock company), firm or other enterprise, association, organization, entity or Governmental Entity and (v) <u>made available</u> means that the Company shall be deemed to have made available a document or item of information if such document or item was made available to Parent or its Representatives for review, provided to Parent or its Representatives or publicly filed with the SEC prior to the date of this Agreement or was prior to the date of this Agreement located in the electronic data room organized by the Company in connection with the due diligence investigation conducted by Parent.

- **9.13** No Waiver; Remedies Cumulative. No failure or delay on the part of any party hereto in the exercise of any right hereunder will impair such right or be construed to be a waiver of, or acquiescence in, any breach of any representation, warranty or agreement herein, nor will any single or partial exercise of any such right preclude other or further exercise thereof or of any other right. All rights and remedies existing under this Agreement are cumulative to, and not exclusive to, and not exclusive of, any rights or remedies otherwise available.
- **9.14** Waiver of Jury Trial. EACH OF PARENT, COMPANY AND SUB HEREBY IRREVOCABLY WAIVES THE RIGHT TO A TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM (WHETHER BASED IN CONTRACT, TORT OR OTHERWISE) ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT OR ANY RELATED DOCUMENT, OR ANY COURSE OF CONDUCT, COURSE OF DEALINGS, STATEMENT OR ACTION RELATED HERETO OR THERETO.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, the parties have executed this Agreement and Plan of Merger under seal as of the date first stated above.

### SOMANETICS CORPORATION

By: /s/ Bruce J. Barrett
Name: Bruce J. Barrett
Title: President and Chief Executive Officer
[Signature Page to Merger Agreement]

### UNITED STATES SURGICAL CORPORATION

By: /s/ Matthew Nicolella Name: Matthew Nicolella

Title: Vice President and Assistant Secretary

# COVIDIEN DE CORP.

By: /s/ Matthew Nicolella Name: Matthew Nicolella

Title: Vice President and Assistant Secretary

[Signature Page to Merger Agreement]

Annex I

#### Conditions to the Offer

Notwithstanding any other provisions of the Offer, and in addition to (and not in limitation of) Sub s rights to extend and amend the Offer at any time in its sole discretion (subject to the provisions of the Merger Agreement), Sub shall not be required to accept for payment or, subject to any applicable rules and regulations of the SEC, including Rule 14e-l(c) under the Exchange Act (relating to Sub s obligation to pay for or return tendered Shares promptly after termination or withdrawal of the Offer), pay for, and may delay the acceptance for payment of or, subject to the restriction referred to above, the payment for, any validly tendered Shares if (i) immediately prior to the expiration of the Offer (as extended in accordance with the Merger Agreement), the number of Shares validly tendered (including by guaranteed delivery) and not validly withdrawn, together with any Shares beneficially owned by Parent or any Affiliate or subsidiary of Parent or with respect to which Parent or Sub has, directly or indirectly, voting power, does not equal at least a majority of the Shares then outstanding on a fully diluted basis (the Minimum Condition ); (ii) immediately prior to the expiration of the Offer (as extended in accordance with the Merger Agreement), any waiting period (and any extensions thereof) and any approvals or clearances applicable to the Offer or the consummation of the Merger under the HSR Act or any provision under comparable applicable Laws of foreign jurisdictions shall not have expired, or been terminated or obtained, as applicable; or (iii) any of the following events shall occur and be continuing (and not be waived):

(a) there shall be pending any suit, action or proceeding (i) seeking to prohibit or impose any material limitations on Parent s or Sub s ownership or operation (or that of any of their respective Subsidiaries or Affiliates) of all or any material portion of the Company s or the Company Subsidiaries businesses or assets, taken as a whole, or to compel Parent or Sub or their respective Subsidiaries or Affiliates to dispose of or hold separate any material portion of the business or assets of the Company, the Company Subsidiaries, Parent, Sub or their respective Affiliates, (ii) seeking to prohibit or make illegal the making or consummation of the Offer or the Merger or the performance of any of the other transactions contemplated by the Merger Agreement, (iii) seeking to impose material limitations on the ability of Sub, or render Sub unable, to accept for payment, pay for or purchase some or all of the Shares pursuant to the Offer or the Merger, (iv) seeking to impose material limitations on the ability of Sub or Parent effectively to exercise full rights of ownership of the Shares, including, without limitation, the right to vote the Shares purchased by it on all matters properly presented to the Company s shareholders, or (v) seeking to require divestiture by Parent or any of its Subsidiaries or Affiliates of any Shares;

(b) there shall be any statute, rule, regulation, judgment, order or injunction enacted, entered, enforced, promulgated or deemed applicable, pursuant to an authoritative interpretation by or on behalf of a Government Entity, to the Offer, the Merger or any other transaction contemplated hereby, or any other action shall be taken by any Governmental Entity, that would reasonably be expected to result, directly or indirectly, in any of the consequences referred to in clauses (i) through (v) of paragraph (a) of this Annex I;

Annex I-1

- (c) (1) the representations and warranties of the Company contained in the Agreement (other than those in Section 3.2 (Authority), Section 3.3(a) and (b) (Capitalization), the first sentence of Section 3.8 (Absence of Adverse Changes) and Section 3.21 (Financial Advisor)) shall not be true and correct (A) as of the date of the Merger Agreement except for representations and warranties that relate to a specific date or time (which need only be true and correct as of such date or time) and (B) as of the Acceptance Time except for representations and warranties that relate to a specific date or time (which need only be true and correct as of such date or time), in each case, without giving effect to the words materially or material or to any qualification based on the defined term. Company Material Adverse Effect., except where the failure to be so true and correct, individually or in the aggregate, does not have or would not reasonably be expected to have a Company Material Adverse Effect; or (2) the representations and warranties of the Company contained in Section 3.2 (Authority), Section 3.3 (a) and (b) (Capitalization), the first sentence of Section 3.8 (Absence of Adverse Changes) and Section 3.21 (Financial Advisor) shall not be true and correct in all respects as of the date of the Merger Agreement and the Acceptance Time; provided however that Sections 3.3(a) and 3.3(b) shall not be deemed to not be true and correct in all respects in the event of any de minimus exceptions involving discrepancies of no more than 20,000 Shares in the aggregate or Shares issuable pursuant to Company Options or Company Restricted Stock;
- (d) since the date of the Merger Agreement, there shall have occurred any events or changes which have had, or which are reasonably likely to have or constitute, individually or in the aggregate, a Company Material Adverse Effect;
- (e) the Company Board of Directors shall have (i) effected an Adverse Recommendation Change, (ii) recommended to the Company s shareholders any proposal other than by Parent or Sub in respect of an Acquisition Proposal, or publicly announced its intention to enter into any agreement in respect of any Acquisition Proposal, (iii) failed to publicly reaffirm its recommendation of the Merger Agreement or the Offer within four (4) business days of Parent s request (provided that Parent is only allowed to make two (2) such reaffirmation requests prior to the Acceptance Time), or (iv) failed to recommend against, or taken a neutral position with respect to, a tender or exchange offer related to an Acquisition Proposal in any position taken pursuant to Rules 14d-9 and 14e-2 under the Exchange Act;
- (f) the Company shall have breached or failed to perform or to comply in any material respect with any agreement, obligation or covenant to be performed or complied with by it under the Merger Agreement and such breach or failure to perform or to comply shall not have been cured by the Company;
- (g) Parent and Sub shall have failed to receive a certificate executed by the Company s Chief Executive Officer or President on behalf of the Company, dated as of the then-scheduled expiration of the Offer, to the effect that the conditions set forth in paragraphs (c), (d) and (e) of this Annex I have not occurred; or
- (h) the Merger Agreement shall have been terminated in accordance with its terms.

Annex I-2

The foregoing conditions are for the sole benefit of Parent and Sub, may be asserted by Parent or Sub regardless of the circumstances giving rise to such condition, and may be waived by Parent or Sub in whole or in part at any time and from time to time and in the sole discretion of Parent or Sub, subject in each case to the terms of the Merger Agreement. The failure by Parent or Sub at any time to exercise any of the foregoing rights shall not be deemed a waiver of any such right and, each such right shall be deemed an ongoing right which may be asserted at any time and from time to time.

The term <u>Merger Agreement</u> shall be deemed to refer to this Agreement to which this <u>Annex I</u> is annexed. The other capitalized terms used in this <u>Annex I</u> shall have the meanings set forth in the Merger Agreement.

Annex I-3

## ANNEX II

## FORM OF TENDER AND VOTING AGREEMENT

[See Exhibit (d)(2) of the Schedule TO]

Annex II-1

Exhibit 3

### TENDER AND VOTING AGREEMENT

THIS TENDER AND VOTING AGREEMENT (this <u>Agreement</u>) dated June 16, 2010, is entered into between United States Surgical Corporation, a Delaware corporation (<u>Parent</u>), Covidien DE Corp., a Delaware corporation and direct or indirect wholly owned subsidiary of Parent (<u>Sub</u>) and the undersigned shareholder of the Company (<u>Shareholder</u>), with respect to (a) the shares of common stock, par value \$0.01 per share (the <u>Shares</u>), of Somanetics Corporation, a Michigan corporation (the <u>Company</u>), (b) all securities exchangeable, exercisable or convertible into Shares and (c) any securities issued or exchanged with respect to such Shares upon any recapitalization, reclassification, merger, consolidation, spin-off, partial or complete liquidation, stock dividend, split-up or combination of the securities of the Company or upon any other change in the Company s capital structure, in each case whether now owned or hereafter acquired by the Shareholder (collectively, the <u>Securities</u>).

### WITNESSETH:

WHEREAS, Parent, Sub and the Company have entered into an Agreement and Plan of Merger dated as of the date hereof (as the same may be amended or supplemented, other than to lower the price to be paid in the Offer or Merger, the <a href="Merger Agreement">Merger Agreement</a>) pursuant to which Sub has agreed to make a cash tender offer as described therein and thereafter merge with and into the Company (the <a href="Merger">Merger</a>) with the result that the Company becomes a wholly owned subsidiary of Parent;

**WHEREAS**, as of the date hereof, Shareholder is the record or beneficial owner and has the power to dispose of the Securities set forth on <u>Schedule I</u> hereto and has the power to vote the Shares set forth thereon, except as otherwise set forth on <u>Schedule I</u>;

WHEREAS, Parent and Sub desire to enter into this Agreement in connection with their efforts to consummate the acquisition of the Company; and

WHEREAS, capitalized terms used in this Agreement and not defined have the meaning given to such terms in the Merger Agreement.

**NOW, THEREFORE**, in contemplation of the foregoing and in consideration of the mutual agreements, covenants, representations and warranties contained herein and intending to be legally bound hereby, the parties hereto agree as follows:

### 1. Certain Covenants.

1.1 <u>Lock-Up</u>. Subject to Section 1.5, except as contemplated by the Merger Agreement, Shareholder hereby covenants and agrees that between the date hereof and the Termination Date (as defined below), Shareholder will not (a) directly or indirectly, sell, transfer, assign, pledge, hypothecate, tender, encumber or otherwise dispose of or limit its right to vote in any manner any of the Securities, or agree to do any of the foregoing, or (b) take any action which would have the effect of preventing or disabling Shareholder from performing its obligations under this Agreement. Notwithstanding the foregoing, in connection with any transfer not involving or relating to any Acquisition Proposal, Shareholder may transfer any or

all of the Securities as follows: (i) in the case of a Shareholder that is an entity, to any subsidiary, partner or member of Shareholder, and (ii) in the case of an individual Shareholder, to Shareholder s spouse, ancestors, descendants or any trust for any of their benefits or to a charitable trust; provided, however, that in any such case, prior to and as a condition to the effectiveness of such transfer, (A) each person to which any of such Securities or any interest in any of such Securities is or may be transferred (1) shall have executed and delivered to Parent and Sub a counterpart to this Agreement pursuant to which such person shall be bound by all of the terms and provisions of this Agreement, and (2) shall have agreed in writing with Parent and Sub to hold such Securities or interest in such Securities subject to all of the terms and provisions of this Agreement, and (B) this Agreement shall be the legal, valid and binding agreement of such person, enforceable against such person in accordance with its terms.

- 1.2 No Solicitation. Between the date hereof and the Termination Date, the Shareholder shall not, directly or indirectly, take any action that the Company is prohibited from taking pursuant to Section 5.2 of the Merger Agreement.
- 1.3 <u>Certain Events</u>. This Agreement and the obligations hereunder will attach to the Securities and will be binding upon any person to which legal or beneficial ownership of any or all of the Securities passes, whether by operation of Law or otherwise, including without limitation, Shareholder s successors or assigns. This Agreement and the obligations hereunder will also attach to any additional Shares or other Securities of the Company issued to or acquired by Shareholder.

### 1.4 Grant of Proxy; Voting Agreement.

- (a) Shareholder has revoked or terminated any proxies, voting agreements or similar arrangements previously given or entered into with respect to the Securities and hereby irrevocably appoints Parent as proxy for Shareholder to vote the Securities for Shareholder and in Shareholder s name, place and stead, at any annual, special or other meeting or action of the shareholders of the Company, as applicable, or at any adjournment thereof or pursuant to any consent of the shareholders of the Company, in lieu of a meeting or otherwise, whether before or after the Acceptance Time, for the adoption of the Merger Agreement and the approval of Merger. Parent hereby acknowledges that the proxy granted hereby shall not be effective for any other purpose. The parties acknowledge and agree that neither Parent, nor Parent s successors, assigns, subsidiaries, divisions, employees, officers, directors, stockholders, agents and affiliates shall owe any duty to, whether in law or otherwise, or incur any liability of any kind whatsoever, including without limitation, with respect to any and all claims, losses, demands, causes of action, costs, expenses (including reasonable attorney s fees) and compensation of any kind or nature whatsoever to Shareholder in connection with or as a result of any voting by Parent of the Securities subject to the irrevocable proxy hereby granted to Parent at any annual, special or other meeting or action or the execution of any consent of the shareholders of the Company for the purpose set forth herein.
- (b) Notwithstanding the foregoing grant to Parent of the irrevocable proxy, if Parent elects not to exercise its rights to vote the Securities pursuant to the irrevocable proxy, Shareholder agrees to vote the Securities during the term of this Agreement in favor of or give its consent to, as applicable, a proposal to adopt and approve the Merger Agreement and thereby approve the Merger at any annual, special or other meeting or action of the shareholders of the Company, in lieu of a meeting or otherwise.

(c) This irrevocable proxy shall not be terminated by any act of Shareholder or by operation of law, whether by the death or incapacity of Shareholder or by the occurrence of any other event or events (including, without limiting the foregoing, the termination of any trust or estate for which Shareholder is acting as a fiduciary or fiduciaries or the dissolution or liquidation of any corporation or partnership). If between the execution hereof and the Termination Date, Shareholder should die or become incapacitated, or if any trust or estate holding the Securities should be terminated, or if any corporation or partnership holding the Securities should be dissolved or liquidated, or if any other such similar event or events shall occur before the Termination Date, certificates representing the Securities shall be delivered by or on behalf of Shareholder in accordance with the terms and conditions of the Merger Agreement and this Agreement, and actions taken by Parent hereunder shall be as valid as if such death, incapacity, termination, dissolution, liquidation or other similar event or events had not occurred, regardless of whether or not Parent has received notice of such death, incapacity, termination, dissolution, liquidation or other event.

1.5 <u>Tender of Securities</u>. Shareholder agrees, in exchange for the consideration described in the Merger Agreement, to (a) tender any Shares owned of record by Shareholder or for which Shareholder has sole dispositive power prior to the Commencement Date to Sub in the Offer not later than five (5) business days following the Commencement Date, and (b) tender any Shares acquired by Shareholder after the Commencement Date to Sub in the Offer not later than the earlier of (i) five (5) business days following the acquisition of such Shares or (ii) the Acceptance Date, and Shareholder shall not withdraw any Shares so tendered under clauses (a) and (b) above unless the Offer is terminated or this Agreement shall have been terminated in accordance with its terms; <u>provided</u>, <u>however</u>, that Shareholder shall not be required, for purposes of this Agreement, to exercise any unexercised Company Options held by such Shareholder or tender any Shares granted to such Shareholder under an employee benefit plan of the Company which are unvested and subject to any risk of forfeiture, and Shareholder shall not have any obligation under this Section 1.5 to tender such Shareholder s Shares into the Offer if that tender could cause such Shareholder to incur liability under Section 16(b) of the Exchange Act.

### 1.6 Option.

(a) On the terms and subject to the conditions set forth herein, Shareholder hereby grants to Parent an irrevocable option (the Option ) to purchase all of the right, title and interest of Shareholder in and to Shareholder s Securities with a price per share equal to the Offer Price; provided however that the Option shall not include Securities issued pursuant to the Plans, which are not transferrable. Parent may exercise the Option in whole, but not in part, if, but only if, (i) Sub has acquired Shares pursuant to the Offer and (ii) Shareholder has failed to tender into the Offer any Shares or shall have withdrawn the tender of any Shares into the Offer in breach of this Agreement. Parent may exercise the Option at any time within the sixty (60) days following the date when such Option first becomes exercisable.

- (b) In the event that Parent is entitled to and wishes to exercise the Option, Parent shall send a written notice to Shareholder specifying the place and the date for the closing of such purchase, which date shall be not more than sixty (60) days after the date of such notice; provided that in the event that prior notification to, or approval of, any Governmental Entity is required in connection with the exercise of the Option or there shall be in effect any preliminary or final injunction or other order issued by any Governmental Entity prohibiting the exercise of the Option, the period of time during which the date of the closing may be fixed shall be extended until the tenth (10th) day following the last date on which all required approvals shall have been obtained, all required waiting periods shall have expired or been terminated and any such prohibition shall have been vacated, terminated or waived.
- (c) At the closing of the purchase of Shareholder s Securities pursuant to exercise of the Option, simultaneously with the payment by Parent of the purchase price for Shareholder s Securities, such Shareholder shall deliver, or cause to be delivered, to the Sub certificates representing such Securities duly endorsed to Parent or accompanied by stock powers or other transfer documents duly executed by the Company in blank, together with any necessary stock transfer stamps properly affixed, free and clear of all Liens (as defined below).
- (d) Parent, Sub or the Company, as applicable, shall be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Section 1.6 to a holder of Securities such amounts as are required to be withheld under the Code or the Treasury Regulations thereunder or any other Tax Law. To the extent that amounts are so deducted and withheld, such amounts shall be treated for all purposes of this Agreement as having been paid to the holder of the Securities in respect of which such deduction and withholding was made. To the extent any amounts are required to be deducted and withheld under the Tax Law of a jurisdiction outside the United States, and such deduction would not have been required were Parent incorporated or organized in the United States or a subdivision thereof, additional amounts shall be paid as necessary to ensure that the Shareholder receives the same amount that would otherwise have been received as if no such deduction and withholding had been made.
- 1.7 <u>Public Announcement</u>. Shareholder shall consult with Parent before issuing any press releases or otherwise making any public statements with respect to the transactions contemplated herein and shall not issue any such press release or make any such public statement without the approval of Parent, except as may be required by Law, including any filings with the SEC pursuant to the Exchange Act. This Section 1.7 shall terminate and be null and void upon the earlier of (a) the Termination Date and (b) consummation of the Merger.
- 1.8 <u>Disclosure</u>. Shareholder hereby authorizes Parent and Sub to publish and disclose in any announcement or disclosure required by the SEC, The NASDAQ Stock Market or the New York Stock Exchange or any other national securities exchange and in the Offer Documents and, if necessary, the Proxy Statement (including all documents and schedules filed with the SEC in connection with either of the foregoing), Shareholder s identity and ownership of the Securities and the nature of Shareholder s commitments, arrangements and understandings under this Agreement. Parent and Sub hereby authorize Shareholder to make such disclosure or filings as may be required by the SEC, the New York Stock Exchange, the NASDAQ Stock Market or any other national securities exchange.

- 2. <u>Representations and Warranties of Shareholder</u>. Shareholder hereby represents and warrants to Parent and Sub, as of the date hereof and as of the date Sub purchases Shares pursuant to the Offer, that:
- 2.1 Ownership. As of the date hereof, Shareholder holds of record or beneficially the Securities set forth on Schedule I, in each case, except as set forth on Schedule I, free and clear of all liabilities, claims, liens, options, proxies, charges, participations and encumbrances of any kind or character whatsoever, other than those arising under the securities laws or under the Company s governance documents (collectively. Liens ). At the time Sub purchases Shares pursuant to the Offer, Shareholder will transfer and convey to Sub or its designee good and marketable title to the Shares included in the Securities, free and clear of all Liens created by or arising through Shareholder.
- 2.2 <u>Authorization</u>. Shareholder has all requisite power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby and has sole voting power and sole power of disposition, with respect to the Securities with no restrictions on its voting rights or rights of disposition pertaining thereto, except as set forth in the Securities or pursuant to applicable community property laws. Shareholder has duly executed and delivered this Agreement and this Agreement is a legal, valid and binding agreement of Shareholder, enforceable against Shareholder in accordance with its terms, except to the extent enforceability may be limited by the effect of applicable bankruptcy, reorganization, insolvency, moratorium or other Laws affecting the enforcement of creditors rights generally and the effect of general principles of equity, regardless of whether such enforceability is considered in a proceeding at Law or in equity. If the Shareholder is spouse, and this Agreement is a legal, valid and binding agreement of the Shareholder is spouse, enforceable against the Shareholder in accordance with its terms, except to the extent enforceability may be limited by the effect of applicable bankruptcy, reorganization, insolvency, moratorium or other Laws affecting the enforcement of creditors rights generally and the effect of general principles of equity, regardless of whether such enforceability is considered in a proceeding at Law or in equity.
- 2.3 No Violation. Neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby will (a) require Shareholder to file or register with, or obtain any permit, authorization, consent or approval of, any governmental agency, authority, administrative or regulatory body, court or other tribunal, foreign or domestic, or any other entity other than filings with the SEC pursuant to the Exchange Act, or (b) violate, or cause a breach of or default under, or conflict with any contract, agreement or understanding, any Law binding upon Shareholder, except for such violations, breaches, defaults or conflicts which are not, individually or in the aggregate, reasonably likely to have an adverse effect on Shareholder's ability to satisfy its obligations under this Agreement. As of the date hereof, no proceedings are pending which, if adversely determined, will have an adverse effect on Shareholder's ability to vote or dispose of any of the Securities.
- 2.4 <u>Shareholder Has Adequate Information</u>. Shareholder is a sophisticated seller with respect to the Securities and has adequate information concerning the business and financial condition of the Company to make an informed decision regarding the sale of the Securities and has independently and without reliance upon either Sub or Parent and based on such information as Shareholder has deemed appropriate, made its own analysis and decision to enter into this Agreement. Shareholder acknowledges that neither Sub nor Parent has made and neither makes any representation or warranty, whether express or implied, of any kind or character except as expressly set forth in this Agreement. Shareholder acknowledges that the

agreements contained herein with respect to the Securities held by Shareholder are irrevocable (prior to the Termination Date), and that Shareholder shall have no recourse to the Securities, Parent or Sub, except with respect to breaches of representations, warranties, covenants and agreements expressly set forth in this Agreement.

- 2.5 No Setoff. Shareholder has no liability or obligation related to or in connection with the Securities other than the obligations to Parent and Sub as set forth in this Agreement.
- 2.6 No Amounts Payable to Shareholder. Except as disclosed in the Merger Agreement, there are no amounts due or payable by the Company or any Company Subsidiary to Shareholder or any of its affiliates or associates in connection with the transactions contemplated by the Merger Agreement or this Agreement or otherwise (other than any payments required under the Merger Agreement solely in exchange for equity securities of the Company or payments, if any, to be made pursuant to Plans disclosed in the Company Disclosure Schedule).
- 3. Representations and Warranties of Parent and Sub. Parent and Sub hereby represent and warrant to Shareholder, as of the date hereof that:
- 3.1 <u>Authorization</u>. Parent and Sub have all requisite corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby. Parent and Sub have duly executed and delivered this Agreement and this Agreement is a legal, valid and binding agreement of each of Parent and Sub, enforceable against each of Parent and Sub in accordance with its terms, except to the extent enforceability may be limited by the effect of applicable bankruptcy, reorganization, insolvency, moratorium or other Laws affecting the enforcement of creditors—rights generally and the effect of general principles of equity, regardless of whether such enforceability is considered in a proceeding at Law or in equity.
- 3.2 No Violation. Neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby will violate, or cause a breach of or default under, any contract or agreement, any statute or law, or any judgment, decree, order, regulation or rule of any governmental agency, authority, administrative or regulatory body, court or other tribunal, foreign or domestic, or any other entity or any arbitration award binding upon Parent or Sub, except for such violations, breaches or defaults which are not reasonably likely to have a material adverse effect on either Parent s or Sub s ability to satisfy its obligations under this Agreement.
- 4. <u>Survival of Representations and Warranties</u>. The respective representations and warranties of Shareholder, Parent and Sub contained herein shall not be deemed waived or otherwise affected by any investigation made by the other party hereto.
- 5. Specific Performance. Shareholder acknowledges that Sub and Parent will be irreparably harmed and that there will be no adequate remedy at law for a violation of any of the covenants or agreements of Shareholder which are contained in this Agreement. It is accordingly agreed that, in addition to any other remedies which may be available to Sub and Parent upon the breach by Shareholder of such covenants and agreements, Sub and Parent shall have the right to obtain injunctive relief to restrain any breach or threatened breach of such covenants or agreements or otherwise to obtain specific performance of any of such covenants or agreements.

### 6. Miscellaneous.

- 6.1 <u>Term</u>. This Agreement and all obligations hereunder shall terminate upon the earlier of (a) the Effective Time, (b) the Outside Date, (c) the date of any modification, waiver or amendment to the Merger Agreement in a manner that reduces the Offer Price and (d) the termination of the Merger Agreement pursuant to Section 8.1 thereof (the earliest of (a), (b), (c) and (d), the <u>Termination Date</u>). Upon termination of this Agreement, no party shall have any further obligations or liabilities under this Agreement; <u>provided</u>, <u>however</u>, that (i) nothing set forth in this Section 6.1 shall relieve any party from liability for any willful breach of this Agreement prior to termination hereof, and (ii) the provisions of this Article 6 shall survive any termination of this Agreement.
- 6.2 <u>Capacity as a Shareholder; Fiduciary Duties.</u> Notwithstanding anything in this Agreement to the contrary: (a) the Shareholder makes no agreement or understanding herein in any capacity other than in the Shareholder s capacity as a record holder and beneficial owner of Securities, and not in such Shareholder s capacity as a director, officer or employee of the Company or any of Company Subsidiaries or in such Shareholder s capacity as a trustee or fiduciary of any employee benefit plan or trust, and (b) nothing herein will be construed to limit or affect any action or inaction by the Shareholder serving on the Company Board of Directors or on the board of directors of any Company Subsidiary or as an officer or fiduciary of the Company, any Company Subsidiary or any employee benefit plan or trust, acting in such person s capacity as a director, officer, trustee and/or fiduciary.
- 6.3 Expenses. Each of the parties hereto shall pay its own expenses incurred in connection with this Agreement. Each of the parties hereto warrants and covenants to the others that it will bear all claims for brokerage fees attributable to action taken by it.
- 6.4 <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of and be enforceable by the parties hereto and their respective representatives and permitted successors and assigns.
- 6.5 Entire Agreement. This Agreement contains the entire understanding of the parties and supersedes all prior agreements and understandings between the parties with respect to its subject matter. This Agreement may be amended only by a written instrument duly executed by the parties hereto.
- 6.6 <u>Headings</u>. The headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.
- 6.7 <u>Assignment</u>. Without limitation to Section 1.1, this Agreement shall be binding upon and inure to the benefit of the parties named herein and their respective successors and permitted assigns. No party may assign either this Agreement or any of its rights, interests, or obligations hereunder without the prior written approval of the other parties; <u>provided</u>, <u>however</u>, that each of Parent and Sub may freely assign its rights to another direct or indirect wholly owned subsidiary of Parent or Sub without such prior written approval but no such assignment shall relieve Parent or Sub of any of its obligations hereunder. Any purported assignment requiring consent without such consent shall be void.

- 6.8 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which shall be an original, but each of which together shall constitute one and the same Agreement.
- 6.9 <u>Notices</u>. Any notice or other communication required or permitted hereunder shall be in writing and shall be deemed given when delivered in person, by overnight courier, by facsimile transmission (with receipt confirmed by telephone or by automatic transmission report) or by electronic mail, or two business days after being sent by registered or certified mail (postage prepaid, return receipt requested), as follows:
  - (a) if to Parent or Sub, to:

c/o Covidien

15 Hampshire Street

Mansfield, MA 02048

Attn: Vice President-Chief Mergers and Acquisition Counsel

Telephone: (508) 261-8000

Facsimile: (508) 261-8544

Email: matthew.nicolella@covidien.com

with a copy to:

Goodwin Procter LLP

53 State Street

Boston, Massachusetts 02109

Attn: Joseph L. Johnson III

Telephone: (617) 570-1633

Facsimile: (617) 523-1231

Email: jjohnson@goodwinprocter.com

without regard to its principles of conflicts of Laws.

- (b) If to Shareholder, to the addresses indicated on Schedule I hereto.

  Any party may by notice given in accordance with this Section 6.9 to the other parties to designate updated information for notices hereunder.
- 6.10 Governing Law. This Agreement shall be governed by and construed and enforced in accordance with the Laws of the State of Delaware,
- 6.11 <u>Enforceability</u>. The invalidity or unenforceability of any provision or provisions of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement, which shall remain in full force and effect. Upon a determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto will,

subject to Section 1.9(d), negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner to the end that the transactions contemplated hereby are fulfilled to the fullest extent possible and, absent agreement among the parties, a court is authorized to so modify this Agreement. Without limiting the foregoing, the covenants contained herein shall be construed as separate covenants, covering their respective subject matters, with respect to each of the separate cities, counties and states of the United States, and each other country, and political subdivision thereof, in which any of the Company or its successors now transacts any business.

- 6.12 <u>Further Assurances</u>. From time to time, at Parent s request and without further consideration, Shareholder shall execute and deliver to Parent such documents and take such action as Parent may reasonably request in order to consummate more effectively the transactions contemplated hereby and to vest in Parent good, valid and marketable title to the Securities, including, but not limited to, using its best efforts to cause the appropriate transfer agent or registrar to transfer of record the Securities.
- 6.13 <u>Remedies Not Exclusive</u>. All rights, powers and remedies provided under this Agreement or otherwise available in respect hereof at law or in equity will be cumulative and not alternative, and the exercise of any thereof by either party will not preclude the simultaneous or later exercise of any other such right, power or remedy by such party.
- 6.14 Waiver of Jury Trial. EACH PARTY HERETO IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.
- 6.15 No Agreement Until Executed. Irrespective of negotiations among the parties or the exchanging of drafts of this Agreement, this Agreement shall not constitute or be deemed to evidence a contract, agreement, arrangement or understanding between the parties hereto unless and until (a) the Company Board of Directors has adopted and approved, for purposes of any applicable anti-takeover laws and regulations, and any applicable provision of the Company s certificate of incorporation, the possible acquisition of the Shares by Parent and Sub pursuant to the Merger Agreement, (b) the Merger Agreement is executed by all parties thereto and (c) this Agreement is executed by all parties hereto.

[The rest of this page has intentionally been left blank]

**IN WITNESS WHEREOF**, Parent, Sub and Shareholder have caused this Agreement to be duly executed as of the day and year first above written.

### UNITED STATES SURGICAL CORPORATION

By: /s/ Matthew Nicolella Name: Matthew Nicolella

Title: Vice President and Assistant Secretary

## COVIDIEN DE CORP.

By: /s/ Matthew Nicolella Name: Matthew Nicolella

Title: Vice President and Assistant Secretary

## SHAREHOLDER:

/s/ Bruce J. Barrett Name: Bruce J. Barrett

[Signature Page to Tender and Voting Agreement]

### **SCHEDULE I TO**

### THE TENDER AND VOTING AGREEMENT

1. Securities held by Shareholder:

Shares under

Options to Company

SharesPurchase SharesRestricted Shares124,691 (including 17,000 owned<br/>jointly with his wife)553,919 (of which 517,919 are exercisable<br/>and an additional 7,200 will vest on 6.29.10)62,800

Transfer of Company Options and Company Restricted Shares are restricted pursuant to the terms of the Plans and the related option and restricted share agreements.

2. Address to which notices or other communications are to be sent in accordance with Section 6.9 of this Agreement:

Shareholder: Bruce J. Barrett

c/o Somanetics Corporation 2600 Troy Center Drive Troy, MI 48084

Facsimile: (866) 741-0504 Email: bbarrett@somanetics.com

Exhibit 4

### **GUARANTY**

United States Surgical Corporation, a Delaware corporation (<u>Parent</u>), Covidien DE Corp., a Delaware corporation (Sub) and Somanetics Corporation, a Michigan corporation (the <u>Company</u>), have entered into, simultaneously with the execution and delivery hereof, an Agreement and Plan of Merger (the <u>Merger Agreement</u>). Parent is a direct subsidiary of Covidien International Finance S.A., a Luxembourg corporation (<u>CIFSA</u>), and CIFSA expects to derive substantial indirect benefits from the consummation of the transactions contemplated by the Merger Agreement. Capitalized terms that are used but not defined herein shall have the meanings ascribed to such terms in the Merger Agreement.

In consideration of, and as an inducement to the Company entering into the Merger Agreement and performing its obligations thereunder, CIFSA hereby unconditionally guarantees to the Company the full performance and payment by Parent of its covenants, obligations and undertakings pursuant to or otherwise in connection with the Merger Agreement, the Offer, the Merger and the other transactions contemplated by the Merger Agreement and hereby represents, acknowledges and agrees that any breach of, or other failure to perform, any such representation, warranty, covenant, obligation, agreement or undertaking of Parent shall also be deemed to be a breach or failure to perform by CIFSA, and the Company shall have the right, exercisable in its sole discretion, to pursue any and all available remedies it may have arising out of any such breach or nonperformance directly against either or both of CIFSA and Parent in the first instance.

CIFSA hereby represents and warrants that: (a) it is a corporation, duly organized, validly existing and in good standing under the Laws of Luxembourg, (b) all of the issued and outstanding capital stock of Parent is owned, beneficially and of record, by CIFSA, (c) it has corporate power and authority to enter into, execute and deliver this guaranty and to perform fully its obligations hereunder, (d) this guaranty has been duly executed and delivered by CIFSA and constitutes a valid and binding obligation, enforceable against CIFSA in accordance with its terms, except to the extent enforceability may be limited by the effect of applicable bankruptcy, reorganization, insolvency, moratorium or other Laws affecting the enforcement of creditors—rights generally and the effect of general principles of equity, regardless of whether such enforceability is considered in a proceeding at Law or in equity, and (e) the execution, delivery and performance of this guaranty by CIFSA does not and will not (i) violate any provision of the certificate of incorporation, bylaws or comparable organizational document of CIFSA; (ii) violate, conflict with or result in the breach of any of the terms or conditions of, result in modification of, require any notice or action under, or otherwise give any other contracting party the right to terminate, accelerate obligations under or receive payment under or constitute (or with notice or lapse of time or both constitute) a default under, any instrument, contract or other agreement to which CIFSA is a party or to which it or its assets or properties is bound or subject, (iii) violate any Law applicable to CIFSA or by which any of its assets or properties is bound, (iv) violate any governmental permit, (v) require any filing with, notice to, or permit, consent or approval of, any Governmental

Entity, excluding from the foregoing clauses (ii), (iii), (iv) and (v) violations, conflicts breaches, modifications and defaults which, and filings, notices, permits, consents and approvals the absence of which, in the aggregate, would not reasonably be expected to have a material adverse effect on the ability of CIFSA to perform its obligations pursuant to this guaranty.

This guaranty and all actions arising under or in connection therewith shall be governed by and construed in accordance with the Laws of the State of Delaware, regardless of the Laws that might otherwise govern under applicable principles of conflicts of law thereof. CIFSA irrevocably agrees that any legal action or proceeding with respect to this guaranty or for recognition and enforcement of any judgment in respect hereof brought the Company shall be brought and determined in the state courts of the State of Delaware or, the United States District Court for the District of Delaware, and CIFSA hereby irrevocably submits with regard to any action or proceeding for itself and in respect to its property, generally and unconditionally, to the exclusive jurisdiction of the aforesaid courts. CIFSA hereby irrevocably waives, and agrees not to assert, by way of motion, as a defense, counterclaim or otherwise, in any action or proceeding with respect to this guaranty, (a) any claim that it is not personally subject to the jurisdiction of the above-named courts for any reason other than the failure to lawfully serve process, (b) that it or its property is exempt or immune from jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise), and (c) to the fullest extent permitted by applicable Law, that (i) the suit, action or proceeding in any such court is brought in an inconvenient forum, (ii) the venue of such suit, action or proceeding is improper and (iii) this guaranty, or the subject matter hereof, may not be enforced in or by such courts.

[Signature Page Follows]

IN WITNESS WHEREOF, the undersigned has executed this guaranty as of the date set forth below.

COVIDIEN INTERNATIONAL FINANCE S.A.

By: /s/ Michelangelo Stefani

Name: Michelangelo Stefani

Title: Managing Director

Date: June 16, 2010